Dissertation zur Erlangung des Doktorgrades der Fakultät für Chemie und Pharmazie der Ludwig-Maximilians-Universität München

SPRAY CONGEALED SOLID LIPID MICROPARTICLES FOR THE SUSTAINED RELEASE OF PEPTIDES IN VETERINARY USE



Kerstin Hoffmann

aus Ruthweiler, Deutschland

2017

Erklärung

Diese Dissertation wurde im Sinne von §7 der Promotionsordnung vom 28. November 2011 von Herrn Prof. Dr. Wolfgang Frieß betreut.

Eidesstattliche Versicherung

Diese Dissertation wurde eigenständig und ohne unerlaubte Hilfe erarbeitet.

Mainz, 09.03.2017

.....

Kerstin Hoffmann

Dissertation eingereicht am: 10.03.2017

1. Gutachter: Prof. Dr. Wolfgang Frieß

2. Gutachter: Prof. Dr. Gerhard Winter

Mündliche Prüfung am 12.04.2017

FÜR MEINE ELTERN

& MEINE SCHWESTER

Für Hubert

ACKNOWLEDGEMENT

This work was prepared between June 2013 and September 2016 in the Department of Pharmacy, Pharmaceutical Technology and Biopharmaceutics at the Ludwig-Maximilians-University in Munich under supervision of Prof Dr. Wolfgang Frieß. The project was conducted in cooperation with Veyx Pharma GmbH in Schwarzenborn and the Faculty of Veterinary Medicine of the University of Leipzig and funded from the Deutsche Bundesumweltstiftung (DBU).

First and foremost, my deepest gratitude goes to my supervisor Prof. Dr. Wolfgang Frieß for his scientific input and highly professional guidance, but also for him being such a great "Doktorvater". I am very thankful to have had the opportunity to join his research group and to work on this relevant topic. I also want to thank Prof. Dr. Gerhard Winter for being the chair of this institute and for creating the outstanding working atmosphere.

Furthermore, I want to thank my cooperation partners PD Dr. Dr. Wolfgang Zaremba from Veyx Pharma GmbH and Prof. Dr. Johannes Kauffold from the Faculty of Veterinary Medicine of the university of Leipzig for the fruitful discussions, their help and the highly professional collaboration. Besides this, I want to thank Dr. Sascha Schott and Dr. Silke Engl from Veyx Pharma GmbH for always being ready answering any questions. I want to thank Dr. Andrej Klikno and Sandra Stange for the performance of HPLC measurements for the stability study. Also, the Deutsche Bundesumweltstiftung should be acknowledged for funding environmentally relevant topics, such as the "Depot-Zyklo"-project. I enjoyed the annual meeting with very interesting topics and presentations, as well as the outstanding food.

Many thanks to Rosa Stark and Catherine Poser, PhD students from the faculty of Veterinary Medicine for in vivo investigations on swine and for taking me into the pig stable (and for providing me a shower afterwards as well). Furthermore, also Dr. Haukur Lindbergh Sigmarsson should be acknowledged for helping me with my investigations on carcasses in the slaughterhouse.

I want to thank Katrin Mayer and Kay Uwe Kraft for their excellent work during their Bachelor and Master Thesis.

Furthermore, I want to thank Christian Minke for his help with SEM measurements and Ilona Vollrath and Wolfgang Wünschheim for her assistance during XRPD measurements.

A big "Thank you" goes to my former colleagues from the LMU, which welcomed me so heartily and made the PhD time one of the best in my life. You have become so much more than just colleagues! I will miss our weekly basketball, funny weekend-trips, the preparation of macarons or just seeing your friendly faces.

I want to thank Kay, Laura & Marie-Paule for your friendship and always being there for me! My lab mates Ellen, Bifeng and Madeleine - I enjoyed our time in the lab so much. Alice, Micha, Corinna, Randy, Elisa, Roman, Steffi and Matze - you made my time unforgettable.

Finally, I want to express my tremendous gratitude to my parents Rainer and Ingrid and to my sister Maren. Thank you a lot for everything you made possible for me, your support and love and an open ear for every tiny problem. Especially to my sister for her encouraging words and her numerous visits.

Last but not least, thank you Christian for your patience, support, never-ending encouragement and for your love.

TABLE OF CONTENT

PUBLICATIO	I DNS ARISING FROM THIS THESIS
LIST OF ABI	BREVIATIONSII
LIST OF FIG	URES IV
LIST OF TAE	BLES VIII
1 Gener	AL INTRODUCTION1
1.1 Int	roduction2
1.2 Dr	ug Delivery Systems for the Sustained Release of Peptides and Proteins3
1.2.1	Implants
1.2.2	Suspensions4
1.2.3	Nanoparticles
1.2.4	Microparticles
1.3 Pre	eparation Techniques for Lipid-Based Microparticles5
1.3.1	Hot emulsion
1.3.2	Double emulsion
1.3.3	Solvent evaporation
1.3.4	Spray drying
1.3.5	Spray congealing
1.4 Est	trus Synchronization7
1.4.1	Hormonal regulation mechanisms
1.4.2	Therapeutic regimen and pharmaceutical products
1.5 Pro	oblems arising from Estrus Synchronization10
1.6 Air	n of the Thesis10
1.7 Ref	ferences12
2 DEVELO	OPMENT OF A SPRAY CONGEALING PROCESS FOR TRIGLYCERIDE
MICROPART	TICLES
3 1 Ak	
2.1 AD	stract
2.2 Int	roduction
2.3 Ma	25 And Methods
2.3.1	Materials
2.3.2	Preparation of triglyceride microparticles by spray congealing
2.3.3	Determination of yield

2.3	.4 Evaluation of the product flow using different sample containers
2.3	.5 Determination of particle size
2.3	.6 Investigations on polymorphic behavior using differential scanning calorimetry (DSC)
2.3	.7 Characterization of particle morphology using scanning electron microscopy (SEM)
2.4	Results and Discussion27
2.4	.1 Evaluation of the product flow into the nozzle using different sample containers
2.4	.2 Preparation of drug-free lipid microparticles by spray congealing
2.4	.3 Evaluation of polymorphic behavior
2.5	Conclusion
2.6	References
2 INC	CORRORATION OF HYDROBULLIC MODEL SUBSTANCES INTO SPRAY CONCEALED
5 INC	UNPORTION OF HYDROPHILIC MODEL SUBSTANCES INTO SPRAY CONGEALED
SOLIDI	JPID MICROPARTICLES
3.1	Abstract
3.2	Introduction
3.3	Materials and Methods
3.3	.1 Materials
3.3	.2 Lyophilization of ASP40
3.3	.3 Cryogenic ball-milling of ASP40
3.3	.4 Mortar grinding of TRP for suspension in the lipid melt
3.3	.5 Preparation of model substance-loaded microparticles by spray congealing40
3.3	.6 Determination of particle size
3.3	.7 Characterization of particle morphology using scanning electron microscopy (SEM)41
3.3	.8 Determination of encapsulation efficiency
3.3	.9 Development of a suitable release model and evaluation of TRP and ASP release behavior
3.4	Results and Discussion42
3.4	.1 Preparation and characterization of model drug-loaded lipid microparticles by spray congealing
3.4	.2 In vitro release behavior of TRP-loaded microparticles
3.4	.3 Influence of lipase addition to the release medium on TRP release
3.4	.4 Evaluation of lipid composition and additive concentration on in vitro release behavior of ASP from lipid-based microparticles
3.5	Conclusion
3.6	References

4 0	PTIM	IZATION OF THE SPRAY CONGEALING PROCESS WITH FOCUS ON YIELD AN	DRUG
LOAD	•••••		55
4.1	Ab	stract	
4.2	Int	troduction	
4.3	Ma	aterials and Methods	
4	.3.1	Materials	
4	.3.2	Cryogenic milling of ASP	
4	.3.3	Preparation of triglyceride-based microparticles by spray congealing	57
4	.3.4	Determination of material distribution in the spray dryer	
4	.3.5	Determination of particle size	58
4	.3.6	Determination of encapsulation efficiency	58
4	.3.7	Determination of lipid-melt viscosities	
4.4	Re	sults and Discussion	59
4	.4.1	Viscosity of the lipid melt	
4	.4.2	Yield distribution and material loss during the spray congealing process	61
4	.4.3	Investigations on particle size of drug-loaded particles	64
4	.4.4	Evaluation of drug distribution and encapsulation efficiency	66
4.5	Co	nclusion	68
4.6	Re	ferences	69
5 Ir	VCODI	POPATION OF CONADOPELIN [6-D-PHF] INTO SPRAV CONCEALED SOL	חופו I חו
MICR	OPAR	FIGUES AND EVALUATION OF RELEASE MECHANISMS	72
-			
5.1	Ab	stract	73
5.2	Int	troduction	74
5.3	Ma	aterials and Methods	76
5	.3.1	Materials	76
5	.3.2	Cryogenic milling of G [6-D-Phe]	76
5	.3.3	Preparation of G [6-D-Phe]-loaded microparticles by spray congealing	76
5	.3.4	Determination of particle size	77
5	.3.5	Investigations on polymorphic behavior using differential scanning cale (DSC)	orimetry 77
5	.3.6	X-Ray powder diffraction (XRPD) of lipid microparticles	77
5	.3.7	Karl-Fischer titration	78
5	.3.8	Characterization of particle morphology using scanning electron mic (SEM)	croscopy 78
5	.3.9	Contact-angle measurements for prediction of lipid surface wettability	78
5	.3.10	Determination of drug-content using RP-HPLC after extraction	78

	5.3.	11	In vitro release behavior of G [6-D-Phe] from lipid microparticles79
	5.3.	12	Preparation of poloxamer gels
	5.3.	13	Temperature-dependent viscosity measurements using a rotation viscometer79
5.	.4	Res	ults and Discussion
	5.4.	1	Preparation of drug-loaded lipid microparticles by spray congealing80
	5.4.	2	Characterization of polymorphic behavior using DSC and XPRD
	5.4.	3	Dependency of water uptake of G [6-D-Phe]-loaded microparticles from the lipid composition
	5.4.	4	Impact of lipid-composition on contact angle and surface wettability87
	5.4.	5	In vitro release behavior of G [6-D-Phe] from lipid microparticles
	5.4.	6	G [6-D-Phe] release from lipid-based microparticles suspended in thermosensitive poloxamer gels
5.	.5	Cor	clusion101
5.	.6	Ref	erences
6	Ex	Viv	O PRE-EVALUATION OF PARTICLE- AND MODEL-DRUG DISTRIBUTION IN
Сар	RCAS	SES	109
UM			
6.	.1	Abs	stract110
6.	.2	Inti	roduction110
6.	.3	Ma	terials and Methods111
	6.3.	1	Materials
	6.3.	2	Preparation of dye loaded lipid microparticles by spray congealing111
	6.3.	3	Scanning Electron Microscopy112
	6.3.	4	Preparation of the reconstitution medium
	6.3.	5	Application of microparticles
6.	.4	Res	ults and Discussion113
6.	.5	Cor	nclusion
6.	.6	Ref	Ferences116
7	In V	VIVO	EVALUATION OF G [6-D-PHE]-LOADED SOLID LIPID MICROPARTICLES WITH
Spe	CIAL	REG	GARD ON ESTRUS CONTROL
7.	.1	Abs	stract
7.	.2	Inti	roduction120
7.	.3	Ma	terials and Methods122
	7.3.	1	Materials
	7.3.	2	Animals
	7.3.	3	Preparation of G [6-D-Phe] loaded microparticles by spray congealing
	7.3.	4	Determination of drug content

1.3.	.5	Aliquoting of the particles under laminar air flow	123
7.3.	.6	Preparation of the reconstitution medium	123
7.3.	.7	Viscosity measurements	124
7.3.	.8	General study set-up and pre-treatment of the animals	124
7.3.	.9	Reconstitution and application of the suspension	125
7.3.	.10	General examination and evaluation of injection sites	126
7.3.	.11	Determination of estrus control	126
7.3.	.12	Ultrasound examination of ovaries	126
7.3.	.13	Statistical Analysis	126
7.4	Res	ults and Discussion	127
7.4.	.1	Characterization of reconstitution behavior	127
7.4.	.2	Applicability of the final suspensions in vivo	128
7.4.	.3	Biocompatibility and documentation of adverse drug reactions	129
7.4.	.4	Effects on cycle blockage in vivo using different formulations and G [6 concentrations	-D-Phe] 130
7.4.	.5	Influence on follicular growth	131
7.4.	.6	Influence of treatment on ovulatory behavior	132
7.4.	.7	Evaluation of the relation between treatment and the occurrence of tempor permanent cysts	rary and 134
7.5	Cor	clusion	135
7.6	Ref	erences	137
8 Lo	NG-T	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES	141
8 Loi 8.1	NG-T Abs	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES	141 142
8 Lor 8.1 8.2	NG-T Abs Inti	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES tract oduction	141 142 142
8 Loi 8.1 8.2 8.3	NG-T Abs Inti Ma	TERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES tract oduction terials and Methods	141 142 142 143
8 Lor 8.1 8.2 8.3	NG-T Abs Inti Ma	TERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES otract oduction terials and Methods Materials	141 142 142 143 143
8 Lor 8.1 8.2 8.3 8.3 8.3	NG-T Abs Inti Ma .1 .2	Term Stability Study of G [6-D-PHE]-LOADED MICROPARTICLES Stract Stract Soduction Toduction terials and Methods Materials Production of Gonadorelin [6-D-Phe]-loaded microparticles	141 142 142 143 143
8 Loi 8.1 8.2 8.3 8.3 8.3 8.3	NG-T Abs Inti Ma .1 .2 .3	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES stract oduction	141 142 142 143 143 143 143
8 Loi 8.1 8.2 8.3 8.3 8.3 8.3	NG-T Abs Inti Mat .1 .2 .3 .4	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES stract	141 142 142 143 143 143 143 143 144
8 Loi 8.1 8.2 8.3 8.3 8.3 8.3 8.3 8.3	NG-T Abs Inti Ma .1 .2 .3 .4 .5	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES tract	141 142 142 143 143 143 143 143 144 EM)144
8 Lor 8.1 8.2 8.3 8.3 8.3 8.3 8.3 8.3 8.3	NG-T Abs Inti Mat .1 .2 .3 .4 .5 .6	Term STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES intract	141 142 142 143 143 143 143 143 144 EM)144 144
 8 Lor 8.1 8.2 8.3 	NG-T Abs Inti Ma .1 .2 .3 .4 .5 .6 .7	Term STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES itract	141 142 142 143 143 143 143 143 144 EM)144 144
 8 Lor 8.1 8.2 8.3 	NG-T Abs Intr Ma .1 .2 .3 .4 .5 .6 .7 .8	erm Stability Study of G [6-D-Phe]-Loaded Microparticles tract	141 142 142 143 143 143 143 143 144 EM)144 144 144 orimetry 144
 8 Lor 8.1 8.2 8.3 	NG-T Abs Inti Ma .1 .2 .3 .4 .5 .6 .7 .8	erm Stability Study of G [6-D-Phe]-Loaded Microparticles stract	141 142 142 143 143 143 143 143 144 EM)144 144 orimetry 144
 8 Lor 8.1 8.2 8.3 	NG-T Abs Intr Ma .1 .2 .3 .4 .5 .6 .7 .8 .9 .10	ERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES ttract	141 142 142 143 143 143 143 144 EM)144 144 144 144 144 144 144

8.4	Res	sults and Discussion1	46
8.4	.1	Influence of storage time and -conditions on particle morphology, -size a polymorphic behavior	and 46
8.4	.2	Drug content and release behavior as a function of storage time and temperature.	ıre 54
8.5	Co	nclusion1	57
8.6	Ref	ferences1	58
9 Su	MMA	RY AND FUTURE WORK1	61
9.1	Ref	ferences1	65

PUBLICATIONS ARISING FROM THIS THESIS

Poster presentations

<u>K. Hoffmann</u>, W. Zaremba, J. Kauffold, W. Frieß "Influence of Additives on Gonadorelin [6-D-Phe] Release from Spray Congealed Lipid Microparticles"

CRS Annual Meeting and Exposition, Seattle, WA, USA, July 17-20, 2016

Oral presentations

<u>K. Hoffmann</u>, W. Zaremba, J. Kauffold, W. Frieß "Release and Long-Term-Stability of Gonadorelin [6-D-Phe]-loaded lipid microparticles produced by spray-congealing

PBP Worldmeeting, Glasgow, United Kingdom, April 4-7, 2016

<u>K. Hoffmann</u>, W. Zaremba, J. Kauffold, W. Frieß "Influence of Monoglycerides on Release of Gonadorelin [6-D-Phe] from Spray-Congealed Solid Lipid Microparticles

CRS Germany Local Chapter Meeting, Muttenz, Switzerland, February 12 - 13, 2015

Patents

W. Zaremba, J. Kauffold, W. Frieß, Y. Yordanova, K. Hoffmann "Veterinary pharmaceutical composition and use thereof "

Application number/Filing date: WO2015EP58265 20150416

LIST OF ABBREVIATIONS

AI	Artificial insemination
AE	Adverse events
ASP	Aspartame
β-hCG	Beta human Chorionic Gonadotropin
BSA	Bovine Serum Albumin
BST, rBGH	Bovine somatotropin, recombinant Bovine
	Growth Hormone
CMC	Carboxymethylcellulose
CSA	Cetylstearyl alcohol
D112, D114, D116, D118, D120	Dynasan112, Dynasan114, Dynasan116,
	Dynasan118, Dynasan120
DDS	Drug Delivery System
DLG	Deutsche Landwirtschaftliche Gesellschaft
DSC	Differential Scanning Calorimetry
eCG	Equine Chorionic Gonadotropin
EE	Encapsulation Efficiency
EPO	Erythropoietin
FDA	Food and Drug Administration
FIV	Feline Immunodeficiency Virus
FSH	Follicle stimulating hormone
G [6-D-Phe]	Gonadorelin [6-D-Phe]
GLP-1	Glucagon-like peptide-1
GMS	Glycerol monostearate
GnRH	Gonadotropin releasing hormone
eCG	Equine Chorionic Gonadotropin
HBsAg	Hepatitis B surface Antigen
HCl	Hydrochloride
IFN-α	Interferone alpha
i.m.	intra muscular
LAF	Laminar Air Flow

LH	Luteinizing hormone
O/W	Oil-in-Water
PBS	Phosphate buffered saline
PEG	Polyethylene glycol
PEG-24-GS	Polyethylene glycol-24-glyceryl stearate
PGF-2α	Prostaglandin F-2 alpha
PLA	Poly lactic acid
PLGA	Poly (lactic-co-glycolic acid)
PMSG	Pregnant mare serum gonadotropin
POD	Polycystic ovarian degeneration
PS	Polysorbate
PVP	Polyvinylpyrrolidone
rhEPO	Recombinant human erythropoietin
RP-HPLC	Reversed-phase high-performance liquid
	chromatography
s.c.	1
	subcutaneous
SEM	Scanning electron microscopy
SEM SLI	Scanning electron microscopy Solid lipid implants
SEM SLI SLM	Scanning electron microscopy Solid lipid implants Solid lipid microparticles
SEM SLI SLM SLN	Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles
SEM SLI SLM SLN TG	Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride
SEM SLI SLM SLN TG TP-4	Subcutaneous Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride Thymocartin
SEM SLI SLM SLN TG TP-4 t-PA	Subcutaneous Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride Thymocartin Tissue-type Plasminogen Activator
SEM SLI SLM SLN TG TP-4 t-PA TRP	Subcutaneous Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride Thymocartin Tissue-type Plasminogen Activator Tryptophan
SEM SLI SLM SLN TG TP-4 t-PA TRP TSPG	Subcutaneous Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride Thymocartin Tissue-type Plasminogen Activator Tryptophan Thermosensitive poloxamer gels
SEM SLI SLM SLN TG TP-4 t-PA TRP TSPG VEGF	Subcutaneous Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride Thymocartin Tissue-type Plasminogen Activator Tryptophan Thermosensitive poloxamer gels Vascular endothelial growth factor
SEM SLI SLM SLN TG TP-4 t-PA TRP TRP TSPG VEGF WPN	Subcutaneous Scanning electron microscopy Solid lipid implants Solid lipid microparticles Solid lipid nanoparticles Triglyceride Thymocartin Tissue-type Plasminogen Activator Tryptophan Thermosensitive poloxamer gels Vascular endothelial growth factor Wide pneumatic nozzle

LIST OF FIGURES

Figure 1-1: Schematic image of the spray congealing equipment based on the Büchi B-290 Mini spray dryer [1]
Figure 2-1: Schematic depiction of the spray congealing equipment additionally installed on top of a B-290 Mini spray dryer
Figure 2-2 : Product flow out of the sample container for D116 at 90 °C (mean and SD, n = 3)
Figure 2-3: SEM images of triglyceride microparticles based on D116 (A), D118 (B) and D120 (C)
Figure 2-4 : Polymorphic behavior of freshly prepared D116 microparticles (A) and after 72 h stored at room temperature (B), after 72 h at 7 °C (C), after 24 h at 45 °C (C), after 48 h at 45 °C (D) and 72 h at 45 °C (E). Representative heating scans are depicted of the first and second heating each. Curves are displaced along the ordinate for better visualization31
Figure 3-1: SEM images of TRP-loaded microparticles (D116: A, D, D118 + 1 % sucrose: B, E, D118 + 5 % GMS: C, F)43
Figure 3-2 : Release of TRP from D116 (A) and D118 (B) microparticles without and with 5 or 10 % GMS in PBS pH 7.4 (mean and SD, n = 3)
Figure 3-3 : Release of TRP from D116 (A) and D118 (B) microparticles with 0, 5 and 10 % GMS in PBS pH 7.4 containing 40 U/ml lipase (mean and SD, n = 3)
Figure 3-4 : D118 + 5 % GMS-based microparticles after 3-weeks incubation in PBS pH 7.4 without lipase (A) and with 40 U/ml lipase (B)
Figure 3-5 : Release of ASP from D114 and D116 microparticles blended with emulsifying additives and/or low melting triglycerides in PBS pH 7.4 (mean and SD, n = 3)49
Figure 4-1 : Schematic plan of the spray dryer. Particles were sampled at (1) sample container, (2) spray tower, (3) spray tower vessel, (4) connection, (5) high-performance cyclone, (6) product container and (7) filter
Figure 4-2 : Viscosity of drug-free lipid melts (D116 and D114) with emulsifiers at 90 °C and 75 °C at a shear rate of 1 s ⁻¹ (Mean and SD, $n = 2$)60
Figure 4-3 : Material distribution after spray congealing experiments using standard conditions (spray flow 401 l/h, aspirator 100 %, 90 °C) (A), reduced aspirator power of 70 % (B), reduced spray flow of 301 l/h (C), increased spray flow of 601 l/h (D), melt temperature 75 °C (E) and installation of an additional stirrer (F). "Others" includes filter, high-performance cyclone and connection tube between tower and high-performance cyclone62
Figure 4-4 : Mean particle size of samples collected in the product container (A), spray tower

Figure 4-5 : Drug loads of collected samples from the product container (A), spray tower (B) and spray tower vessel (C). Standard settings were defined as spray flow 414 l/h, spray pressure 6 bar, aspirator 100 %, melt temperature 90 °C (black arrows)
Figure 5-1 : DSC thermograms of G [6-D-Phe]-loaded microparticles (n = 2). Thermograms represent the first heating scan of one sample exemplarily and were shifted on the ordinate for better visualization
Figure 5-2 : Representative DSC thermograms of G [6-D-Phe]-loaded triglyceride microparticles before and after incubation (4 h, 24 h, n = 3) in PBS buffer pH 7.4 at 39 °C. Scans were shifted on the ordinate for better visualization
Figure 5-3 . Representative XRPD measurements of lipid raw materials and spray congealed lipid microparticles (n = 3). Diffraction patterns were shifted on the ordinate for better visualization
Figure 5-4: Water content of spray congealed microparticles consisting of D114 with different emulsifying additives (mean and SD, $n = 3$)
Figure 5-5 : Water uptake of spray congealed lipid microparticles after incubation in PBS at 39 °C determined by Karl-Fischer titration (mean and SD, $n = 3$)
Figure 5-6 : Contact angle of highly purified water on triglycerides (D114, D116, D118, D120) and GMS (mean and SD, n = 3)
Figure 5-7: Contact angle of highly purified water on D114-emulsifier mixtures (mean and SD, n = 3)
Figure 5-8: Contact angle of highly purified water on D116-emulsifier mixtures (mean and SD, n = 3)
Figure 5-9 : G [6-D-Phe] release from triglyceride microparticles (D116 solid line, D118 dashed line; mean and SD, n = 3)90
Figure 5-10 : G [6-D-Phe] release from triglyceride microparticles with varied amounts of CSA (10, 20, 40 %) (A). G [6-D-Phe] release from D116 (B) and D118 (C) microparticles with varied concentrations of GMS (10 %, 20 %, 40 %, 60 %), mean and SD, n = 391
Figure 5-11 : SEM images of freshly prepared microparticles of D118 + 10 % GMS (left) and after a 4-week incubation in PBS at 39 °C (right)
Figure 5-12 : G [6-D-Phe] release from D114 microparticles with addition of Span 40 and GMS (A) and Span 80 and 85 (B), D116 microparticles with addition of Span 85 (C) and Span 80 (D) and with addition of D112/GMS and D112 (E), mean and SD, n = 393
Figure 5-13 : SEM images of freshly prepared (left) and incubated D114 and D116-based particles with addition of Span 40, 60, 80, 85, GMS and D112 for 2 weeks in PBS buffer at 39 °C (right)
Figure 5-14 : Identified candidates for clinical studies. (A) Suitable microparticle batches based on D116 and D118 with 10 % GMS applied in pre-clinical study I (September 2014). (B) Formulations prepared of D114 and D116 with different amounts of Span 40 and GMS used in pre-clinical study II (August 2015)

Figure 7-6: O	bserved ovulation percentage after treatment with lipid-based micropar	ticles in
the first stu	dy. Each bar represents the percentage out of 5 animals per group. St	tatistical
analysis wa	s performed using the one-way ANOVA test followed by a Holm-Sidak	pairwise
comparison		133

Figure 7-8 : Percentage of observed cysts (permanent: black bars, temporary: grey bars) during the treatment with G [6-D-Phe]-loaded lipid based microparticles monitored in the first preclinical study (A) and the follow-up study (B). Each bar represents the percentage out of the population of 5 animals per group. Statistical analysis was performed using the Kruskal-Wallis one-way ANOVA test on ranks
Figure 8-1: Appearance of microparticles directly after production
Figure 8-2: Appearance of microparticles after 1 M storage at 2-8, 25 and 40 °C147
Figure 8-3: Appearance of microparticles after 3 M storage at 2-8, 25 and 40 °C148
Figure 8-4 : Mean particle size measurement using laser diffraction over a 12-month storage time at 2-8, 25 and 40 °C (Mean and SD, n = 3, left); Reconstitution of 2-8 °C-stored samples after 6 months exemplarily for acceptable reconstitution properties (right)149
Figure 8-5: Appearance of microparticles after 6 M storage at 2-8, 25 and 40 °C150

Figure 8-6: Appearance of microparticles after 12 M storage at 2-8, 25 and 40 °C151

Figure 8-7: Representative DSC thermograms of microparticles stored at (A) 2-8 °C; (B) 25 °C; (C) 40 °C (n = 2); sampling was performed after 0, 1, 3, 6, 9 and 12 months 152

Figure 8-8: Representative XRPD measurement of the 12 M-stored samples at 2-8, 25 and 40 °C. Single graphs were displaced along the ordinate for better visualization (n = 2)...153

Figure 8-9 : G [6-D-Phe]	content in D118 + 10 $\%$	GMS micropartic	cles during a 12-month
storage time at 2-8, 25	and 40 °C (mean and SD, 1	n = 3)	

LIST OF TABLES

Table 1-1: Structural changes in amino acid sequence of GnRH analogues compared to GnRH
Table 2-1 : Overview of placebo microparticle batches produced with varied apparatus settings.
Table 3-1 : Overview of TRP-loaded microparticle batches sprayed with different apparatus settings and with addition of GMS and sucrose
Table 3-2 : Overview of 1.67 % ASP-loaded microparticle batches (s = solid, l = liquid, $ss = semi-solid$); Three emulsifies with different HLB values were investigated concerningparticle size exemplarily (data shown mean and SD, n = 3)
Table 5-1: First generation G [6-D-Phe]-loaded formulations produced by spray congealing
Table 5-2: Second generation G [6-D-Phe]-loaded formulations produced by spray congealing
Table 7-1 : Overview of applied treatments and G [6-D-Phe] concentrations per group. Groups1-5 were investigated during the first pre-clinical study in September 2014, groups 6-10 wereinvestigated in August 2015 in the second study
Table 7-2: Viscosities of reconstitution media and final suspensions 127

GENERAL INTRODUCTION

1.1 Introduction

Biopharmaceutical drugs have gained more and more importance in the field of pharmaceutical research [1] [2]. Being able to cure severe and rare diseases, the development of pharmaceutical products for the delivery of peptides and proteins became an important task in recent years [3] [4]. Challenges within the development, manufacturing, production and storage of these products are, that especially proteins are sensitive to a variety of external factors, like shear stress, temperature and pH [5] [6]. Due to poor oral absorption, peptide and protein drugs are typically administered parenterally. Furthermore a short half-life in the body causes a high frequency of administration via the parenteral route, which negatively impacts the patient compliance [2]. For this reason, the development of sustained release delivery systems is of high importance to achieve sustained blood levels with reduced side effects and improved patient acceptance [7].

In the field of veterinary medicine, the focus is set on facilitating therapy for owners and veterinarians and to reduce distress and potential side effects for the treated animal [8]. For livestock breeding, the application of biopharmaceutical drugs opens new doors with respect to wait-times and residuals in animal products. With the application of biopharmaceutical drugs in a sustained release delivery system, their main drawback, the short half-life can be overcome [9]. As strict rules for wait times concerning milk and meat exist for small molecules due to accumulation, this problem is not given for peptides, e.g. G [6-D-Phe] in Gonvet[®]Veyx (Veyx Pharma GmbH) [10]. Consequently, sustained release formulations entered the marketplace in veterinary medicine. ProHeart[®] 6 by Zoetis, an injectable sustained delivery system for protection against the canine heartworm disease, based on lipid microparticles is FDA approved [11]-[13]. Another example is the application of the GnRH-agonist deslorelin via the subcutaneously injected lipid implant Suprelorin[®] by Virbac for the chemical castration in male, mature dogs [14] [15]. Suprelorin[®] was recently also investigated for taint removal of boars and for suppression of sexual function in male cats [16] [17]. Bovine somatotropin (recombinant bovine growth hormone, BST, rBGH) is the active ingredient formulated in an oily zinc suspension in Posilac[®] (Elanco Animal Health). The controlled release induces an improved lactation in dairy cows [8] [18] [19]. For livestock animals, the requirements are to provide an acceptable and effective therapy with low costs, facilitating the already existing treatment.

1.2 Drug Delivery Systems for the Sustained Release of Peptides and Proteins

A variety of drug delivery systems can be utilized to sustain the release of peptides and proteins.

1.2.1 Implants

Implants usually have a cylindrical shape and can be manufactured by compression, molding or extrusion [20]–[22]. A distinction between biodegradable and non-biodegradable implants is commonly made in literature. Advantages of biodegradable implants are the lack of surgical removal and the high biocompatibility [2] [23] [24]. Release is mainly controlled by diffusion, which takes place in both degradable and non-degradable implants. Furthermore, for biodegradable implants also matrix erosion, swelling and osmotic effects need to be taken into consideration [25] [26].

Poly (lactide-co-glycolide) (PLGA) derived implants

The delivery of proteins via PLGA-based implants was extensively studied, revealing outstanding biocompatibility and degradation properties [27]–[29]. A disadvantage concerning implants based on PLGA is the formation of an acidic microenvironment within the implant, leading to an enhanced degradation of the protein and potentially covalently modified peptide and protein [27] [30] [31]. Nevertheless, PLGA-based implants delivering the peptide drugs goserelin (Zoladex[®], Astra Zeneca) [32] and buserelin (Profact[®] Depot, Sanofi Aventis Deutschland GmbH) [33], as well as an in-situ forming implant containing leuprolide (Eligard[®], Astellas) [34] are approved for breast and prostate cancer treatment, whereas Zoladex[®] is further approved for the treatment of severe endometriosis [32] [35].

Solid lipid implants

The manufacturing of biodegradable implants made from solid lipids, like triglycerides, also in combination with monoglycerides and cholesterol for the delivery of therapeutic proteins is of high interest in the field of controlled delivery [22] [26] [36] [37]. An advantageous production system is the twin-screw extrusion process, allowing to treat the lipid excipient slightly below the melting point which helps to avoid polymorphic changes [22] [37] [38]. Drug release can be managed in time frames from weeks to months [39] [40]. The addition of additives like polyethylene glycol (PEG) can enhance and complete the release, due to the formation of an interconnected pore network [26] [40] [41].

1.2.2 Suspensions

Oily suspensions are suitable delivery systems for peptides and proteins intended for controlled release [2]. The sustained delivery is due to the formation of a depot after injection in the tissue, which slowly releases the drug [8]. In human as well as veterinary medicine oily suspensions are of high interest. In veterinary medicine, e.g. the products Excede[®] and Excenel[®] RTU by Zoetis containing ceftiofur and its hydrochloride, a cephalosporin against infectious diseases in cattle, is applied via an oily suspension [8] [42] [43].

1.2.3 Nanoparticles

Nanoparticles ranging from 10 to 1000 nm can e.g. be produced by spray drying and emulsionbased preparation methods, like hot or cold homogenization techniques [44]–[47]. Natural or synthetic polymers, like gelatin, PLGA, or triglycerides can be used for their preparation [46] [48]–[51]. Triglycerides, like tristearin, trimyristin or composed lipids like Compritol[®] 888 ATO are commonly used as lipid bases upon manufacturing [52]–[54]. Regarding biopharmaceutical drugs, proteins like BSA and peptides like calcitonin and gonadorelin have been entrapped successfully into nanoparticles [55]–[58].

1.2.4 Microparticles

Microparticles from natural and synthetic polymers, for example solid lipids [1] [59] [60], PLGA [61]–[63] and poly lactic acid (PLA) [64], are suitable for parenteral drug delivery. Various microparticulate sustained release systems are described for biopharmaceuticals [1] [65]–[68].

Poly lactic acid (PLA) and Poly lactic co-glycolic acid (PLGA) microparticles

PLGA is described as suitable polymer for the preparation of microparticles, also for protein and peptide delivery [24]. Recombinant human erythropoietin (rhEPO), vascular endothelial growth factor (VEGF) and the peptide somatostatin could be successfully entrapped into PLA and PLGA-based particles [3] [69] [70]. Problems arising from PLGA are changes in the microclimate which afford a special stabilization of the encapsulated proteins [24] [71] and the high costs which explain the only few available veterinary products. The drug delivery system SMARTshot[®] B₁₂ Prime Lamb by Virbac delivers vitamin B₁₂ in a PLGA suspension in an oily vehicle for suckling lambs [8] [72]. For diabetes treatment, the PLGA-based microparticle system Bydureon[®] by AstraZeneca is approved delivering exenatide [73].

Solid lipid microparticles

Solid lipid microparticles are of particular interest for the delivery of biopharmaceuticals due to several advantages compared to other polymers like PLGA [1] [67] [74]–[76]. The solid lipids are highly biocompatible and degraded by endogenous lipases [5] [23] [77]. A pH shift, which may destabilize peptides and proteins, cannot be observed [78]. The peptides somatostatin, insulin, thymocartin, desmopressin and the GnRH antagonist antide have been loaded into solid lipid microparticles [1] [68] [74]–[76]. The sustained delivery of proteins like BSA has also been realized with incorporation into lipid microparticles [79] [80].

1.3 Preparation Techniques for Lipid-Based Microparticles

Lipid-based microparticles are often produced by emulsion-based methods [81]–[83]. Furthermore, preparation techniques with presence of organic solvents are common [75] [76]. Ideally, the manufacturing technique works without an additional water phase, which may induce degradation or aggregation processes at the interface, and without organic solvents to avoid the problem of solvent residuals in lipid microparticles.

1.3.1 Hot emulsion

This method is preferably applied for preparation of small molecule-loaded lipid microparticles. Lipid excipients are melted together and mixed with a hot aqueous phase containing emulsifier to obtain a O/W emulsion, potentially via phase inversion. Upon cooling, the lipid droplets solidify which leads to the formation of microparticles [84] [85].

1.3.2 **Double emulsion**

In the double emulsion method, the drug is dissolved in an aqueous phase, which is afterwards emulsified into a lipid melt. Often stabilizers such as gelatin or poloxamers are used. Subsequently, this primary emulsion is transferred into a second aqueous phase. Upon stirring and cooling, the microparticles solidify and can be removed by filtration [81] [86].

1.3.3 Solvent evaporation

The solvent evaporation method was also reported as suitable for the encapsulation of peptides into microparticles [68] [74]. The underlying principle is, that the solid triglyceride is dissolved in methylene chloride or another organic solvent and the grinded peptide afterwards added. The suspension is subsequently added to an aqueous solution, which might be cooled, and stirred to allow the evaporation of the organic solvent leading to the formation of the microparticles [74]. Another technique deriving from this process is the so called solvent stripping method used for the entrapment of the GnRH antagonist antide [76]. Other researchers also successfully

incorporated hepatitis B surface antigen (HbsAg) and somatostatin via a modified emulsionsolvent evaporation method into lipid microparticles [67] [75].

1.3.4 Spray drying

Spray drying is an efficient drying process in a preheated airstream. Although the contact of the liquid phase with the hot air is quite short, this drying technique leads to a considerably higher thermal stress than freeze-drying [87]–[89]. For this reason, the co-current preparation method should be favored compared to the counter current method, as the air of the lowest temperature is getting in contact with the drug solution, e.g. as described for recombinant methionyl human growth hormone (hGH) and tissue-type plasminogen activator (t-PA) [87] [88]. To achieve a sustained release, atomization can either occur out of an organic solution, an emulsion or a suspension, whereas the removal of one phase commonly leads to the formation of microparticles [3] [90]–[92]. Used lipids for the spray drying process are usually triglycerides also in combination with emulsifiers like lecithin [91]. But also the use of other polymers like PLGA [90] [61], PLA [64], cellulose derivatives [93] or chitosan [94] is described in literature. Thus, this preparation method is used for the preparation of microparticles loaded with biopharmaceutical drugs, like erythropoietin and insulin [3] [95].

1.3.5 Spray congealing

Spray congealing is related from spray drying using a similar apparatus setup with installation of additional parts (*Figure 1-1*) [1]. The lipid melt with suspended peptide or protein particles, is sprayed into an air stream, which is either kept at room temperature or cooled down [1] [96]–[99]. Atomization typically occurs with the use of a two-fluid nozzle whereas compressed air is commonly used to disperse the melt [100] [101]. Furthermore, the use of atomization via ultrasound [102]–[105] and rotating discs or wheels [96] [106] [107] is described. The contact with the air of lower temperature leads to solidification of the produced droplets. The separation in different size fractions occurs in analogy to the spray drying process [108].



Figure 1-1: Schematic image of the spray congealing equipment based on the Büchi B-290 Mini spray dryer [1]

In literature, the successful incorporation of a variety of small molecule drugs, like indomethacin [102], verapamil HCl [97] and aminolevulinic acid [101] is described. The encapsulation of proteins via spray congealing may be critical due to thermal stress affecting the ternary and quaternary structure. Nevertheless the peptides insulin [1], somatostatin [75] and thymocartin [74] as well as proteins like BSA were successfully incorporated into spray congealed lipid microparticles [79] [109]. Major benefits of the spray congealing process are, that no further aqueous phase is needed which helps to avoid peptide or protein aggregation [5] [110].

1.4 Estrus Synchronization

The term estrus synchronization describes the simultaneous achievement of the fertile state in female animals by the help of pharmaceuticals [111]–[113]. It is a suitable tool to reduce costs and efforts for veterinarians and farmers, due to termination of separation of the female animals, as well as littering and weaning [114] [115]. Species treated to achieve estrus synchronization are mainly swine, but protocols do also exist for cattle, sheep, goats and mares [115]–[123]. Over decades, the application of synthetic gestagens turned out to be highly efficient in synchronizing the estrus in swine, whereas the treatment is successful for both gilts and sows [111] [113] [117]. Moreover, gestagen treatment leads to absence of non-responders and no severe side effects [113]. These advantages made this therapy to the gold-standard in estrus synchronization in recent years. Drawbacks of this therapy arise from the environmental entry of the gestagens by the produced manure [124] [125]. Analysis of the surrounding runoff to large agricultural holdings evidenced the presence of gestagens in levels above non-harmful concentrations [126] [127]. Gestagens exhibit toxicity for water organisms by leading to reproductive disorders or even infertility, further discussed in *1.5* [128] [129].

1.4.1 Hormonal regulation mechanisms

To understand the mode of action underlying estrus synchronization, the hormonal regulation of sex hormones in female mammalians is briefly discussed. Under control of luteinizing hormone (LH) and follicle stimulating hormone (FSH) progesterone and estrogen are released. LH and FSH release again is controlled by the decapeptide gonadotropin-releasing hormone (GnRH), also referred to as luteinizing hormone-releasing hormone (LHRH), with the sequence p-Glu-His-Trp-Ser-Tyr-Gly-Leu-Arg-Pro-Gly-NH₂ [130]–[133]. A single or pulsatile application of GnRH or its analogues results in an increase in LH and FSH levels. If applied continuously, these hormones are down regulated due to a negative feedback mechanism taking place in the gonadal-pituitary-hypothalamic axis and a following receptor down regulation [132] [133].

1.4.2 Therapeutic regimen and pharmaceutical products

Estrogens and gestagens

Altrenogest, also referred to as allyl trenbolone, is the gestagen mainly used for estrus synchronization in swine. It is usually fed in amounts of 15-20 mg per day over a time period of 14-18 d [111] [113] [117]. Besides, also norgestomet, a gestagen used in goats for estrus synchronization [121]. Melengestrol acetate is used in postpartum beef cattle and heifers in combination with prostaglandin derivatives to synchronize the estrus [134] [135].

GnRH and its analogues

GnRH has a half-life of approximately 1 h in the blood plasma and is released physiologically in a pulsatile manner [136]. GnRH analogues, or so called "superagonists" with a higher affinity to GnRH receptors and a higher stability in vivo were developed [130] [137]. GnRH analogues are widely applied in both human and veterinary medicine for a variety of indications (*Table 1-1*). Gonadorelin [6-D-Phe] is one analogue, reported to have a higher affinity to GnRH receptors [130]. Other related substances are buserelin, goserelin and triptorelin, which are indicated in human medicine for different cancer types, endometriosis and infertility [35]. Recently, goserelin was also tested as add-on to conventional medication to protect the ovarian function against aggressive chemotherapy in breast cancer in pre-menopausal females [138] [139].

Substance	Amino acid sequence	
GnRH	p-Glu-His-Trp-Ser-Tyr-Gly-Leu-Arg-Pro-Gly-NH ₂	[133]
G [6-D-Phe]	p-Glu-His-Trp-Ser-Tyr- D-Phe -Leu-Arg-Pro-Gly-NH ₂	[10]
Triptorelin	p-Glu-His-Trp-Ser-Tyr- D-Trp -Leu-Arg-Pro-Gly-NH ₂	[136]
Buserelin	p-Glu-His-Trp-Ser-Tyr- D-Ser (tBu) -Leu-Arg- Pro(NHEt)	[35]
Goserelin	p-Glu-His-Trp-Ser-Tyr- D-Ser (tBu)- Leu-Arg-Pro(NEt)- AzaGly -NH ₂	[140]
Peforelin	pGlu-His-Trp-Ser-His-Asp-Trp-Lys-Pro-Gly-NH ₂	[141]
Leuprolide	p-Glu-His-Trp-Ser-Tyr- D-Leu -Leu-Arg- Pro(NHEt)	[142]

Table 1-1: Structural changes in amino acid sequence of GnRH analogues compared to GnRH

In veterinary medicine, G [6-D-Phe] (Gonavet[®] Veyx, Veyx Pharma GmbH), goserelin (Fertilan[®], Bremer Pharma) and peforelin (Maprelin[®], Veyx Pharma GmbH) formulated in immediate releasing formulations are commonly used for a successful induction of ovulation in sows and gilts [123] [143] [144]. According to the professional information of Gonavet[®] Veyx, G [6-D-Phe] should be administered approx. 80 h respectively after the administration of PMSG or peforelin to gilts [10]. In sows, the new estrus should be induced with the application of PMSG 24 h after weaning. Depending on the duration of the suckling period, Gonavet[®] Veyx should be administered 56 to 80 h after peforelin or PMSG. Artificial insemination should be performed twice, 24-26 and 40-42 h after application of G [6-D-Phe] [10]. Furthermore, the application of GnRH analogues in a sustained release delivery system was successfully introduced into the market with the implant Suprelorin[®] (Virbac) leading to a chemical castration in male mature dogs as alternative to surgical castration [14] [145].

Beta-human chorionic gonadotropin (β-hCG) and equine chorionic gonadotrophin (eCG)/ pregnant mare serum gonadotropin (PMSG)

 β -hCG, another gonadotropin, can usually be found in blood plasma of pregnant females, responsible for the sustaining of pregnancy [146]. ECG, also called PMSG, is obtained from blood of pregnant mares. It is the active compound in Fertipig[®] (Ceva Santé Animale) and Suigonan[®] (MSD Animal Health) in combination with hCG [147] [148]. Intergonan[®] (MSD Animal Health) contains only eCG [149]. For induction of estrus, hCG is used successfully for treatment of weaned sows [143]. In 1982, von Kaufmann and Holtz already reported the successful treatment of prepubertal gilts coming into estrus by a single injection of hCG and PMSG [150].

Prostaglandin F2-α (PGF2-α)

PGF2- α , a prostaglandin deriving from the metabolism of arachidonic acid, is reported to have positive effects on artificial insemination in swine [151] [152]. Due to local effects on the myometrium, they are able to enhance the success of artificial insemination (AI) by induction of local uterus contractions with positive effects on synchronizing the ovulation [122] [153].

1.5 Problems arising from Estrus Synchronization

Meat and milk consumption is very high in western countries and is continuously increasing in developing countries which makes an effective livestock breeding necessary [154]. Estrus synchronization by gestagens is one element to achieve this. Orlando et al. reviewed problems going hand in hand with high doses of gestagens getting into the ground water, as side products of large agricultural facilities [126]. Collecting samples from a cattle runoff in the US revealed high amounts of gestagens and their metabolites in the manure and dry feedlot surface soil. The authors assume, that excretes of large agricultural holdings play an important role in the environmental entry of steroid hormones reaching levels over the predicted no-effect concentration [127]. High steroid concentrations of up to 375 ng/l were found in the melt water in the neighborhood of large agricultural holdings in Wisconsin [124]. Possible harmful effects on reproduction are reported for the fathead minnow and zebrafish, as well as for alligators and frogs [124] [128] [155]–[158].

1.6 Aim of the Thesis

The present thesis was performed to develop an environmentally friendly alternative for estrus synchronization in swine. Treatment with gestagens is the current gold standard for estrus synchronization. But substantial entry into the environment has been shown and it is likely that gestagens negatively influence reproduction of water organisms [126] [128] [129]. This work should focus on the development of a delivery system for the decapeptide G [6-D-Phe], which is able to guarantee a sustained release over two weeks. The delivery of the peptide via triglyceride microparticles seems to be the best choice for our purpose based on costs and tolerability. The system should guarantee a drug release in an effective dose over a time period of 15 days, followed by a more or less immediate drop to allow the simultaneous onset of a new estrus in all treated animals for a fixed-time insemination program.

Chapter Two describes the first tests to establish a spray congealing process focusing on the impact of melt viscosities and apparatus settings on particle size and yield. Furthermore, lipid microparticles are evaluated for polymorphism.

In Chapter Three the incorporation of model substances, like the amino acid tryptophan and the dipeptide aspartame with focus on yield and drug content is studied. First release experiments from pure triglyceride matrices and in combination with emulsifiers and sugars were performed.

Chapter Four focuses on the optimization of the spray congealing process with respect to yield and encapsulation efficiency. Material distribution during the spray congealing process and evaluation of possible sources for high material loss are studied. Possibilities for the optimization of the process are discussed.

In Chapter Five G [6-D-Phe]-loaded microparticles obtained via the spray congealing process were tested. Factors influencing the release behavior, like surface morphology, water uptake upon incubation and polymorphic behavior are investigated. The wettability of lipid matrices is determined, also in combination with surfactants, by contact angle measurements using the sessile-drop method. The addition of different emulsifier types at various concentrations is screened to achieve a release over two weeks.

Chapter Six shows the results of a particle and drug distribution study in carcasses after application in the lateral neck muscles.

Chapter Seven summarizes the results of the two performed pre-clinical studies. The first study focuses on the dose finding of a future delivery system by application of two different G [6-D-Phe] doses via two different formulations. The second study evaluates four different formulations using only the low dose of the first study. Treated swine are investigated concerning biocompatibility of the tested formulations and possible adverse reactions. Most important is the effect on cycle blockage and onset of follicular growth, as well as determination of ovulation. Furthermore, the side effect of ovulatory cysts is discussed.

Chapter Eight describes the results of a long-term stability study performed over a 12-month time at 2-8, 25 and 40 °C. Microparticles are investigated concerning drug content, peptide stability, particle morphology and size, suspension in the reconstitution medium and release profile. Possible instabilities arising from the storage are explained.

1.7 References

- A. Maschke, C. Becker, D. Eyrich, J. Kiermaier, T. Blunk, and A. Gopferich, "Development of a spray congealing process for the preparation of insulin-loaded lipid microparticles and characterization thereof," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 2, pp. 175–187, 2007.
- [2] M. Rawat, D. Singh, S. Saraf, and S. Saraf, "Lipid carriers: a versatile delivery vehicle for proteins and peptides," *J. Pharm. Soc. Japan*, vol. 128, no. 2, pp. 269–280, 2008.
- [3] B. Bittner, M. Morlock, H. Koll, G. Winter, and T. Kissel, "Recombinant human erythropoietin (rhEPO) loaded poly(lactide-co- glycolide) microspheres: Influence of the encapsulation technique and polymer purity on microsphere characteristics," *Eur. J. Pharm. Biopharm.*, vol. 45, no. 3, pp. 295–305, 1998.
- [4] S. Mohl and G. Winter, "Continuous release of rh-interferon alpha-2a from triglyceride matrices," J. Control. Release, vol. 97, no. 1, pp. 67–78, 2004.
- [5] U. Bilati, E. Allémann, and E. Doelker, "Strategic approaches for overcoming peptide and protein instability within biodegradable nano- and microparticles," *Eur. J. Pharm. Biopharm.*, vol. 59, no. 3, pp. 375–388, 2005.
- [6] W. Jiskoot, T. W. Randolph, D. B. Volkin, C. R. Middaugh, C. Schöneich, G. Winter, W. Friess, D. J. A. Crommelin, and J. F. Carpenter, "Protein instability and immunogenicity: Roadblocks to clinical application of injectable protein delivery systems for sustained release," *J. Pharm. Sci.*, vol. 101, no. 3, pp. 946–954, 2012.
- S. D. Putney and P. A. Burke, "Improving protein therapeutics with sustained-release formulations," *Nat. Biotechnol.*, vol. 16, no. 2, pp. 153–157, 1998.
- [8] N. J. Medlicott, N. A. Waldron, and T. P. Foster, "Sustained release veterinary parenteral products," *Adv. Drug Deliv. Rev.*, vol. 56, no. 10, pp. 1345–1365, 2004.
- [9] M. A. Hoitink, J. H. Beijnen, A. Bult, O. A. van der Houwen, J. Nijholt, and W. J. Underberg, "Degradation kinetics of gonadorelin in aqueous solution," *J. Pharm. Sci.*, vol. 85, no. 10, pp. 1053–1059, 1996.
- [10] W. Zaremba, S. Engl, and A. Wehrend, "Gonavet®Veyx," 2015. [Online]. Available: https://www.veyx.de/uploads/tx_sitscontentattribute/KB_Gonavet_Veyx_DE_150421.pdf. [Accessed: 18-Oct-2016].
- [11] Food and Drug Administration, "ProHeart® 6 (Moxidectin)," 2005. [Online]. Available: http://www.fda.gov/downloads/AdvisoryCommittees/CommitteesMeetingMaterials/VeterinaryMedicine AdvisoryCommittee/UCM205054.pdf. [Accessed: 17-Dec-2016].
- [12] L. T. Glickman, N. W. Glickman, G. E. Moore, R. Cobb, S. A. Connell, M. Morrison, and H. B. Lewis,
 "Safety profile of moxidectin (ProHeart 6) and two oral heartworm preventives in dogs," *Int. J. Appl. Res. Vet. Med.*, vol. 3, no. 2, pp. 49–61, 2005.
- [13] E. Curry-Galvin, "The ProHeart debate," J. Am. Vet. Med. Assoc., vol. 226, no. 8, pp. 1280–1281, 2005.
- [14] C. Maenhoudt, N. Santos, and A. Fontbonne, "Suppression of fertility in adult dogs," *Reprod. Domest. Anim.*, vol. 49, no. SUPPL.2, pp. 58–63, 2014.
- [15] C. A. Herbert and T. E. Trigg, "Applications of GnRH in the control and management of fertility in female animals," *Anim. Reprod. Sci.*, vol. 88, no. 1–2, pp. 141–153, 2005.
- [16] J. Kauffold, H. Rohrmann, J. Boehm, and A. Wehrend, "Effects of long-term treatment with the GnrH

agonist deslorelin (Suprelorin®) on sexual function in boars," *Theriogenology*, vol. 74, no. 5, pp. 733–740, 2010.

- [17] S. Goericke-Pesch, P. Georgiev, A. Antonov, M. Albouy, and A. Wehrend, "Clinical efficacy of a GnRHagonist implant containing 4.7 mg deslorelin, Suprelorin®, regarding suppression of reproductive function in tomcats," *Theriogenology*, vol. 75, no. 5, pp. 803–810, 2011.
- [18] D. E. Bauman, R. W. Everett, W. H. Weiland, and R. J. Collier, "Production responses to bovine somatotropin in northeast dairy herds.," *J. Dairy Sci.*, vol. 82, no. 12, pp. 2564–2573, 1999.
- [19] M. S. Gulay, M. J. Hayen, M. Liboni, T. I. Belloso, C. J. Wilcox, and H. H. Head, "Low doses of bovine somatotropin during the transition period and early lactation improves milk yield, efficiency of production, and other physiological responses of Holstein cows," *J. Dairy Sci.*, vol. 87, no. 4, pp. 948–960, 2004.
- [20] S. Koennings, A. Sapin, T. Blunk, P. Menei, and A. Goepferich, "Towards controlled release of BDNF -Manufacturing strategies for protein-loaded lipid implants and biocompatibility evaluation in the brain," *J. Control. Release*, vol. 119, no. 2, pp. 163–172, 2007.
- [21] D. Bodmer, T. Kissel, and E. Traechslin, "Factors influencing the release of peptides and proteins from biodegradable parenteral depot systems," *J. Control. Release*, vol. 21, no. 1–3, pp. 129–137, 1992.
- [22] M. P. Even, K. Young, G. Winter, S. Hook, and J. Engert, "In vivo investigation of twin-screw extruded lipid implants for vaccine delivery," *Eur. J. Pharm. Biopharm.*, vol. 87, no. 2, pp. 338–346, 2014.
- [23] C. Guse, S. Koennings, A. Maschke, M. Hacker, C. Becker, S. Schreiner, T. Blunk, T. Spruss, and A. Goepferich, "Biocompatibility and erosion behavior of implants made of triglycerides and blends with cholesterol and phospholipids," *Int. J. Pharm.*, vol. 314, no. 2, pp. 153–160, 2006.
- [24] S. P. Schwendeman, R. B. Shah, B. A. Bailey, and A. S. Schwendeman, "Injectable controlled release depots for large molecules," *J. Control. Release*, vol. 190, pp. 240–253, 2014.
- [25] J. Siepmann and A. Göpferich, "Mathematical modeling of bioerodible, polymeric drug delivery systems," *Adv. Drug Deliv. Rev.*, vol. 48, no. 2–3, pp. 229–247, 2001.
- [26] S. Herrmann, S. Mohl, F. Siepmann, J. Siepmann, and G. Winter, "New insight into the role of polyethylene glycol acting as protein release modifier in lipidic implants," *Pharm. Res.*, vol. 24, no. 8, pp. 1527–1537, 2007.
- [27] J. Kang and S. P. Schwendeman, "Comparison of the effects of Mg(OH)2 and sucrose on the stability of bovine serum albumin encapsulated in injectable poly(D,L-lactide-co-glycolide) implants," *Biomaterials*, vol. 23, no. 1, pp. 239–245, 2002.
- [28] S. V. Fulzele, P. M. Satturwar, and A. K. Dorle, "Study of the biodegradation and in vivo biocompatibility of novel biomaterials," *Eur. J. Pharm. Sci.*, vol. 20, no. 1, pp. 53–61, 2003.
- [29] C. Wischke, Y. Zhang, S. Mittal, and S. P. Schwendeman, "Development of PLGA-based injectable delivery systems for hydrophobic fenretinide," *Pharm. Res.*, vol. 27, no. 10, pp. 2063–2074, 2010.
- [30] A. Lucke, E. Fustella, J. Teßmar, A. Gazzaniga, and A. Göpferich, "The effect of poly(ethylene glycol)poly(D,L-lactic acid) diblock copolymers on peptide acylation," *J. Control. Release*, vol. 80, no. 1–3, pp. 157–168, 2002.
- [31] A. H. Ghassemi, M. J. Van Steenbergen, A. Barendregt, H. Talsma, R. J. Kok, C. F. Van Nostrum, D. J. A. Crommelin, and W. E. Hennink, "Controlled release of octreotide and assessment of peptide acylation from poly(D,L-lactide-co-hydroxymethyl glycolide) compared to PLGA microspheres," *Pharm. Res.*, vol. 29, no. 1, pp. 110–120, 2012.

- [32] M. Wong, S. O'Neill, G. Walsh, and I. E. Smith, "Goserelin with chemotherapy to preserve ovarian function in pre-menopausal women with early breast cancer: Menstruation and pregnancy outcomes," *Ann. Oncol.*, vol. 24, no. 1, pp. 133–138, 2013.
- [33] H. M. Fraser, J. Sandow, G. M. Cowen, M. A. Lumsden, R. Haining, and S. K. Smith, "Long-term suppression of ovarian function by a luteinizing-hormone releasing hormone agonist implant in patients with endometriosis.," *Fertil. Steril.*, vol. 53, no. 1, pp. 61–68, 1990.
- [34] Astellas, "Eligard 22.5 mg Package Insert," 2015. [Online]. Available: http://www.astellas.de/service/packungsbeilagen.html. [Accessed: 20-Dec-2016].
- [35] H. M. Fraser and D. T. Baird, "Clinical applications of LHRH analogues," *Baillières Clin. Endocrinol. Metab.*, vol. 1, no. 1, pp. 43–70, 1987.
- [36] G. Sax and G. Winter, "Mechanistic studies on the release of lysozyme from twin-screw extruded lipid implants," J. Control. Release, vol. 163, no. 2, pp. 187–194, 2012.
- [37] M. P. Even, S. Bobbala, K. L. Kooi, S. Hook, G. Winter, and J. Engert, "Impact of implant composition of twin-screw extruded lipid implants on the release behavior," *Int. J. Pharm.*, vol. 493, no. 1–2, pp. 102– 110, 2015.
- [38] C. Reitz, C. Strachan, and P. Kleinebudde, "Solid lipid extrudates as sustained-release matrices: The effect of surface structure on drug release properties," *Eur. J. Pharm. Sci.*, vol. 35, no. 4, pp. 335–343, 2008.
- [39] W. Vogelhuber, E. Magni, A. Gazzaniga, and A. Gopferich, "Monolithic glyceryl trimyristate matrices for parenteral drug release applications," *Eur. J. Pharm. Biopharm.*, vol. 55, no. 1, pp. 133–138, 2003.
- [40] S. Schulze and G. Winter, "Lipid extrudates as novel sustained release systems for pharmaceutical proteins," J. Control. Release, vol. 134, no. 3, pp. 177–185, 2009.
- [41] S. Herrmann, G. Winter, S. Mohl, F. Siepmann, and J. Siepmann, "Mechanisms controlling protein release from lipidic implants: Effects of PEG addition," *J. Control. Release*, vol. 118, no. 2, pp. 161–168, 2007.
- [42] Zoetis, "Excede®," 2013. [Online]. Available: https://www.zoetisus.com/_locale-assets/mcm-portalassets/msds_pi/pi/excede-horses.pdf. [Accessed: 12-Dec-2016].
- [43] Zoetis, "Excenel® RTU Package insert," 2013. [Online]. Available: https://www.zoetisus.com/_localeassets/mcm-portal-assets/products/pdf/excenel_rtu_pi_fullpg.pdf. [Accessed: 12-Dec-2016].
- [44] J. Kreuter, "Nanoparticles and microparticles for drug and vaccine delivery.," J. Anat., vol. 189 (Pt 3), pp. 503–505, 1996.
- [45] C. Freitas and R. H. Müller, "Spray-drying of solid lipid nanoparticles (SLN(TM))," Eur. J. Pharm. Biopharm., vol. 46, no. 2, pp. 145–151, 1998.
- [46] B. V. N. Nagavarma, H. K. S. Yadav, A. Ayaz, L. S. Vasudha, and H. G. Shivakumar, "Different techniques for preparation of polymeric nanoparticles- A review," *Asian J. Pharm. Clin. Res.*, vol. 5, no. SUPPL. 3, pp. 16–23, 2012.
- [47] R. H. Müller, K. Mäder, and S. Gohla, "Solid lipid nanoparticles (SLN) for controlled drug delivery a review of the state of the art.," *Eur. J. Pharm. Biopharm.*, vol. 50, no. 1, pp. 161–177, 2000.
- [48] M. Bivas-Benita, S. Romeijn, H. E. Junginger, and G. Borchard, "PLGA-PEI nanoparticles for gene delivery to pulmonary epithelium," *Eur. J. Pharm. Biopharm.*, vol. 58, no. 1, pp. 1–6, 2004.
- [49] M. T. Peracchia, E. Fattal, D. Desmaële, M. Besnard, J. P. Noël, J. M. Gomis, M. Appel, J. D'Angelo, and P. Couvreur, "Stealth(®) PEGylated polycyanoacrylate nanoparticles for intravenous administration and splenic targeting," *J. Control. Release*, vol. 60, no. 1, pp. 121–128, 1999.

- [50] Y. Tabata and Y. Ikada, "Protein release from gelatin matrices," Adv. Drug Deliv. Rev., vol. 31, no. 3, pp. 287–301, 1998.
- [51] M. J. Santander-Ortega, N. Csaba, L. González, D. Bastos-González, J. L. Ortega-Vinuesa, and M. J. Alonso, "Protein-loaded PLGA-PEO blend nanoparticles: Encapsulation, release and degradation characteristics," *Colloid Polym. Sci.*, vol. 288, no. 2, pp. 141–150, 2010.
- [52] H. Bunjes, K. Westesen, and M. H. J. Koch, "Crystallization tendency and polymorphic transitions in triglyceride nanoparticles," *Int. J. Pharm.*, vol. 129, no. 1–2, pp. 159–173, 1996.
- [53] K. M. Rosenblatt and H. Bunjes, "Poly(vinyl alcohol) as emulsifier stabilizes solid triglyceride drug carrier nanoparticles in the alpha-modification," *Mol. Pharm.*, vol. 6, no. 1, pp. 105–120, 2009.
- [54] E. B. Souto, W. Mehnert, and R. H. Müller, "Polymorphic behaviour of Comprise 888 ATO as bulk lipid and as SLN and NLC.," J. Microencapsul., vol. 23, no. 4, pp. 417–33, 2006.
- [55] J. K. Li, N. Wang, and X. S. Wu, "A novel biodegradable system based on gelatin nanoparticles and poly(lactic-co-glycolic acid) microspheres for protein and peptide drug delivery," *J. Pharm. Sci.*, vol. 86, no. 8, pp. 891–895, 1997.
- [56] J. K. Li, N. Wang, and X. S. Wu, "Gelatin nanoencapsulation of protein/peptide drugs using an emulsifierfree emulsion method.," *J. Microencapsul.*, vol. 15, no. 2, pp. 163–72, 1998.
- [57] M. Garcia-Fuentes, D. Torres, and M. J. Alonso, "New surface-modified lipid nanoparticles as delivery vehicles for salmon calcitonin," *Int. J. Pharm.*, vol. 296, pp. 122–132, 2005.
- [58] F. Q. Hu, Y. Hong, and H. Yuan, "Preparation and characterization of solid lipid nanoparticles containing peptide," *Int. J. Pharm.*, vol. 273, no. 1–2, pp. 29–35, 2004.
- [59] S. Scalia, P. M. Young, and D. Traini, "Solid lipid microparticles as an approach to drug delivery," *Expert Opin. Drug Deliv.*, vol. 12, no. 4, pp. 583–599, 2015.
- [60] Vanna Sanna, N. Kirschvink, P. Gustin, E. Gavini, I. Roland, L. Delattre, and B. Evrard, "Preparation and in vivo toxicity study of solid lipid microparticles as carrier for pulmonary administration," AAPS PharmSciTech, vol. 5, no. 2, pp. 17–23, 2004.
- [61] E. Gavini, P. Chetoni, M. Cossu, M. G. Alvarez, M. F. Saettone, and P. Giunchedi, "PLGA microspheres for the ocular delivery of a peptide drug, vancomycin using emulsification/spray-drying as the preparation method: in vitro/in vivo studies," *Eur. J. Pharm. Biopharm.*, vol. 57, no. 2, pp. 207–212, 2004.
- [62] S. P. Schwendeman, M. Tobio, M. Joworowicz, M. J. Alonso, and R. Langer, "New strategies for the microencapsulation of tetanus vaccine," *J. Microencapsul.*, vol. 15, no. 3, pp. 299–318, 1998.
- [63] D. Yin, Y. Lu, H. Zhang, G. Zhang, H. Zou, D. Sun, and Y. Zhong, "Preparation of glucagon-like peptide-1 loaded PLGA microspheres: characterizations, release studies and bioactivities in vitro/in vivo," *Chem. Pharm. Bull.*, vol. 56, no. 2, pp. 156–161, 2008.
- [64] B. Gander, E. Wehrli, R. Alder, and H. P. Merkle, "Quality improvement of spray-dried, protein-loaded D,L-PLA microspheres by appropriate polymer solvent selection.," *J. Microencapsul.*, vol. 12, no. 1, pp. 83–97, 1995.
- [65] M. L. Tan, P. F. M. Choong, and C. R. Dass, "Recent developments in liposomes, microparticles and nanoparticles for protein and peptide drug delivery," *Peptides*, vol. 31, no. 1. pp. 184–193, 2010.
- [66] T. Morita, Y. Sakamura, Y. Horikiri, T. Suzuki, and H. Yoshino, "Protein encapsulation into biodegradable microspheres by a novel S/O/W emulsion method using poly(ethylene glycol) as a protein micronization adjuvant," J. Control. Release, vol. 69, no. 3, pp. 435–444, 2000.

- [67] S. Saraf, D. Mishra, A. Asthana, R. Jain, S. Singh, and N. K. Jain, "Lipid microparticles for mucosal immunization against hepatitis B," *Vaccine*, vol. 24, no. 1, pp. 45–56, 2006.
- [68] P. C. Christophersen, L. Zhang, A. Müllertz, H. M. Nielsen, M. Yang, and H. Mu, "Solid Lipid Particles for Oral Delivery of Peptide and Protein Drugs II - The Digestion of Trilaurin Protects Desmopressin from Proteolytic Degradation," *Pharm. Res.*, pp. 1–9, 2014.
- [69] J. Herrmann and R. Bodmeier, "The effect of particle microstructure on the somatostatin release from poly(lactide) microspheres prepared by a W/O/W solvent evaporation method," J. Control. Release, vol. 36, no. 1–2, pp. 63–71, 1995.
- [70] T. Simón-Yarza, F. R. Formiga, E. Tamayo, B. Pelacho, F. Prosper, and M. J. Blanco-Prieto, "PEGylated-PLGA microparticles containing VEGF for long term drug delivery," *Int. J. Pharm.*, vol. 440, no. 1, pp. 13–18, 2013.
- [71] A. G. Ding and S. P. Schwendeman, "Acidic microclimate pH Distribution in PLGA microspheres monitored by confocal laser scanning microscopy," *Pharm. Res.*, vol. 25, no. 9, pp. 2041–2052, 2008.
- [72] Virbac, "SMARTshot® B12 Prime Lamb," 2010. [Online]. Available: https://www.virbac.co.nz/home/ruminants/products/trace-elements/smartshot-b12-prime-lamb.htm. [Accessed: 12-Dec-2016].
- [73] AstraZeneca, "Bydureon® Package insert," 2015. [Online]. Available: http://www.patienteninfoservice.de/a-z-liste/b/bydureonR-2-mg-pulver-und-loesungsmittel-zur-herstellung-einer-depotinjektionssuspension/. [Accessed: 15-Dec-2016].
- [74] H. Reithmeier, J. Herrmann, and A. Gopferich, "Lipid microparticles as a parenteral controlled release device for peptides," J. Control. Release, vol. 73, no. 2–3, pp. 339–350, 2001.
- [75] H. Reithmeier, J. Herrmann, and A. Gopferich, "Development and characterization of lipid microparticles as a drug carrier for somatostatin," *Int. J. Pharm.*, vol. 218, no. 1–2, pp. 133–143, 2001.
- [76] M. D. Del Curto, D. Chicco, M. D'Antonio, V. Ciolli, H. Dannan, S. D'Urso, B. Neuteboom, S. Pompili,
 S. Schiesaro, and P. Esposito, "Lipid microparticles as sustained release system for a GnRH antagonist (Antide)," *J. Control. Release*, vol. 89, no. 2, pp. 297–310, 2003.
- [77] M. Schwab, C. M. McGoverin, K. C. Gordon, G. Winter, T. Rades, J. Myschik, and C. J. Strachan, "Studies on the lipase-induced degradation of lipid-based drug delivery systems. Part II - Investigations on the mechanisms leading to collapse of the lipid structure," *Eur. J. Pharm. Biopharm.*, vol. 84, no. 3, pp. 456– 463, 2013.
- [78] M. van de Weert, W. E. Hennink, and W. Jiskoot, "Protein instability in poly(lactic-co-glycolic acid) microparticles.," *Pharm. Res.*, vol. 17, no. 10, pp. 1159–1167, 2000.
- [79] A. Zaky, A. Elbakry, A. Ehmer, M. Breunig, and A. Goepferich, "The mechanism of protein release from triglyceride microspheres," J. Control. Release, vol. 147, no. 2, pp. 202–210, 2010.
- [80] W. Vogelhuber, E. Magni, M. Mouro, T. Spruss, C. Guse, A. Gazzaniga, and A. Gopferich, "Monolithic triglyceride matrices: a controlled-release system for proteins," *Pharm. Dev. Technol.*, vol. 8, no. 1, pp. 71–79, 2003.
- [81] R. Cortesi, E. Esposito, G. Luca, and C. Nastruzzi, "Production of lipospheres as carriers for bioactive compounds," *Biomaterials*, vol. 23, no. 11, pp. 2283–2294, 2002.
- [82] M. Mezzena, S. Scalia, P. M. Young, and D. Traini, "Solid lipid budesonide microparticles for controlled release inhalation therapy," AAPS J., vol. 11, no. 4, pp. 771–778, 2009.
- [83] S. Scalia, D. Traini, P. M. Young, M. Di Sabatino, N. Passerini, and B. Albertini, "Comparison of spray congealing and melt emulsification methods for the incorporation of the water-soluble salbutamol sulphate in lipid microparticles," *Pharm. Dev. Technol.*, vol. 18, no. 1, pp. 266–273, 2013.
- [84] S. Jaspart, P. Bertholet, G. Piel, J. M. Dogne, L. Delattre, and B. Evrard, "Solid lipid microparticles as a sustained release system for pulmonary drug delivery," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 1, pp. 47– 56, 2007.
- [85] S. Scalia and M. Mezzena, "Incorporation of quercetin in lipid microparticles: Effect on photo- and chemical-stability," J. Pharm. Biomed. Anal., vol. 49, no. 1, pp. 90–94, 2009.
- [86] R. Bodmeier, J. Wang, and H. Bhagwatwar, "Process and formulation variables in the preparation of wax microparticles by a melt dispersion technique. II. W/O/W multiple emulsion technique for water-soluble drugs.," J. Microencapsul., vol. 9, no. 1, pp. 99–107, 1992.
- [87] M. Mumenthaler, C. C. Hsu, and R. Pearlman, "Feasibility study on spray-drying protein pharmaceuticals: recombinant human growth hormone and tissue-type plasminogen activator," *Pharm. Res.*, vol. 11, no. 1, pp. 12–20, 1994.
- [88] J. Broadhead, C. T. Rhodes, and S. K. E. Rouan, "The spray drying of pharmaceuticals," *Drug Dev. Ind. Pharm.*, vol. 18, no. 11–12, pp. 1169–1206, 1992.
- [89] J. C. Kasper, G. Winter, and W. Friess, "Recent advances and further challenges in lyophilization," *Eur. J. Pharm. Biopharm.*, vol. 85, no. 2, pp. 162–169, 2013.
- [90] F. Quaglia, G. De Rosa, E. Granata, F. Ungaro, E. Fattal, and M. I. La Rotonda, "Feeding liquid, non-ionic surfactant and cyclodextrin affect the properties of insulin-loaded poly(lactide-co-glycolide) microspheres prepared by spray-drying," *J. Control. Release*, vol. 86, no. 2–3, pp. 267–278, 2003.
- [91] T. Eldem, P. Speiser, and A. Hincal, "Optimization of spray-dried and -congealed lipid micropellets and characterization of their surface morphology by scanning electron microscopy," *Pharm. Res.*, vol. 8, no. 1, pp. 47–54, 1991.
- [92] A. Cambronero-Rojas, P. Torres-Vergara, R. Godoy, C. Von Plessing, J. Sepúlveda, and C. Gómez-Gaete, "Capreomycin oleate microparticles for intramuscular administration: Preparation, in vitro release and preliminary in vivo evaluation," *J. Control. Release*, vol. 209, pp. 229–237, 2015.
- [93] M.-I. Ré, "Formulating Drug Delivery Systems by Spray Drying," Dry. Technol., vol. 24, no. 4, pp. 433–446, 2006.
- [94] R. Vehring, "Pharmaceutical particle engineering via spray drying," *Pharm. Res.*, vol. 25, no. 5, pp. 999– 1022, 2008.
- [95] K. Ståhl, M. Claesson, P. Lilliehorn, H. Lindén, and K. Bäckström, "The effect of process variables on the degradation and physical properties of spray dried insulin intended for inhalation," *Int. J. Pharm.*, vol. 233, no. 1–2, pp. 227–237, 2002.
- [96] M. B. Mackaplow, I. E. Zarraga, and J. F. Morris, "Rotary spray congealing of a suspension: effect of disk speed and dispersed particle properties.," *J. Microencapsul.*, vol. 23, no. November, pp. 793–809, 2006.
- [97] N. Passerini, B. Perissutti, B. Albertini, D. Voinovich, M. Moneghini, and L. Rodriguez, "Controlled release of verapamil hydrochloride from waxy microparticles prepared by spray congealing," J. Control. Release, vol. 88, no. 2, pp. 263–275, 2003.
- [98] M. Savolainen, C. Khoo, H. Glad, C. Dahlqvist, and A. M. Juppo, "Evaluation of controlled-release polar lipid microparticles," *Int. J. Pharm.*, vol. 244, no. 1–2, pp. 151–161, 2002.

- [99] B. Albertini, N. Passerini, F. Pattarino, and L. Rodriguez, "New spray congealing atomizer for the microencapsulation of highly concentrated solid and liquid substances," *Eur. J. Pharm. Biopharm.*, vol. 69, no. 1, pp. 348–357, 2008.
- [100] B. Albertini, M. Mezzena, N. Passerini, L. Rodriguez, and S. Scalia, "Evaluation of spray congealing as technique for the preparation of highly loaded solid lipid microparticles containing the sunscreen agent, avobenzone," J. Pharm. Sci., vol. 98, no. 8, pp. 2759–2769, 2009.
- [101] R. Al-Kassas, R. F. Donnelly, and P. A. McCarron, "Aminolevulinic acid-loaded Witepsol microparticles manufactured using a spray congealing procedure: implications for topical photodynamic therapy," J. *Pharm. Pharmacol.*, vol. 61, no. 9, pp. 1125–1135, 2009.
- [102] A. Fini, L. Rodriguez, C. Cavallari, B. Albertini, and N. Passerini, "Ultrasound-compacted and spraycongealed indomethacin/polyethyleneglycol systems," *Int. J. Pharm.*, vol. 247, no. 1–2, pp. 11–22, 2002.
- [103] A. Fini, C. Cavallari, G. Ceschel, and A. M. Rabasco, "Bimodal release of olanzapine from lipid microspheres," J. Pharm. Sci., vol. 99, no. 10, pp. 4251–4260, 2010.
- [104] L. Rodriguez, N. Passerini, C. Cavallari, M. Cini, P. Sancin, and A. Fini, "Description and preliminary evaluation of a new ultrasonic atomizer for spray-congealing processes," *Int. J. Pharm.*, vol. 183, no. 2, pp. 133–143, 1999.
- [105] C. Cavallari, B. Luppi, A. M. Di Pietra, L. Rodriguez, and A. Fini, "Enhanced release of indomethacin from Pvp/stearic acid microcapsules prepared coupling Co-freeze-drying and ultrasound assisted spraycongealing process," *Pharm. Res.*, vol. 24, no. 3, pp. 521–529, 2007.
- [106] M. W. Scott, M. J. Robinson, J. F. Pauls, and R. J. Lantz, "Spray Congealing: Particle Size Relationships Using a Centrifugal Wheel Atomizer," J. Pharm. Sci., vol. 53, pp. 670–675, 1964.
- [107] T. Yajima, N. Umeki, and S. Itai, "Optimum spray congealing conditions for masking the bitter taste of clarithromycin in wax matrix," *Chem. Pharm. Bull.*, vol. 47, no. 2, pp. 220–225, 1999.
- [108] S. Jaspart, G. Piel, L. Delattre, and B. Evrard, "Solid lipid microparticles: formulation, preparation, characterisation, drug release and applications," *Expert Opin. Drug Deliv.*, vol. 2, no. 1, pp. 75–87, 2005.
- [109] M. Di Sabatino, B. Albertini, V. L. Kett, and N. Passerini, "Spray congealed lipid microparticles with high protein loading: preparation and solid state characterisation," *Eur. J. Pharm. Sci.*, vol. 46, no. 5, pp. 346– 356, 2012.
- [110] W. Wang, "Protein aggregation and its inhibition in biopharmaceutics," *Int. J. Pharm.*, vol. 289, no. 1–2, pp. 1–30, 2005.
- [111] J. S. Stevenson and D. L. Davis, "Estrous synchronization and fertility in gilts after 14- or 18-day feeding of altrenogest beginning at estrus or diestrus.," J. Anim. Sci., vol. 55, no. 1, pp. 119–123, 1982.
- [112] J. Kauffold, J. Beckjunker, A. Kanora, and W. Zaremba, "Synchronization of estrus and ovulation in sows not conceiving in a scheduled fixed-time insemination program," *Anim. Reprod. Sci.*, vol. 97, no. 1–2, pp. 84–93, 2007.
- [113] C. M. Wood, E. T. Kornegay, and C. F. Shipley, "Efficacy of altrenogest in synchronizing estrus in two swine breeding programs and effects on subsequent reproductive performance of sows," *J. Anim. Sci.*, vol. 70, no. 5, pp. 1357–1364, 1992.
- [114] R. V Knox, K. L. Willenburg, S. L. Rodriguez-Zas, D. L. Greger, H. D. Hafs, and M. E. Swanson, "Synchronization of ovulation and fertility in weaned sows treated with intravaginal triptorelin is influenced by timing of administration and follicle size," *Theriogenology*, vol. 75, no. 2, pp. 308–319,

2011.

- [115] L. L. Larson and P. J. H. Ball, "Regulation of estrous cycles in dairy cattle: A review," *Theriogenology*, vol. 38, no. 2. pp. 255–267, 1992.
- [116] K. P. Brussow, F. Schneider, W. Kanitz, J. Ratky, J. Kauffold, and M. Wahner, "Studies on fixed-time ovulation induction in the pig," Soc. Reprod. Fertil. Suppl., vol. 66, pp. 187–195, 2009.
- [117] F. Martinat-Botté, F. Bariteau, Y. Forgerit, C. Macar, P. Poirier, and M. Terqui, "Synchronization of oestrus in gilts with altrenogest: effects on ovulation rate and foetal survival," *Anim. Reprod. Sci.*, vol. 39, no. 4, pp. 267–274, 1995.
- [118] F. Martinat-Botte, F. Bariteau, Y. Forgerit, C. Macar, A. Moreau, M. Terqui, and J. P. Signoret, "Control of oestrus in gilts II. Synchronization of oestrus with a progestagen, altrenogest (Regumate): Effect on fertility and litter size," *Anim. Reprod. Sci.*, vol. 22, no. 3, pp. 227–233, 1990.
- [119] J. A. Abecia, F. Forcada, and A. González-Bulnes, "Pharmaceutical Control of Reproduction in Sheep and Goats," Vet. Clin. North Am. Anim. Pract., vol. 27, no. 1, pp. 67–79, 2011.
- [120] N. C. Whitley and D. J. Jackson, "An update on estrus synchronization in goats: a minor species.," J. Anim. Sci., vol. 82 E-Suppl, 2004.
- [121] K. N. Bretzlaff, L. C. Nuti, R. G. Elmore, S. A. Meyers, J. N. Rugila, S. P. Brinsko, T. L. Blanchard, and P. G. Weston, "Synchronization of estrus in dairy goats given norgestomet and estradiol valerate at various stages of the estrous cycle," *Am. J. Vet. Res.*, vol. 53, no. 6, pp. 930–934, 1992.
- [122] T. L. Blanchard, D. D. Varner, P. J. Burns, K. A. Everett, S. P. Brinsko, and L. Boehnke, "Regulation of estrus and ovulation in mares with progesterone or progesterone and estradiol biodegradable microspheres with or without PGF2alpha," *Theriogenology*, vol. 38, no. 6, pp. 1091–1106, 1992.
- [123] W. Kanitz, F. Schneider, H. O. Hoppen, C. Unger, G. Nurnberg, and F. Becker, "Pregnancy rates, LH and progesterone concentrations in mares treated with a GnRH agonist," *Anim. Reprod. Sci.*, vol. 97, no. 1–2, pp. 55–62, 2007.
- Z. A. Dequattro, E. J. Peissig, D. S. Antkiewicz, E. J. Lundgren, C. J. Hedman, J. D. C. Hemming, and T. P. Barry, "Effects of progesterone on reproduction and embryonic development in the fathead minnow (Pimephales promelas)," *Environ. Toxicol. Chem.*, vol. 31, no. 4, pp. 851–856, 2012.
- [125] S. M. Havens, C. J. Hedman, J. D. C. Hemming, M. G. Mieritz, M. M. Shafer, and J. J. Schauer, "Stability, preservation, and quantification of hormones and estrogenic and androgenic activities in surface water runoff," *Environ. Toxicol. Chem.*, vol. 29, no. 11, pp. 2481–2490, 2010.
- [126] E. F. Orlando and L. E. Ellestad, "Sources, concentrations, and exposure effects of environmental gestagens on fish and other aquatic wildlife, with an emphasis on reproduction," *Gen. Comp. Endocrinol.*, vol. 203, pp. 241–249, 2014.
- S. L. Bartelt-Hunt, D. D. Snow, W. L. Kranz, T. L. Mader, C. A. Shapiro, S. J. Van Donk, D. P. Shelton, D. D. Tarkalson, and T. C. Zhang, "Effect of growth promotants on the occurrence of endogenous and synthetic steroid hormones on feedlot soils and in runoff from beef cattle feeding operations," *Environ. Sci. Technol.*, vol. 46, no. 3, pp. 1352–1360, 2012.
- [128] J. Zeilinger, T. Steger-Hartmann, E. Maser, S. Goller, R. Vonk, and R. Länge, "Effects of synthetic gestagens on fish reproduction.," *Environ. Toxicol. Chem.*, vol. 28, no. 12, pp. 2663–2670, 2009.
- [129] J. P. Besse and J. Garric, "Progestagens for human use, exposure and hazard assessment for the aquatic environment," *Environ. Pollut.*, vol. 157, no. 12, pp. 3485–3494, 2009.

- [130] H. Berger, N. Heinrich, E. Albrecht, U. Kertscher, J. Oehlke, M. Bienert, H. Schafer, I. Baeger, and B. Mehlis, "Gonadotropin-releasing hormone (GnRH) analogs: relationship between their structure, proteolytic inactivation and pharmacokinetics in rats," *Regul. Pept.*, vol. 33, no. 3, pp. 299–311, 1991.
- [131] W. Hansel and E. M. Convey, "Physiology of the estrous cycle.," J. Anim. Sci., vol. 57 Suppl 2, pp. 404–424, 1983.
- [132] V. J. Helm and B. W. Muller, "Stability of gonadorelin and triptorelin in aqueous solution," *Pharm. Res.*, vol. 7, no. 12, pp. 1253–1256, 1990.
- [133] P. M. Conn and W. F. Crowley Jr., "Gonadotropin-releasing hormone and its analogs," Annu. Rev. Med., vol. 45, pp. 391–405, 1994.
- [134] L. N. Brown, K. G. Odde, M. E. King, D. G. LeFever, and C. J. Neubauer, "Comparison of Melengestrol Acetate-Prostaglandin F2-alpha to Syncro-Mate B for estrus synchronization in beef heifers," *Theriogenology*, vol. 30, no. 1, pp. 1–12, 1988.
- [135] F. N. Kojima, B. E. Salfen, J. F. Bader, W. A. Ricke, M. C. Lucy, M. F. Smith, and D. J. Patterson, "Development of an estrus synchronization protocol for beef cattle with short-term feeding of melengestrol acetate: 7-11 synch," J. Anim. Sci., vol. 78, no. 8, pp. 2186–2191, 2000.
- [136] F. Schneider, W. Tomek, and C. Grundker, "Gonadotropin-releasing hormone (GnRH) and its natural analogues: a review," *Theriogenology*, vol. 66, no. 4, pp. 691–709, 2006.
- [137] A. M. Padula, "GnRH analogues—agonists and antagonists.," Anim. Reprod. Sci., vol. 88, pp. 115–126, 2005.
- [138] H. C. F. Moore, J. M. Unger, K.-A. Phillips, F. Boyle, E. Hitre, D. Porter, P. A. Francis, L. J. Goldstein, H. L. Gomez, C. S. Vallejos, A. H. Partridge, S. R. Dakhil, A. A. Garcia, J. Gralow, J. M. Lombard, J. F. Forbes, S. Martino, W. E. Barlow, C. J. Fabian, L. Minasian, F. L. Meyskens, R. D. Gelber, G. N. Hortobagyi, and K. S. Albain, "Goserelin for ovarian protection during breast-cancer adjuvant chemotherapy.," *N. Engl. J. Med.*, vol. 372, no. 10, pp. 923–32, 2015.
- [139] P. Karlsson, Z. Sun, D. Braun, K. N. Price, M. Castiglione-gertsch, M. Rabaglio, R. D. Gelber, D. Crivellari, J. Collins, E. Murray, K. Zaman, M. Colleoni, B. A. Gusterson, G. Viale, M. M. Regan, A. S. Coates, and A. Goldhirsch, "Long-term results of International Breast Cancer Study Group Trial Viii: Adjuvant chemotherapy plus goserelin compared with either therapy alone for premenopausal patients with node-negative breast cancer," *Ann. Oncol.*, vol. 22, no. 10, pp. 2216–2226, 2011.
- [140] R. P. Millar, Z. L. Lu, A. J. Pawson, C. A. Flanagan, K. Morgan, and S. R. Maudsley, "Gonadotropinreleasing hormone receptors," *Endocrine Reviews*, vol. 25, no. 2. pp. 235–275, 2004.
- [141] Veyx Pharma, W. Zaremba, and S. Engl, "Maprelin®," 2013. [Online]. Available: https://www.veyx.de/uploads/tx_sitscontentattribute/KB_Maprelin_DE_130805_02.pdf. [Accessed: 13-Dec-2016].
- [142] S. C. Hall, M. M. Tan, J. J. Leonard, and C. L. Stevenson, "Characterization and comparison of leuprolide degradation profiles in water and dimethyl sulfoxide," *J. Pept. Res.*, vol. 53, no. 4, pp. 432–441, 1999.
- [143] R. N. Kirkwood and J. Kauffold, "Advances in Breeding Management and Use of Ovulation Induction for Fixed-time AI," *Reprod. Domest. Anim.*, vol. 50, pp. 85–89, 2015.
- [144] K. P. Brussow, F. Schneider, A. Tuchscherer, J. Ratky, R. R. Kraeling, and W. Kanitz, "Luteinizing hormone release after administration of the gonadotropin-releasing hormone agonist Fertilan (goserelin) for synchronization of ovulation in pigs," *J. Anim. Sci.*, vol. 85, no. 1, pp. 129–137, 2007.

- [145] T. E. Trigg, P. J. Wright, A. F. Armour, P. E. Williamson, A. Junaidi, G. B. Martin, A. G. Doyle, and J. Walsh, "Use of a GnRH Analogue Implant to Produce Reversible Long-term Suppression of Reproductive Function in Male and Female Domestic Dogs.," *J. Reprod. Fertil. Suppl.*, vol. 57, pp. 255–261, 2001.
- [146] D. Ezcurra and P. Humaidan, "A review of luteinising hormone and human chorionic gonadotropin when used in assisted reproductive technology.," *Reprod. Biol. Endocrinol.*, vol. 12, no. 1, p. 95, 2014.
- [147] Ceva Santé animale, "Fertipig® Brochure." [Online]. Available: http://www.ceva.asia/Products/Product-List/FERTIPIG-R. [Accessed: 01-Dec-2016].
- [148] MSD Tiergesundheit, "Suigonan® 80/40 I.E./ml," 2013. [Online]. Available: http://www.msd-tiergesundheit.de/products/suigonan.aspx. [Accessed: 01-Dec-2016].
- [149] MSD Tiergesundheit, "Intergonan® 240 I.E./ml," 2014. [Online]. Available: http://www.msd-tiergesundheit.de/products/intergonan/intergonan.aspx. [Accessed: 01-Dec-2016].
- [150] F. von Kaufmann and W. Holtz, "Induction of ovulation in gonadotropin treated gilts with synthetic gonadotropin releasing hormone," *Theriogenology*, vol. 17, no. 2, pp. 141–157, 1982.
- [151] T. T. Jung, S. K. Juhn, D. Hwang, and R. Stewart, "Prostaglandins, leukotrienes, and other arachidonic acid metabolites in nasal polyps and nasal mucosa.," *Laryngoscope*, vol. 97, no. 2, pp. 184–9, 1987.
- [152] H. D. Guthrie, "Estrous synchronization and fertility in gilts treated with estradiol-benzoate and prostaglandin F2α," *Theriogenology*, vol. 4, no. 2–3, pp. 69–75, 1975.
- [153] K. L. Willenburg, R. V Knox, and R. N. Kirkwood, "Effect of estrogen formulation and its site of deposition on serum PGFM concentrations, uterine contractility, and time of ovulation in artificially inseminated weaned sows.," *Anim. Reprod. Sci.*, vol. 80, no. 1–2, pp. 147–156, 2004.
- [154] C. Delgado, "Rising consumption of meat and milk in developing countries has created a new food revolution," J. Nutr., vol. 133, no. 11, p. 3875S–4061S, 2003.
- [155] J. E. Morthorst, H. Holbech, and P. Bjerregaard, "Trenbolone causes irreversible masculinization of zebrafish at environmentally relevant concentrations," *Aquat. Toxicol.*, vol. 98, no. 4, pp. 336–343, 2010.
- [156] H. J. Hamlin, R. H. Lowers, L. C. Albergotti, M. W. McCoy, J. Mutz, and L. J. Guillette, "Environmental influence on yolk steroids in American alligators (Alligator mississippiensis).," *Biol. Reprod.*, vol. 83, no. 5, pp. 736–41, 2010.
- [157] L. J. Guillette, T. M. Edwards, and B. C. Moore, "Alligators, contaminants and steroid hormones.," *Environ. Sci.*, vol. 14, no. 6, pp. 331–347, 2015.
- [158] A. Ogawa, J. Dake, Y. Iwashina, and T. Tokumoto, "Induction of ovulation in Xenopus without hCG injection: the effect of adding steroids into the aquatic environment.," *Reprod. Biol. Endocrinol.*, vol. 9, no. 1, p. 11, 2011.

2 DEVELOPMENT OF A SPRAY CONGEALING PROCESS FOR TRIGLYCERIDE MICROPARTICLES

2.1 Abstract

A spray congealing process for the manufacturing of solid triglyceride microparticles was to be developed. The instrument settings, like product flow, spray pressure and spray flow were evaluated with respect to their effect on particle size and yield. Spray flow was identified as the most important factor with a moderate effect on yield and mean particle size. An improvement of yield and size reduction was obtained with higher spray flow. Particles of approximately 50 μ m were obtained at 20 to 50 % yield. Upon spray congealing, the triglycerides crystallized in the metastable α -polymorph. Transition into the thermodynamically stable β -form could be achieved by an additional tempering step.

2.2 Introduction

Microparticles are widely used delivery systems for both small molecules and biopharmaceutic drugs [1]–[3]. Polymer-based microparticles, especially based on lipids are of particular interest [4]–[9]. Different preparation techniques exist for lipid microparticles mainly based on emulsion or spray techniques [10] [11]. Molten lipids can be emulsified into an aqueous phase at higher temperatures forming solid, spherical microparticles upon cooling [11]. The solvent evaporation technique is based on an organic solution of lipid dispersed in an aqueous phase followed by solvent removal resulting in the formation of particles [12] [13]. The disadvantages of these techniques are problems arising from residual organic solvents and a high loss of water soluble drug due to partitioning into the aqueous phase [13]. Alternatively, a solution of lipid in organic solvent can be spray dried [14].

Spray congealing, in literature also referred to as spray cooling or spray chilling, is a process without the use of organic solvents characterized by the atomization of a lipid melt [15] [16]. Typically, the small droplets are formed by the use of internal mixing two-fluid nozzles using compressed air or nitrogen as spray gas [17] [18]. Depending on the lipid's melting/solidification point, the droplets are sprayed into a chamber, which is kept at room temperature [19] or below [7]. In some cases a subsequent freeze-drying step is applied [7]. Alternatively, atomization of the lipid melt is performed with a Venturi effect-based wide pneumatic nozzle (WPN) [20]. Furthermore, by the use of an ultrasound-based system, molten lipid can be split up into small droplets upon contact with the sonotrode followed by solidification at room temperature [21]–[25]. In addition, centrifugal wheel [26] [27] and rotating disc atomizers [28] are described.

In the first steps of development of lipid-based microparticles for sustained release of G [6-D-Phe] by spray congealing, the influence of different process parameters in a two-fluid nozzle-based process was evaluated. Additionally, the use of different triglycerides was assessed. The results with placebo particles should form the base for the subsequent studies with model substances and the peptide drug.

2.3 Materials and Methods

2.3.1 Materials

Dynasan 116 (tripalmitin, D116), Dynasan 118 (tristearin, D118) and Dynasan 120 (triarachidin, D120) were kindly provided from Cremer Oleo (Witten, Germany). The thickening agent aluminum stearate was supplied from Fluka (Fluka, Sigma-Aldrich, Taufkirchen, Germany). Ultrapure deionized water with a conductivity of less than $0.055 \,\mu$ S/cm (Milli-Q Water systems, Millipore, MA, USA) was used throughout all experiments. Polysorbate 20 (PS 20) was purchased from Merck KGaA (Darmstadt, Germany).

2.3.2 Preparation of triglyceride microparticles by spray congealing



Figure 2-1: Schematic depiction of the spray congealing equipment additionally installed on top of a B-290 Mini spray dryer

12 g of triglyceride was melted on a hot plate and transferred into the sample container of a B-290 Mini spray dryer with an additional spray chilling setup (Büchi, Flawil, Switzerland) as visualized in *Figure 2-1*. The whole system was pre-conditioned for 1.5 hours prior to particle production (sample container 90 °C for D116 and 110 °C for D118 and D120, spray tower temperature 15 °C). The original sample container with a plane bottom by Büchi was used for production of microparticles. Additionally, a custom-made sample container with a coned bottom (tilt 13 °) to minimize material loss was tested for flow-out studies. The coned bottom container was made by the LMU fine mechanics department at the same dimensions as the original flat bottom container. The connector passage between sample container and nozzle was heated to 70 °C with a self-made resistance wire-based heating. The spray tower was fed with

pre-cooled air via an additional dehumidifier Deltatherm[®] LT (Deltatherm[®] Hirmer GmbH, Much, Germany). The lipid melt was sprayed using nitrogen at a pressure between 4 and 6 bar. A high-performance cyclone was used to separate the microparticles. Resulting microparticles were stored at 2-8 °C until further analysis was performed.

2.3.3 Determination of yield

Yield was determined by weighing the product container before and after microparticle production. Only microparticles collected in the product container were taken into account for yield calculation.

2.3.4 Evaluation of the product flow using different sample containers

The product flow into the nozzle depends on the temperature as well as viscosity of the used melt. D116 was used as lipid material. The valve was opened and the resulting product flow within 20 s collected and the flow calculated in ml/min. Both the flat bottom sample container as well as the coned bottom container were evaluated.

2.3.5 **Determination of particle size**

Approximately 20 mg of microparticles were suspended in 10 ml of an aqueous solution of 0.02 % PS 20. The sample was analyzed in the laser diffraction system Horiba Partica LA-950 (Horiba, Kyoto, Japan) while stirring.

2.3.6 Investigations on polymorphic behavior using differential scanning calorimetry (DSC)

Microparticles were analyzed using a Mettler DSC 821e (Mettler Toledo, Columbus, OH, USA). 5-15 mg were weighed into aluminum crucibles and analyzed at a heating and cooling rate of 10 K/min from 0 to 110 °C in two cycles against an empty crucible as reference. To determine polymorphic behavior upon tempering, particles were stored in glass vials at 2-8, 25 and 45 °C for 24, 48 and 72 h, respectively.

2.3.7 Characterization of particle morphology using scanning electron microscopy (SEM)

Surface characteristics of lipid microparticles were assessed by SEM using a Jeol JSM 6500-F SEM (Jeol, Tokyo, Japan) without additional coating at 150, 900 and 1500-fold magnification, after attaching on aluminum sample holders with double adhesive tape (Plano, Wetzlar, Germany).

2.4 Results and Discussion

2.4.1 Evaluation of the product flow into the nozzle using different sample containers

As the product flow into the nozzle depends on the used melt and temperature as well as valve opening, the product flow in ml/min was determined experimentally for D116 at 90 °C for both the flat bottom container, as well as the coned bottom container. A turnable metal screw controlled the flow out of the container into the nozzle. *Figure 2-2* shows the product flow depending on the valve opening out of the sample container.



Figure 2-2: Product flow out of the sample container for D116 at 90 °C (mean and SD, n = 3)

Flowing of the melt started at 0.75 turns with 0.23 ml/min (flat bottom) and 0.33 ml/min (coned bottom) respectively. In general, both container configurations showed a similar flow rate. At a valve opening of 1.75 turns, the flat container showed a deviating product flow of 5.5 ml/min compared to the coned bottom container with 2.6 ml/min. Increasing the valve opening to 2.0 turns, maximum flow rates of 7.1 ml/min (flat bottom container) and 7.6 ml/min (coned bottom container) were reached. Thus, the flow from the sample container was only minimally affected of the container shape. The product flow increased almost exponentially with turning of the valve.

2.4.2 **Preparation of drug-free lipid microparticles by spray congealing**

Subsequently, microparticles were prepared by spray congealing using the triglycerides D116, D118 and D120. *Table 2-1* summarizes the tested settings, as well as yields and particle sizes. The attempt to generate larger particles by addition of 1 % aluminum stearate via increasing the viscosity led to valve blockages and was not further pursued.



Figure 2-3: SEM images of triglyceride microparticles based on D116 (A), D118 (B) and D120 (C) Particles showed a spherical shape and the use of D116 led to a flaky surface appearance whereas D118 and D120 particles obtained a very smooth surface (*Figure 2-3*).

Spray pressure is reported to be a factor of major influence on particle characteristics in spray congealing [29] [30]. Yield was determined as 39.0 % (4 bar), 43.7 % (5 bar) and 42.9 % (6 bar). Particle size did not change upon varying the spray pressure. Thus, spray pressure did not have major influence on product characteristics, as the actual spray gas volume (or spray flow) getting into the nozzle was regulated separately.

Another important parameter is the lipid feed rate. The feed rate was varied in a range between 0.75 and 2.0 valve turns, corresponding to an effective product flow rate between 0.23 and 7.1 ml/min. With application of a product flow above 3 ml/min corresponding to 1.75 turns, an

irregular spray cone and a high material loss at the spray tower wall could be monitored, which reasoned slightly lower yields.

TG	Spray	Temp.	Spray flow	Valve	Yield	Mean
	pressure	[°C]	[Normlitres/h]	opening	[%]	particle size
	[bar]			[Turns]		[µm]
D116	4	90	473	0.75	39.0	40.4 (± 12.0)
D116	6	90	473	0.75	42.9	66.7 (± 28.8)
D116	5	90	473	0.75	43.7	40.2 (± 15.5)
D116	5	90	473	0.75	43.7	40.2 (± 15.5)
D116	5	90	473	1	53.2	45.1 (± 14.8)
D116	5	90	473	1.125	51.3	42.9 (± 8.0)
D116	5	90	473	1.25	46.0	48.5 (± 10.9)
D116	5	90	473	1.5	43.7	$40.0 (\pm 6.0)$
D116	5	90	473	1.75	37.8	40.8 (± 5.5)
D116	5	90	473	2	38.3	47.4 (± 2.6)
D116	5	90	301	1	23.3	43.3 (± 3.4)
D116	5	90	357	1	21.1	45.6 (± 7.0)
D116	5	90	473	1	53.2	28.9 (± 16.3)
D116	4	90	473	1	42.2	49.3 (± 12.1)
D118	4	110	473	1	34.6	35.5 (± 14.1)

Table 2-1: Overview of placebo microparticle batches produced with varied apparatus settings.

The spray flow, a factor with high influence on particle size, was assessed between 246 and 473 l/h [31]. At a spray flow below 301 l/h spray congealing was not possible because of inadequate droplet formation. At 301 and 357 l/h, yields of 22 % with particle sizes of approximately 45 μ m could be achieved. The increase of spray flow to 473 l/h resulted in the highest observed yield of about 50 % and the size of the collected particles was reduced to 28.9 μ m, indicating a more effective atomization and the formation of smaller droplets. Consequently, the spray congealing process was rather robust over a wide range of process settings. The strongest influence regarding yield and particle size was observed for the variation of spray flow.

2.4.3 Evaluation of polymorphic behavior

The lipid modification of the microparticles plays a crucial role [7] [13]. After the spray congealing process, D116 microparticles were present in the α -polymorph as indicated by the melting endotherm at 46 °C in the first and the second heating scan (*Figure 2-4 A*) [32]. This hexagonal modification with only minor and unspecific chain-chain-interactions is thermodynamically unstable upon storage [7] [33]–[35]. According to Lutton, it is only stable for several weeks at room temperature [36]. Upon further heating, the exothermic recrystallization to the higher melting β' -polymorph occurred [32]. This orthorhombic-arrayed polymorph is also metastable and not always detectable in DSC [33] [37]. Further energy led to the formation of the most stable β -form, which should be favored for reasons of long-term storage. In this triclinic modification, the carbon chains are parallel to each other with a high potential of interaction [33] [38]. The observed polymorphic behavior went in accordance with the behavior of spray congealed microparticles manufactured with different natural or synthetic lipids as well as lipid implants and solid lipid nanoparticles [6] [19] [39]–[42].

Tempering at 2-8 °C and room temperature did not lead to changes in the polymorph. Incubation at slightly increased temperatures induced a transition of the microparticles to the β -polymorph with only one single melting endotherm at 65 °C already after 24 h (*Figure 2-4 E*). Storage at 45 °C, which is slightly over the distinct melting point of the α -form, induces crystal lattice formation to the more densely packed β -form. This tempering of the lipid matrix is a suitable tool to induce formation in the thermodynamically stable β -modification for higher stability of the drug delivery systems [32] [39] [43].



Figure 2-4: Polymorphic behavior of freshly prepared D116 microparticles (A) and after 72 h stored at room temperature (B), after 72 h at 7 °C (C), after 24 h at 45 °C (C), after 48 h at 45 °C (D) and 72 h at 45 °C (E). Representative heating scans are depicted of the first and second heating each. Curves are displaced along the ordinate for better visualization

2.5 Conclusion

It was possible to produce triglyceride microparticles by the means of spray congealing. The choice of sample container had no pronounced influence on the product outflow, but could possibly minimize material loss inside the container. The evaluation of process parameters showed that the process was rather robust against their variation. The apparatus settings spray pressure, product flow and spray flow are reported to have the highest influence, which could only be partly confirmed in our experiment. Low product and spray flow led to insufficient atomization which caused a high product loss in the spray dryer. The spray flow was evaluated to be the most influencing factor on particle size and yield. DSC measurements showed that microparticles were crystalline in the α -polymorph directly after production which did not change upon storage at 2-8 and 25 °C over 72 h. Storage at 45 °C induced the transformation to the thermodynamically stable and desirable β -form after 24 h.

2.6 References

- [1] E. Gavini, P. Chetoni, M. Cossu, M. G. Alvarez, M. F. Saettone, and P. Giunchedi, "PLGA microspheres for the ocular delivery of a peptide drug, vancomycin using emulsification/spray-drying as the preparation method: in vitro/in vivo studies," *Eur. J. Pharm. Biopharm.*, vol. 57, no. 2, pp. 207–212, 2004.
- [2] T. Morita, Y. Sakamura, Y. Horikiri, T. Suzuki, and H. Yoshino, "Protein encapsulation into biodegradable microspheres by a novel S/O/W emulsion method using poly(ethylene glycol) as a protein micronization adjuvant," *J. Control. Release*, vol. 69, no. 3, pp. 435–444, 2000.
- [3] M. Mezzena, S. Scalia, P. M. Young, and D. Traini, "Solid lipid budesonide microparticles for controlled release inhalation therapy," *AAPS J.*, vol. 11, no. 4, pp. 771–778, 2009.
- [4] T. Yajima, N. Umeki, and S. Itai, "Optimum spray congealing conditions for masking the bitter taste of clarithromycin in wax matrix," *Chem. Pharm. Bull.*, vol. 47, no. 2, pp. 220–225, 1999.
- [5] N. Passerini, B. Perissutti, B. Albertini, E. Franceschinis, D. Lenaz, D. Hasa, I. Locatelli, and D. Voinovich, "A new approach to enhance oral bioavailability of Silybum Marianum dry extract: association of mechanochemical activation and spray congealing," *Phytomedicine*, vol. 19, no. 2, pp. 160–168, 2012.
- [6] H. Reithmeier, J. Herrmann, and A. Gopferich, "Lipid microparticles as a parenteral controlled release device for peptides," *J. Control. Release*, vol. 73, no. 2–3, pp. 339–350, 2001.
- [7] A. Maschke, C. Becker, D. Eyrich, J. Kiermaier, T. Blunk, and A. Gopferich, "Development of a spray congealing process for the preparation of insulin-loaded lipid microparticles and characterization thereof," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 2, pp. 175–187, 2007.
- [8] S. P. Schwendeman, M. Tobio, M. Joworowicz, M. J. Alonso, and R. Langer, "New strategies for the microencapsulation of tetanus vaccine," *J. Microencapsul.*, vol. 15, no. 3, pp. 299–318, 1998.
- [9] A. Zaky, A. Elbakry, A. Ehmer, M. Breunig, and A. Goepferich, "The mechanism of protein release from triglyceride microspheres," *J. Control. Release*, vol. 147, no. 2, pp. 202–210, 2010.
- [10] R. Cortesi, E. Esposito, G. Luca, and C. Nastruzzi, "Production of lipospheres as carriers for bioactive compounds," *Biomaterials*, vol. 23, no. 11, pp. 2283–2294, 2002.
- [11] S. Jaspart, P. Bertholet, G. Piel, J. M. Dogne, L. Delattre, and B. Evrard, "Solid lipid microparticles as a sustained release system for pulmonary drug delivery," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 1, pp. 47– 56, 2007.
- M. D. Del Curto, D. Chicco, M. D'Antonio, V. Ciolli, H. Dannan, S. D'Urso, B. Neuteboom, S. Pompili,
 S. Schiesaro, and P. Esposito, "Lipid microparticles as sustained release system for a GnRH antagonist (Antide)," *J. Control. Release*, vol. 89, no. 2, pp. 297–310, 2003.
- [13] H. Reithmeier, J. Herrmann, and A. Gopferich, "Development and characterization of lipid microparticles as a drug carrier for somatostatin," *Int. J. Pharm.*, vol. 218, no. 1–2, pp. 133–143, 2001.
- [14] J. Broadhead, C. T. Rhodes, and S. K. E. Rouan, "The spray drying of pharmaceuticals," *Drug Dev. Ind. Pharm.*, vol. 18, no. 11–12, pp. 1169–1206, 1992.
- [15] R. M. Martins, S. Siqueira, M. O. Machado, and L. A. Freitas, "The effect of homogenization method on the properties of carbamazepine microparticles prepared by spray congealing," *J. Microencapsul.*, vol. 30, no. 7, pp. 692–700, 2013.
- [16] T. Eldem, P. Speiser, and A. Hincal, "Optimization of spray-dried and -congealed lipid micropellets and characterization of their surface morphology by scanning electron microscopy," *Pharm. Res.*, vol. 8, no. 1, pp. 47–54, 1991.

- [17] P. A. McCarron, R. F. Donnelly, and R. Al-Kassas, "Comparison of a novel spray congealing procedure with emulsion-based methods for the micro-encapsulation of water-soluble drugs in low melting point triglycerides," *J. Microencapsul.*, vol. 25, no. 6, pp. 365–378, 2008.
- [18] A. L. R. Rattes and W. P. Oliveira, "Spray drying conditions and encapsulating composition effects on formation and properties of sodium diclofenac microparticles," *Powder Technol.*, vol. 171, no. 1, pp. 7– 14, 2007.
- [19] M. Di Sabatino, B. Albertini, V. L. Kett, and N. Passerini, "Spray congealed lipid microparticles with high protein loading: preparation and solid state characterisation," *Eur. J. Pharm. Sci.*, vol. 46, no. 5, pp. 346– 356, 2012.
- [20] B. Albertini, N. Passerini, F. Pattarino, and L. Rodriguez, "New spray congealing atomizer for the microencapsulation of highly concentrated solid and liquid substances," *Eur. J. Pharm. Biopharm.*, vol. 69, no. 1, pp. 348–357, 2008.
- [21] A. Fini, L. Rodriguez, C. Cavallari, B. Albertini, and N. Passerini, "Ultrasound-compacted and spraycongealed indomethacin/polyethyleneglycol systems," *Int. J. Pharm.*, vol. 247, no. 1–2, pp. 11–22, 2002.
- [22] B. Albertini, N. Passerini, M. L. Gonzalez-Rodriguez, B. Perissutti, and L. Rodriguez, "Effect of Aerosil on the properties of lipid controlled release microparticles," *J. Control. Release*, vol. 100, no. 2, pp. 233– 246, 2004.
- [23] A. Fini, C. Cavallari, G. Ceschel, and A. M. Rabasco, "Bimodal release of olanzapine from lipid microspheres," J. Pharm. Sci., vol. 99, no. 10, pp. 4251–4260, 2010.
- [24] L. Rodriguez, N. Passerini, C. Cavallari, M. Cini, P. Sancin, and A. Fini, "Description and preliminary evaluation of a new ultrasonic atomizer for spray-congealing processes," *Int. J. Pharm.*, vol. 183, no. 2, pp. 133–143, 1999.
- [25] N. Passerini, B. Albertini, B. Perissutti, and L. Rodriguez, "Evaluation of melt granulation and ultrasonic spray congealing as techniques to enhance the dissolution of praziquantel," *Int. J. Pharm.*, vol. 318, no. 1–2, pp. 92–102, 2006.
- [26] J. B. Lo, L. E. Appel, S. M. Herbig, S. B. McCray, and A. G. Thombre, "Formulation design and pharmaceutical development of a novel controlled release form of azithromycin for single-dose therapy," *Drug Dev. Ind. Pharm.*, vol. 35, no. 12, pp. 1522–1529, 2009.
- [27] M. W. Scott, M. J. Robinson, J. F. Pauls, and R. J. Lantz, "Spray Congealing: Particle Size Relationships Using a Centrifugal Wheel Atomizer," J. Pharm. Sci., vol. 53, pp. 670–675, 1964.
- [28] M. Emas and H. Nyqvist, "Methods of studying aging and stabilization of spray-congealed solid dispersions with carnauba wax. 1. Microcalorimetric investigation," *Int. J. Pharm.*, vol. 197, no. 1–2, pp. 117–127, 2000.
- [29] I. Ilic, R. Dreu, M. Burjak, M. Homar, J. Kerc, and S. Srcic, "Microparticle size control and glimepiride microencapsulation using spray congealing technology," *Int. J. Pharm.*, vol. 381, no. 2, pp. 176–183, 2009.
- [30] S. Scalia, D. Traini, P. M. Young, M. Di Sabatino, N. Passerini, and B. Albertini, "Comparison of spray congealing and melt emulsification methods for the incorporation of the water-soluble salbutamol sulphate in lipid microparticles," *Pharm. Dev. Technol.*, vol. 18, no. 1, pp. 266–273, 2013.
- [31] Büchi Labortechnik AG, "Training Papers Spray Drying." [Online]. Available: http://static1.buchi.com/sites/default/files/downloads/Set_3_Training_Papers_Spray_Drying_en_01.pdf? 996b2db24007502bd69c913b675467cfc63880ba. [Accessed: 10-Nov-2016].

- [32] M. Windbergs, C. J. Strachan, and P. Kleinebudde, "Investigating the Principles of Recrystallization from Glyceride Melts," AAPS PharmSciTech, vol. 10, no. 4, pp. 1224–1233, 2009.
- [33] D. R. Kodali, D. Atkinson, T. G. Redgrave, and D. M. Small, "Structure and polymorphism of 18-carbon fatty acyl triacylglycerols: effect of unsaturation and substitution in the 2-position.," *J. Lipid Res.*, vol. 28, no. 4, pp. 403–413, 1987.
- [34] H. Bunjes, M. H. J. Koch, and K. Westesen, "Effects of surfactants on the crystallization and polymorphism of lipid nanoparticles," *Mol. Organ. Interfaces*, vol. 121, pp. 7–10, 2002.
- [35] K. Sato and T. Kuroda, "Kinetics of melt crystallization and transformation of tripalmitin polymorphs," J. Am. Oil Chem. Soc., vol. 64, no. 1, pp. 124–127, 1987.
- [36] E. S. Lutton, "Review of the polymorphism of saturated even glycerides," J. Am. Oil Chem. Soc., vol. 27, no. 7, pp. 276–281, 1950.
- [37] M. Kellens, W. Meeussen, C. Riekel, and H. Reynaers, "Time resolved x-ray diffraction studies of the polymorphic behaviour of tripalmitin using synchrotron radiation," *Chem. Phys. Lipids*, vol. 52, no. 2, pp. 79–98, 1990.
- [38] V. D'Souza, J. M. DeMan, and L. DeMan, "Short spacings and polymorphic forms of natural and commercial solid fats: A review," J. Am. Oil Chem. Soc., vol. 67, no. 11, pp. 835–843, 1990.
- [39] F. Kreye, F. Siepmann, A. Zimmer, J. F. Willart, M. Descamps, and J. Siepmann, "Cast lipid implants for controlled drug delivery: importance of the tempering conditions," *J. Pharm. Sci.*, vol. 100, no. 8, pp. 3471–3481, 2011.
- [40] E. B. Souto, W. Mehnert, and R. H. Müller, "Polymorphic behaviour of Comprise 888 ATO as bulk lipid and as SLN and NLC.," J. Microencapsul., vol. 23, no. 4, pp. 417–33, 2006.
- [41] H. Bunjes, F. Steiniger, and W. Richter, "Visualizing the structure of triglyceride nanoparticles in different crystal modifications," *Langmuir*, vol. 23, no. 7, pp. 4005–4011, 2007.
- [42] H. Bunjes, K. Westesen, and M. H. J. Koch, "Crystallization tendency and polymorphic transitions in triglyceride nanoparticles," *Int. J. Pharm.*, vol. 129, no. 1–2, pp. 159–173, 1996.
- [43] K. Becker, S. Salar-Behzadi, and A. Zimmer, "Solvent-Free Melting Techniques for the Preparation of Lipid-Based Solid Oral Formulations," *Pharm. Res.*, vol. 32, no. 5, pp. 1519–1545, 2015.

3 INCORPORATION OF HYDROPHILIC MODEL SUBSTANCES INTO SPRAY CONGEALED SOLID LIPID MICROPARTICLES

Parts of this chapter are intended for publication

Data of amino acid-loaded microparticles was collected and evaluated by **Katrin Mayer** during her master thesis entitled "*Development and in vitro evaluation of parenteral lipid-based depot formulations*" (2014)

3.1 Abstract

Two different hydrophilic model substances, the amino acid tryptophan and the dipeptide aspartame, were incorporated into spray congealed lipid microparticles as a surrogate for the decapeptide G [6-D-Phe]. Aspartame was lyophilized to mimic the properties of the freezedried drug. The spray congealing process as well as the release behavior from the obtained microparticles were evaluated using different triglycerides, also in combination with release modifying additives, such as emulsifiers and sugars. Additionally, the influence of lipases on drug liberation from the particles was investigated. The addition of emulsifying agents led to better spray performance. A slow and incomplete release for 20 d (D116 + 5 % GMS) with an initial burst of less than 20 % could be achieved. The formulation of D118 + 10 % GMS resulted in a sustained release for 24 d, whereas the initially released drug could be kept below 25 %. The addition of lipase to the release medium led to matrix changes visible in SEM images also leading to a faster release when GMS concentrations higher than 5 % were used. The released aspartame was different compared to tryptophan and could be controlled over approx. 10 d by the addition of Span 40 and Span 85.

3.2 Introduction

The incorporation of drugs into biodegradable matrices for parenteral application is an important field within pharmaceutical research [1]–[3]. Lipid drug delivery systems, like microparticles [4], solid lipid nanoparticles [5] and implants [6] [7] are suitable tools to prolong and control the delivery of drugs. Hydrophilic proteins and peptides, like somatostatin [8], insulin [4], thymocartin (TP-4) [9], ovalbumin [6], bovine serum albumin [10] and lysozyme [11] have successfully been loaded into lipid microspheres. The spray congealing process, technically evaluated in the previous chapter, has been reported to be a suitable technique for the production of lipid microparticles [4] [10] [12]. The absence of organic solvents, high yields and encapsulation efficiencies are benefits arising from this technique compared to other emulsion-based methods [4] [9] [13].

Widely used excipients for lipid matrices are for example solid triglycerides of different chain length [14]–[17] and waxes [18]. Additionally, the application of composed fats, consisting of mono-, di- and triglycerides in variable amounts, like Compritol[®] 888 ATO [19] [20] or Gelucire[®] -qualities, combining glycerides and hydrophilic polyethylene glycols [21] [22] are common.

Release kinetics achieved with lipid delivery systems range from several minutes up to months [23] [24]. With the use of spray congealed microparticles consisting of Gelucire[®] 50/13 loaded with diclofenac a complete dissolution below 10 minutes was achieved [24]. A sustained release was reported in case of implants consisting of high-melting triglycerides for over three months [14] [25]. The modification with partial glycerides improves surface wettability, as well as increased matrix swelling which enables drug dissolution [22] [26]. The addition of poreformers, like polyethylene glycol [27] [28], sucrose [14] or trehalose [29], induces the formation of a connected pore network above a certain concentration. Test conditions, such as temperature [23] [30], addition of emulsifiers to the medium [31] [32] and the presence of lipases [33] [34] are important variables influencing release rates.

As a delivery system for release of G [6-D-Phe] over a 15 days' time frame should be developed, lipid microparticles produced by spray congealing were selected. The following study focused on the incorporation of model substances as a surrogate for the peptide drug. Surfactants, like monostearates, sorbitan esters and fat alcohols, as well as sugars like sucrose, were added to influence release rates and to adjust the formulation to the required release profile. Furthermore, suitable release conditions with or without the presence of lipase and a

suitable release method were evaluated based on previously reported dialysis membrane models for oily suspensions and other colloidal systems [35] [36].

3.3 Materials and Methods

3.3.1 Materials

The peptide drug aspartame (ASP) was obtained from Sandoz AG, Nuremberg, Germany. The amino acid tryptophan (TRP) was purchased from Sigma-Aldrich (Taufkirchen, Germany). Triglycerides Dynasan 112 (trilaurin, D112), Dynasan 114 (trimyristin, D114), Dynasan 116 (tripalmitin, D116) and Dynasan 118 (tristearin, D118) were kindly provided from Cremer Oleo (Witten, Germany). Glycerol monostearate (GMS) with a monoester content of 40-55 % was purchased from Caelo (Caesar & Loretz, Hilden Germany). Cetylstearyl alcohol (CSA, Lanette[®] O) and Poloxamer 188 (Lutrol[®] F 68) were purchased from BASF (Ludwigshafen, Germany). Span 40, 65 and 85 were kindly provided from Croda (Nettetal, Germany). PEG-24-glyceryl stearate (PEG-24 GS), tradename Cutina® E 24, was a kind gift from BASF (Ludwigshafen, Germany). Colloidal silicium dioxide (Aerosil[®]) was purchased from Evonik Industries (Essen, Germany). Sucrose was purchased from Südzucker (Mannheim, Germany). Ultrapure deionized water with a conductivity of less than 0.055 µS/cm (Milli-Q Water systems, Millipore, MA, USA) was used for preparation of buffers in this study. Visking dialysis tubings with a diameter of 1.6 mm and a cut-off of 12-14 kDa for performance of release studies were sourced from Serva Electrophoresis GmbH (Heidelberg, Germany). Centrifuge tubes were purchased from VWR International GmbH (Darmstadt, Germany). Lipase isolated from *rhizopus oryzae* was purchased from Sigma-Aldrich (Taufkirchen, Germany). All other chemicals were of analytical grade.

3.3.2 Lyophilization of ASP

ASP was bulk lyophilized in a concentration of 0.5 % (w/v) on a tray without addition of further stabilizers using an Epsilon 2-6 D freeze dryer (Martin Christ Gefriertrocknungsanlagen GmbH, Osterode, Germany) and the following cycle:

Step	Ramp [°C/min]	Temperature [°C]	Pressure	Hold time
			[Torr]	[min]
Freezing	1	-50	750	60
Annealing	1	-20	750	240
Freezing	1	-50	750	60
Primary drying	0.5	-20	0.0675	2100
Secondary	0.1	30	0.0675	720
drying				

3.3.3 Cryogenic ball-milling of ASP

Prior to the milling step, ASP was blended with 0.5 % (w/w) Aerosil[®] to enable particle separation during the milling process. The cryogenic milling was conducted with the Cryomill (Retsch[®] Technologies GmbH, Haan, Germany) using a stainless steel beaker and two milling balls with a diameter of 10 mm. A precooling time of 10 min at 5 Hz was applied, followed by two milling cycles of 2 min each at a frequency of 20 Hz. The milled mixture was aliquoted and stored at 2-8 °C until spray congealing was performed.

3.3.4 Mortar grinding of TRP for suspension in the lipid melt

Approximately 7 g of lipid components consisting of triglycerides (D116, D118) and the additive (GMS or sucrose) were ground together with 130 mg (1 %) or 260 mg (2 %) TRP manually for 30 min using mortar and pestle without heating. The remainder of the lipid components (total batch volume 13 g) were added and the mixture heated up on a water bath to 90 °C. The mixture was homogenized for 1 min using an Ultraturrax T-10 basic (IKA Laboratory Technology, Staufen, Germany).

3.3.5 Preparation of model substance-loaded microparticles by spray congealing

For preparation of TRP-loaded microparticles, the lipid dispersion was filled in the sample container of the B-290 Mini spray dryer with additional spray chilling setup (Büchi, Flawil, Switzerland), which was pre-conditioned for 1.5 h prior to the production step (sample container 90-110 °C, spray tower temperature 15 °C). The additional equipment consisted of a sample container, surrounded by a bath filled with polyethylene glycol (PEG) 400, also heating the nozzle. The connection between nozzle and sample container was heated externally by a

resistance wire-based heating, generating a temperature of approximately 70 °C. The spray tower was fed with pre-cooled air produced by an additional dehumidifier Deltatherm[®] LT (Deltatherm[®] Hirmer GmbH, Much, Germany). The lipid melt was atomized using nitrogen as spray gas at a pressure of 6 bar and a modified nozzle with an inner diameter of 2.5 mm.

ASP-loaded microparticles were prepared by melting lipid components and adding the milled ASP under high shear mixing using an Ultraturrax for 2 min. The suspension was fed into the sample container of the spray dryer and atomized in analogy to amino acid-loaded microparticles. The theoretical drug load was 1.67 %.

3.3.6 **Determination of particle size**

Approximately 20 mg of microparticles were suspended in 10 ml of an aqueous solution of 0.02 % PS 20 prior to analysis. The dispersion was analyzed using the laser diffraction system Horiba Partica LA-950 (Horiba, Kyoto, Japan) while stirring.

3.3.7 Characterization of particle morphology using scanning electron microscopy (SEM)

Particle characteristics of lipid microparticles were assessed using a Jeol JSM-6500 F SEM (Jeol, Tokyo, Japan) without additional coating of the particles after attaching on aluminum sample holders with double adhesive tape (Plano, Wetzlar, Germany). Cross sections of particles were prepared by cutting the particles with the use of a surgical scalpel by hand.

3.3.8 **Determination of encapsulation efficiency**

30 mg of TRP-loaded microparticles were dissolved in 2.0 ml of methylene chloride and 2.0 ml of PBS pH 7.4 were added. Prior to analysis of the water phase, the two-phase system was shaken vigorously for 5 min and let stand afterwards for another 30 min. The aqueous phase was analyzed using the NanoDrop 2000 spectrophotometer at 280 nm [37] (Thermo Fisher Scientific, Waltham, MA, USA). Each batch was extracted in triplicate.

50 mg of ASP-loaded microparticles were dissolved in 0.5 ml of methylene chloride and the same amount of highly purified water was added. The mixture was incubated for 18 h on the GFL 3031 horizontal shaking incubator (Gesellschaft für Labortechnik, Burgwedel, Germany) at a temperature of 39 °C and 60 rpm. All microparticular batches were extracted in triplicate and measured using an Agilent UV-Vis spectrophotometer 8453 (Agilent Technologies, Santa Clara, CA, USA) at 258 nm [38].

3.3.9 Development of a suitable release model and evaluation of TRP and ASP release behavior

For release studies of TRP-loaded microparticles 100 mg with 2 % and 200 mg with 1 % drug load were weighed into dialysis tubings. After addition of 0.5 ml (100 mg microparticles) or 1.0 ml (200 mg microparticles) PBS pH 7.4 with 0.05 % NaN₃ and 0.02 % Poloxamer 188, the tubings were closed at both ends and placed in a 15-ml centrifuge tube, release medium was added to a total volume of 5.0 ml. Samples were incubated on a horizontal shaker at 39 °C and 60 rpm. Additionally, the same experiment was carried out with addition of lipase from *rhizopus oryzae* in a concentration of 1.0 U/mg microparticles (corresponding to a concentration of 20 or 40 U/ml, respectively) in the dialysis membrane. All release studies were performed in triplicate. Samples of 100 μ l per time point were taken, replaced with fresh release medium and analyzed using a NanoDrop 2000 spectrophotometer at 280 nm. 100 mg of microparticles without amino acid were used as control. A solution of 1 mg/ml TRP in PBS was filled into the membrane as positive control to proof the permeability through the membrane.

100 mg of ASP-loaded microparticles were filled in dialysis tubings and 2 ml of PBS buffer pH 7.4 were added, tubings were closed with threads at both ends and placed in 15-ml centrifuge tubes. 3 ml of release medium was added and samples were treated according to TRP-loaded microparticles. Each sample was performed and analyzed in triplicate. Samples of 100 μ l were taken for measurement of released drug using an Agilent spectrophotometer 8453 at a wavelength of 258 nm.

3.4 Results and Discussion

3.4.1 Preparation and characterization of model drug-loaded lipid microparticles by spray congealing

All TRP-loaded batches could be sprayed successfully resulting in different yield and particle size (*Table 3-1*). An increased spray flow of 473 l/h compared to 414 l/h did not influence the microparticle size. Both concentrations of TRP could be sprayed without affecting yield. Microparticles were obtained at yields in a range between 16.8 (D116) and 29.0 % (D118) with particle sizes between 30.2 and 53.9 μ m. Addition of GMS resulted in slightly larger particles. With addition of GMS, yield was not affected for D118-based microparticles, but slightly increased for the formulations based on D116. The emulsifier GMS is reported to enable a more effective encapsulation, which may result in larger particles [39].

Model	Drug Load	Spray	TG	Excipient	[%]	Yield	Mean Particle
drug	[%] (w/w)	Flow [l/h]				[%]	Size [µm]
TRP	2	414	D118	-	-	29.0	30.2
TRP	1	414	D118	GMS	5	22.9	49.7
TRP	1	414	D118	GMS	10	22.0	42.4
TRP	2	473	D116	-	-	16.8	41.8
TRP	2	473	D116	Sucrose	1	25.5	39.0
TRP	2	473	D116	GMS	5	28.8	50.6
TRP	2	473	D116	GMS	10	25.9	53.9

Table 3-1: Overview of TRP-loaded microparticle batches sprayed with different apparatus settings and with addition of GMS and sucrose

The addition of sucrose as pore former resulted in an irregular spray cone and valve blockages and was therefore not further investigated. *Figure 3-1* depicts SEM images of triglyceride-based microparticles loaded with TRP. Microparticles showed an overall smooth surface with small crystals. Cross-sections confirmed a dense, compact matrix without noticeable cavities. The incorporated TRP could not be visualized in contrast to other researchers [25] [40].



Figure 3-1: SEM images of TRP-loaded microparticles (D116: A, D, D118 + 1 % sucrose: B, E, D118 + 5 % GMS: C, F)

ASP lyophilisate could not be incorporated into spray congealed microparticles. Upon homogenization using Ultraturrax, the viscosity of the melt increased substantially and could not be sprayed. For this reason, all other batches were prepared using cryomilled ASP. These formulations could be spray congealed successfully (*Table 3-2*).

High yield of approx. 50 % were obtained for D116 with addition of 5 and 15 % CSA achieving values of 50 % usable product. Span 65 and 85 caused a decrease in yield. With addition of the low melting triglyceride D112 only 10 % product was obtained. This effect was less pronounced when combinations of D112 with GMS were used. The highest product yield was achieved with addition of 20 % PEG-24 GS. Particle size was exemplarily measured for triglyceride particles with addition of two intermediate and one very lipophilic emulsifiers. A slightly higher particle size was observed with the use of emulsifiers with lowest HLB values (Span 85, HLB 1.8 [41]) as well as the production with pure triglycerides. Using emulsifiers with higher HLB values (Span 40 and 65), particle size slightly decreased indicating a more ineffective drug encapsulation [42] [43].

Triglyceride	Excipient	[%]	Yield [%]	Mean particle
				size [µm]
D116	GMS (s)	5	37.0	-
D116	GMS (s)	10	51.9	-
D116	GMS (s)	15 %	38.1	-
D116	CSA (s)	5	50.5	-
D116	CSA (s)	15	50.3	-
D116	Span 40 (s)	5	44.3	42.7 (± 1.6)
D116	Span 40 (s)	10	48.1	33.3 (± 1.1)
D116	Span 40 (s)	20	43.7	-
D116	Span 65 (s)	5	28.9	25.0 (± 1.6)
D116	Span 65 (s)	10	43.9	32.4 (± 2.3)
D116	Span 65 (s)	20	42.2	21.0 (± 1.9)
D116	Span 85 (1)	2	7.6	84.6 (± 5.0)
D116	Span 85 (l)	5	23.9	62.5 (± 12.9)
D116	Span 85 (1)	10	20.4	65.8 (± 27.7)
D116	PEG-24 GS (ss)	0.7	26.3	-
D116	PEG-24 GS (ss)	10	48.8	-
D116	PEG-24 GS (ss)	20 %	74.3	-
D116	D112 (s)	10	10.4	69.6 (± 3.2)
D116	GMS/D112 (s/s)	10/5	40.4	-
D116	GMS/D112 (s/s)	10/10	42.9	-
D114	GMS/D112 (s/s)	10/5	39.3	-
D114	GMS/D112 (s/s)	10/10	50.3	-

Table 3-2: Overview of 1.67 % ASP-loaded microparticle batches (s = solid, l = liquid, ss = semi-solid); Three emulsifies with different HLB values were investigated concerning particle size exemplarily (data shown mean and SD, n = 3)

3.4.2 In vitro release behavior of TRP-loaded microparticles

Figure 3-2 shows the release profiles of TRP from pure triglyceride microspheres and with 5 or 10 % GMS. Pure triglyceride microparticles showed a very low initial burst release (≤ 10 %) within the first day, which was also stated by other researchers when investigating peptide and protein release from triglyceride microparticles and cylinders [9] [44]. A high amount of drug remained inside the particles and the total released drug after three weeks was about 15 %. With addition of 5 % GMS to D116 the burst release increased to approx. 15 %. A continuous release

phase followed until day 20 reaching 45 % total release. With 10 % GMS, 40 % of incorporated TRP was released within the first day of the study and a steady release which slowed down after approximately one week followed. The incorporated TRP was almost completely released. McCarron and co-workers also found enhanced release rates for lipid particles in combination with emulsifiers [45]. Penetration appears to be enhanced with the higher emulsifier content.



Figure 3-2: Release of TRP from D116 (A) and D118 (B) microparticles without and with 5 or 10 % GMS in PBS pH 7.4 (mean and SD, n = 3)

Particles based on pure D118 exhibited an initial TRP burst of less than 15 % and total drug release of approx. 20 %. Addition of 5 % GMS resulted in a higher burst of around 40 %, followed by further release of 5-10 % until day 6. In total, about 50 % of TRP was released. With 10 % GMS, the initial burst was low, followed by a lag time until day 10 and a continuous release phase until day 24. The total amount of drug released was 45 %. The incomplete drug release might be due to, on the one hand, an insufficient emulsifier concentration to facilitate the penetration of medium into the core of the particle or the adsorption to the lipid in solution [8] [46]. Incorporation of cholesterol and mono- or diglycerides has been described as suitable means to tailor release kinetics from triglyceride matrices [26] [47]. The concentration dependent emulsifier effect was also shown by other researchers who found enhanced release of hydrophilic drugs, like diclofenac [39]. Thus, the release can be adapted by the addition of a W/O surfactant.

3.4.3 Influence of lipase addition to the release medium on TRP release

Lipases are present in a variety of organs of the mammalian body and digest lipids typically by cleaving fatty acids from the triglyceride backbone [48]. The activity in the human body differs depending on the organ and is reported to be 0.01 U/ml in serum and subcutaneous fat [33]. In this study, the lipase concentration was set at 20 U/ml to see the effect during the 20 day-study period more clearly.



Figure 3-3: Release of TRP from D116 (A) and D118 (B) microparticles with 0, 5 and 10 % GMS in PBS pH 7.4 containing 40 U/ml lipase (mean and SD, n = 3)

Lipase addition resulted in slightly enhanced drug release from D116-based microparticles (*Figure 3-3*). Compared to the release in PBS buffer, the burst was increased by about 10 % and a continuous release could be monitored until day 11. In total, 25 % of incorporated drug were released, approximately 10 % more than in absence of lipase. For D116 particles with GMS, a general trend to a faster release could not be stated. Slightly enhanced release profiles could be observed regarding the pure triglyceride formulation and the combination with 5 % GMS. The most pronounced effect was noticed for microparticles formed with 10 % GMS. A burst of 40 % was followed by a continuous drug release until the end of the monitored time. The addition of higher amounts of monostearates enhanced the digestion. Surface structure was affected by the presence of lipase (*Figure 3-4*). Whereas microparticles in absence of lipase kept the initially observed smooth surface with slightly increased roughness potentially due to swelling [26], particle surface became uneven with larger flaky structures [47] and a more porous surface [33] [49] in presence of lipase. Keeping the lipase activity in the body in mind, the effect of lipase in vivo is most likely only minimal.



Figure 3-4: D118 + 5 % GMS-based microparticles after 3-weeks incubation in PBS pH 7.4 without lipase (A) and with 40 U/ml lipase (B)

3.4.4 Evaluation of lipid composition and additive concentration on in vitro release behavior of ASP from lipid-based microparticles

In addition to the release of TRP, particles containing the dipeptide ASP were tested. *Figure 3-5* shows selected ASP release profiles. Specifically, the influence of surfactants as release modifiers has been studied [50]. Vilivalam and Adeyeye showed, that the addition of stearic acid led to an enhanced release of diclofenac from waxy microspheres [39]. Furthermore, the hydrophilic and lipophilic properties of the drug influence the ability to be released from the matrix. More hydrophilic drugs are dissolved easier out of a lipid matrix than more lipophilic ones [20]. We studied the effect of both W/O or O/W-emulsifiers on the release behavior.



Figure 3-5: Release of ASP from D114 and D116 microparticles blended with emulsifying additives and/or low melting triglycerides in PBS pH 7.4 (mean and SD, n = 3)

The addition of GMS to D116 did not result in an improved release profile but only a burst of 15-25 % without further release. The use of CSA with a lower HLB value showed a comparable effect at both tested concentrations as a sustained release profile could not be obtained. To achieve a more complete ASP release, combinations of GMS and D112 were tested using the base lipids D116 and D114. Again, the desired profile could not be obtained. PEG-24 GS, an O/W emulsifier, led to a more complete drug release at higher concentrations, but did not render a sustained release profile. Khan and co-workers investigated the Gelucire[®] family, consisting of glycerides in combination with PEG-esters. Depending on the composition, a swelling and enhanced water penetration could be achieved leading to a faster drug release [51].

49

Most promising results could be obtained with the use of sorbitan fatty acid esters. Span 40, with an HLB value of 6.7 [41], resulted in promising profiles with a release up to 10 d when added in concentrations of 5 and 10 % to D116. 5 and 10 % Span 85, with an HLB value of 1.8 [41] rendered a sustained release profile for 12 d with an initial burst of 15 and 30 % respectively for the 5 and 10 %-addition. Thus, with ASP as model compound, no adequate release profile over 15 d could be achieved. Comparing the TRP and ASP release profiles led to the assumption, that the release behavior of G [6-D-Phe] is hard to predict. Drug properties appear to play an important role. Having been able to find the required release profile using TRP in a triglyceride formulation using GMS, this approach seemed most suitable for G [6-D-Phe]. An addition of Span 40 and 85 could form a promising base for further development.

3.5 Conclusion

The spray congealing process was determined as suitable production process for the preparation of TRP- and ASP-loaded microparticles. In general, microparticles with a spherical shape and smooth surface could be produced. Variation of process parameters did not result in pronounced changes regarding yield and particle size, confirming an adequate robustness of the process. Exclusive use of triglycerides D116 and D118 did not result in adequate release profiles, as a preferably low burst was not followed by a sustained release. With addition of 5 and 10 % GMS to D116 and D118, release profiles over 20 d could be obtained for TRP. The addition of lipases to the release medium led to a slightly accelerated and more complete release, but the enzymatic degradation is expected to have little influence in vivo. None of the tested formulations using ASP as model substance showed the aspired release profile. In most of the cases, a continuous release could not be observed. Formulations which could provide a springboard for the following study using G [6-D-Phe] could be based on D116 and D118 with 10 % GMS.

3.6 References

- V. R. Sinha and A. Trehan, "Biodegradable microspheres for protein delivery," *J. Control. Release*, vol. 90, no. 3, pp. 261–280, 2003.
- [2] W. Jiskoot, T. W. Randolph, D. B. Volkin, C. R. Middaugh, C. Schöneich, G. Winter, W. Friess, D. J. A. Crommelin, and J. F. Carpenter, "Protein instability and immunogenicity: Roadblocks to clinical application of injectable protein delivery systems for sustained release," *J. Pharm. Sci.*, vol. 101, no. 3, pp. 946–954, 2012.
- [3] B. Appel, A. Maschke, B. Weiser, H. Sarhan, C. Englert, P. Angele, T. Blunk, and A. Gopferich, "Lipidic implants for controlled release of bioactive insulin: effects on cartilage engineered in vitro," *Int. J. Pharm.*, vol. 314, no. 2, pp. 170–178, 2006.
- [4] A. Maschke, C. Becker, D. Eyrich, J. Kiermaier, T. Blunk, and A. Gopferich, "Development of a spray congealing process for the preparation of insulin-loaded lipid microparticles and characterization thereof," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 2, pp. 175–187, 2007.
- [5] A. J. Almeida and E. Souto, "Solid lipid nanoparticles as a drug delivery system for peptides and proteins," *Adv. Drug Deliv. Rev.*, vol. 59, no. 6, pp. 478–490, 2007.
- [6] M. P. Even, K. Young, G. Winter, S. Hook, and J. Engert, "In vivo investigation of twin-screw extruded lipid implants for vaccine delivery," *Eur. J. Pharm. Biopharm.*, vol. 87, no. 2, pp. 338–346, 2014.
- S. Mohl and G. Winter, "Continuous release of rh-interferon alpha-2a from triglyceride matrices," J. Control. Release, vol. 97, no. 1, pp. 67–78, 2004.
- [8] H. Reithmeier, J. Herrmann, and A. Gopferich, "Development and characterization of lipid microparticles as a drug carrier for somatostatin," *Int. J. Pharm.*, vol. 218, no. 1–2, pp. 133–143, 2001.
- [9] H. Reithmeier, J. Herrmann, and A. Gopferich, "Lipid microparticles as a parenteral controlled release device for peptides," *J. Control. Release*, vol. 73, no. 2–3, pp. 339–350, 2001.
- [10] M. Di Sabatino, B. Albertini, V. L. Kett, and N. Passerini, "Spray congealed lipid microparticles with high protein loading: preparation and solid state characterisation," *Eur. J. Pharm. Sci.*, vol. 46, no. 5, pp. 346– 356, 2012.
- [11] P. C. Christophersen, L. Zhang, M. Yang, H. M. Nielsen, A. Mullertz, and H. Mu, "Solid lipid particles for oral delivery of peptide and protein drugs I--elucidating the release mechanism of lysozyme during lipolysis," *Eur. J. Pharm. Biopharm.*, vol. 85, no. 3 Pt A, pp. 473–480, 2013.
- [12] T. Yajima, N. Umeki, and S. Itai, "Optimum spray congealing conditions for masking the bitter taste of clarithromycin in wax matrix," *Chem. Pharm. Bull.*, vol. 47, no. 2, pp. 220–225, 1999.
- [13] T. Eldem, P. Speiser, and A. Hincal, "Optimization of spray-dried and -congealed lipid micropellets and characterization of their surface morphology by scanning electron microscopy," *Pharm. Res.*, vol. 8, no. 1, pp. 47–54, 1991.
- [14] C. Guse, S. Koennings, F. Kreye, F. Siepmann, A. Goepferich, and J. Siepmann, "Drug release from lipidbased implants: elucidation of the underlying mass transport mechanisms," *Int. J. Pharm.*, vol. 314, no. 2, pp. 137–144, 2006.
- [15] C. Guse, S. Koennings, A. Maschke, M. Hacker, C. Becker, S. Schreiner, T. Blunk, T. Spruss, and A. Goepferich, "Biocompatibility and erosion behavior of implants made of triglycerides and blends with cholesterol and phospholipids," *Int. J. Pharm.*, vol. 314, no. 2, pp. 153–160, 2006.
- [16] H. Bunjes and M. H. Koch, "Saturated phospholipids promote crystallization but slow down polymorphic

transitions in triglyceride nanoparticles," J. Control. Release, vol. 107, no. 2, pp. 229-243, 2005.

- [17] V. Venkateswarlu and K. Manjunath, "Preparation, characterization and in vitro release kinetics of clozapine solid lipid nanoparticles," J. Control. Release, vol. 95, no. 3, pp. 627–638, 2004.
- [18] B. Albertini, N. Passerini, F. Pattarino, and L. Rodriguez, "New spray congealing atomizer for the microencapsulation of highly concentrated solid and liquid substances," *Eur. J. Pharm. Biopharm.*, vol. 69, no. 1, pp. 348–357, 2008.
- M. D. Del Curto, D. Chicco, M. D'Antonio, V. Ciolli, H. Dannan, S. D'Urso, B. Neuteboom, S. Pompili,
 S. Schiesaro, and P. Esposito, "Lipid microparticles as sustained release system for a GnRH antagonist (Antide)," *J. Control. Release*, vol. 89, no. 2, pp. 297–310, 2003.
- [20] L. Rodriguez, N. Passerini, C. Cavallari, M. Cini, P. Sancin, and A. Fini, "Description and preliminary evaluation of a new ultrasonic atomizer for spray-congealing processes," *Int. J. Pharm.*, vol. 183, no. 2, pp. 133–143, 1999.
- [21] N. Passerini, B. Perissutti, M. Moneghini, D. Voinovich, B. Albertini, C. Cavallari, and L. Rodriguez, "Characterization of carbamazepine-Gelucire 50/13 microparticles prepared by a spray-congealing process using ultrasounds," *J. Pharm. Sci.*, vol. 91, no. 3, pp. 699–707, 2002.
- [22] W. Sutananta, D. Q. Craig, and J. M. Newton, "An evaluation of the mechanisms of drug release from glyceride bases," *J. Pharm. Pharmacol.*, vol. 47, no. 3, pp. 182–187, 1995.
- [23] G. Sax and G. Winter, "Mechanistic studies on the release of lysozyme from twin-screw extruded lipid implants," J. Control. Release, vol. 163, no. 2, pp. 187–194, 2012.
- [24] C. Cavallari, L. Rodriguez, B. Albertini, N. Passerini, F. Rosetti, and A. Fini, "Thermal and fractal analysis of diclofenac/Gelucire 50/13 microparticles obtained by ultrasound-assisted atomization," *J. Pharm. Sci.*, vol. 94, no. 5, pp. 1124–1134, 2005.
- [25] F. Kreye, F. Siepmann, A. Zimmer, J. F. Willart, M. Descamps, and J. Siepmann, "Cast lipid implants for controlled drug delivery: importance of the tempering conditions," *J. Pharm. Sci.*, vol. 100, no. 8, pp. 3471–3481, 2011.
- [26] W. Vogelhuber, E. Magni, A. Gazzaniga, and A. Gopferich, "Monolithic glyceryl trimyristate matrices for parenteral drug release applications," *Eur. J. Pharm. Biopharm.*, vol. 55, no. 1, pp. 133–138, 2003.
- [27] S. Güres and P. Kleinebudde, "Dissolution from solid lipid extrudates containing release modifiers," *Int. J. Pharm.*, vol. 412, no. 1–2, pp. 77–84, 2011.
- [28] S. Herrmann, G. Winter, S. Mohl, F. Siepmann, and J. Siepmann, "Mechanisms controlling protein release from lipidic implants: Effects of PEG addition," *J. Control. Release*, vol. 118, no. 2, pp. 161–168, 2007.
- [29] S. E. Reinhold and S. P. Schwendeman, "Effect of polymer porosity on aqueous self-healing encapsulation of proteins in PLGA microspheres," *Macromol. Biosci.*, vol. 13, no. 12, pp. 1700–1710, 2013.
- [30] S. Koennings, A. Sapin, T. Blunk, P. Menei, and A. Goepferich, "Towards controlled release of BDNF -Manufacturing strategies for protein-loaded lipid implants and biocompatibility evaluation in the brain," *J. Control. Release*, vol. 119, no. 2, pp. 163–172, 2007.
- [31] A. Zaky, A. Elbakry, A. Ehmer, M. Breunig, and A. Goepferich, "The mechanism of protein release from triglyceride microspheres," *J. Control. Release*, vol. 147, no. 2, pp. 202–210, 2010.
- [32] C. Reitz, C. Strachan, and P. Kleinebudde, "Solid lipid extrudates as sustained-release matrices: The effect of surface structure on drug release properties," *Eur. J. Pharm. Sci.*, vol. 35, no. 4, pp. 335–343, 2008.
- [33] M. Schwab, G. Sax, S. Schulze, and G. Winter, "Studies on the lipase induced degradation of lipid based
drug delivery systems," J. Control. Release, vol. 140, no. 1, pp. 27-33, 2009.

- [34] C. Olbrich, O. Kayser, and R. H. Müller, "Enzymatic degradation of Dynasan 114 SLN Effect of surfactants and particle size," J. Nanoparticle Res., vol. 4, no. 1–2, pp. 121–129, 2002.
- [35] S. W. Larsen, A. B. Frost, J. Østergaard, H. Marcher, and C. Larsen, "On the mechanism of drug release from oil suspensions in vitro using local anesthetics as model drug compounds," *Eur. J. Pharm. Sci.*, vol. 34, no. 1, pp. 37–44, 2008.
- [36] C. Washington, "Evaluation of non-sink dialysis methods for the measurement of drug release from colloids: effects of drug partition," *Int. J. Pharm.*, vol. 56, no. 1, pp. 71–74, 1989.
- [37] H. Mach, C. R. Middaugh, and R. V. Lewis, "Statistical determination of the average values of the extinction coefficients of tryptophan and tyrosine in native proteins," *Anal. Biochem.*, vol. 200, no. 1, pp. 74–80, 1992.
- [38] B. S. Dattatreya, D. V Usha, N. S. Susheelamma, and K. K. Bhat, "Aspartame: studies on UV spectral characteristics," *Int. J. Food Sci. Technol.*, vol. 38, no. 7, pp. 767–775, 2003.
- [39] V. D. Vilivalam and C. M. Adeyeye, "Development and evaluation of controlled-release diclofenac microspheres and tabletted microspheres.," J. Microencapsul., vol. 11, no. 4, pp. 455–470, 1994.
- [40] R. Bodmeier, J. Wang, and H. Bhagwatwar, "Process and formulation variables in the preparation of wax microparticles by a melt dispersion technique. II. W/O/W multiple emulsion technique for water-soluble drugs.," J. Microencapsul., vol. 9, no. 1, pp. 99–107, 1992.
- [41] T. Yoshioka, B. Sternberg, and A. T. Florence, "Preparation and properties of vesicles (niosomes) of sorbitan monoesters (Span 20, 40, 60 and 80) and a sorbitan triester (Span 85)," *Int. J. Pharm.*, vol. 105, no. 1, pp. 1–6, 1994.
- [42] S. Jaspart, G. Piel, L. Delattre, and B. Evrard, "Solid lipid microparticles: formulation, preparation, characterisation, drug release and applications," *Expert Opin. Drug Deliv.*, vol. 2, no. 1, pp. 75–87, 2005.
- [43] W. C. Griffin, "Classification of surface-active agents by 'HLB," *Chicago Chapter Meeting*. Journal of the society of cosmetic chemists, Chicago, pp. 311–326, 1949.
- [44] W. Vogelhuber, E. Magni, M. Mouro, T. Spruss, C. Guse, A. Gazzaniga, and A. Gopferich, "Monolithic triglyceride matrices: a controlled-release system for proteins," *Pharm. Dev. Technol.*, vol. 8, no. 1, pp. 71–79, 2003.
- [45] P. A. McCarron, R. F. Donnelly, and R. Al-Kassas, "Comparison of a novel spray congealing procedure with emulsion-based methods for the micro-encapsulation of water-soluble drugs in low melting point triglycerides," *J. Microencapsul.*, vol. 25, no. 6, pp. 365–378, 2008.
- [46] C. M. Adeyeye and J. C. Price, "Development and Evaluation of Sustained-Release Ibuprofen–Wax Microspheres. II. In Vitro Dissolution Studies," *Pharm. Res.*, vol. 11, no. 4, pp. 575–579, 1994.
- [47] P. C. Christophersen, D. Vaghela, A. Müllertz, M. Yang, H. M. Nielsen, and H. Mu, "Solid lipid particles for oral delivery of peptide and protein drugs III — the effect of fed state conditions on the in vitro release and degradation of desmopressin," AAPS J., vol. 16, no. 4, pp. 875–883, 2014.
- [48] R. H. Müller, D. Rühl, and S. A. Runge, "Biodegradation of solid lipid nanoparticles as a function of lipase incubation time," *Int. J. Pharm.*, vol. 144, no. 1, pp. 115–121, 1996.
- [49] M. Schwab, C. M. McGoverin, K. C. Gordon, G. Winter, T. Rades, J. Myschik, and C. J. Strachan, "Studies on the lipase-induced degradation of lipid-based drug delivery systems. Part II - Investigations on the mechanisms leading to collapse of the lipid structure," *Eur. J. Pharm. Biopharm.*, vol. 84, no. 3, pp. 456–

463, 2013.

- [50] Y. Akiyama, M. Yoshioka, H. Horibe, S. Hirai, N. Kitamori, and H. Toguchi, "Novel oral controlledrelease microspheres using polyglycerol esters of fatty acids," J. Control. Release, no. 26 (1993), pp. 1 – 10, 1993.
- [51] N. Khan and D. Q. Craig, "The influence of drug incorporation on the structure and release properties of solid dispersions in lipid matrices," J. Control. Release, vol. 93, no. 3, pp. 355–368, 2003.

4 OPTIMIZATION OF THE SPRAY CONGEALING PROCESS WITH FOCUS ON YIELD AND DRUG LOAD

Parts of this chapter are intended for publication

Data shown in this chapter was collected by **Kay Uwe Kraft** during his Bachelor thesis entitled *"Optimization of the spray congealing process for lipid microparticles"* (2015)

4.1 Abstract

The spray congealing process was evaluated concerning material loss, yield, particle size of different fractions deposited in the spray dryer and drug load using aspartame as model substance. The apparatus settings spray flow, aspirator power, melt temperature and installation of an additional stirrer were varied. Viscosities of lipid melts were determined using a rotational viscometer. Viscosity was mainly influenced by the temperature of the melt and only marginally by the lipid composition. At a spray flow of 414 l/h, a product yield of approximately 25 % resulted with substantial material loss in the spray tower and the melt container. Other components like high-performance cyclone, filter and connection between spray tower and cyclone were not identified as critical parts. Increasing the spray flow to 601 l/h resulted in higher product yield of approx. 45 % whereas a decrease in spray flow to 301 l/h and a reduction of the aspirator capacity reduced the yield. Melt temperature reduction (yield approx. 40 %) and installation of a stirrer within the sample container (yield approx. 50 %) had positive effects on product yield. Mean particle size changed with spray flow, from 70 (301 l/h) over 45 (414 l/h) to 20.7 µm (601 l/h). Highest encapsulation efficiencies in the product could be achieved with reduction of the spray flow to 301 l/h, as well as with the reduction of the melt temperature from 90 to 75 °C.

4.2 Introduction

The spray congealing process is a widely used technique suitable for encapsulation of drugs [1]–[3]. Having been able to develop a method suitable for the encapsulation of the amino acid tryptophan as well as the dipeptide aspartame, the process and influencing parameters should be further investigated. A lot of experimental data and knowledge is available on the influence of different nozzle types on the process and the product, but only little research has been done with the focus on yield distribution and material loss [4]–[7]. Particle size is typically influenced by spray flow and temperature difference between the melt and tower [8]. In general, for the spray congealing process rather high yields are reported [4], [9]–[11]. Additionally, the drug load depends on the apparatus settings. In general it is stated, that a faster solidification of microparticles leads to a desirably high entrapment efficiency [12] [13].

This chapter deals with the adjustment of the process with focus on the reduction of material loss, improvement of yield, a homogenous drug distribution within the melt, as well as particle sizes adequate for veterinary i.m. injection.

4.3 Materials and Methods

4.3.1 Materials

Triglycerides Dynasan 114 (trimyristin, D114) and Dynasan 116 (tripalmitin, D116) were kindly provided from Cremer Oleo (Witten, Germany). Aspartame (ASP) was purchased from Sandoz AG (Nuremberg, Germany). Glycerol monostearate (GMS) with a monoester content of 40-55 % was supplied from Caelo (Caesar & Loretz GmbH, Hilden, Germany). Colloidal silicium dioxide (Aerosil[®]) was purchased from Evonik (Hanau, Germany). Sorbitan monopalmitate (Span 40) was purchased from Croda (Nettetal, Germany). Polysorbate 20 (PS 20) was supplied from Merck KGaA (Darmstadt, Germany). All other chemicals were of analytical grade.

4.3.2 Cryogenic milling of ASP

ASP was blended with 0.5 % of colloidal silicium dioxide prior to being loaded into stainless steel milling beakers of the Cryomill (Retsch[®] Technologies GmbH, Haan, Germany). The powder was milled using two stainless steel milling balls with a diameter of 10 mm. A precooling time of 10 min at 5 Hz was followed by the actual milling step for 2 x 2 min at 20 Hz.

4.3.3 Preparation of triglyceride-based microparticles by spray congealing

Lipid components were melted on a hot plate, 100 mg (0.83 % drug load) or 200 mg (1.67 % drug load) ASP were added and homogenized using a T-10 basic Ultraturrax (IKA, Staufen, Germany) for 2 min (total batch volume 12 g). The melt was transferred into the sample container of the B-290 Mini spray dryer with additional spray chilling setup (Büchi, Flawil, Switzerland), which was pre-conditioned for 1.5 hours prior to the production step (sample container 75-90 °C, spray tower 15 °C), monitored with a PeakTech infrared thermometer (PeakTech Prüf- und Messtechnik GmbH, Ahrensburg, Germany). A modified nozzle with an inner diameter of 2.5 mm was used to avoid nozzle blockages. The connecting passage between sample container and nozzle was heated with a resistance wire-based heating to 70 °C. The spray tower was fed with pre-cooled air produced via an additional dehumidifier Deltatherm[®] LT (Deltatherm[®] Hirmer GmbH, Much, Germany). The lipid melt was sprayed using nitrogen at a pressure of 6 bar. A high-performance cyclone was used to separate the microparticles.

4.3.4 **Determination of material distribution in the spray dryer**

The formulation D116 + 5 % Span 40 with an ASP load of 0.83 % was sprayed as described in 4.3.3. After the manufacturing process, the dryer was dismounted and the solidified material removed using a flexible soft rubber scraper. High-performance cyclone and filter were weighed before and after the experiment. Sampling points are depicted in *Figure 4-1*.



Figure 4-1: Schematic plan of the spray dryer. Particles were sampled at (1) sample container, (2) spray tower, (3) spray tower vessel, (4) connection, (5) high-performance cyclone, (6) product container and (7) filter

4.3.5 **Determination of particle size**

Approximately 20 mg of microparticles were suspended in 10 ml of an aqueous solution of 0.02 % PS 20 and analyzed using the laser diffraction system Horiba Partica LA-950 (Horiba, Kyoto, Japan).

4.3.6 **Determination of encapsulation efficiency**

Depending on theoretical drug load, 50 or 100 mg microparticles were dissolved in 1.0 ml of methylene chloride and the same amount of water was added. The tubes were incubated overnight on a horizontal shaker GFL 3031 (Gesellschaft für Labortechnik, Burgwedel, Germany) at a temperature of 39 °C at 60 rpm. Each sample was extracted in triplicate and measured using an Agilent Spectrophotometer 8453 (Agilent Technologies, Santa, Clara, CA, USA) at a wavelength of 258 nm [14].

4.3.7 Determination of lipid-melt viscosities

An MCR 100 rheometer (Physica, Anton Paar, Ostfildern, Germany) was used for determination of viscosity of the melt using the cone/plate method and the CP-50 1 measurement system at a logarithmic shear rate from $1-100 \text{ s}^{-1}$. The plate was preheated at the desired temperature of 75 or 90 °C.

4.4 Results and Discussion

4.4.1 Viscosity of the lipid melt

The viscosity of lipid melts can have an influence on the properties of the spray congealed solid lipid microparticles and the process [6] [9] [15]. In particular particle size of the generated spheres [4] [16], yield [9] and their time for solidification [17] are affected. Recently, Wong et al. investigated the temperature-dependent viscosity of lipids suitable for spray congealing, and revealed that viscosity of the melt decreases up to a certain temperature, above which the viscosity does not decrease any more with further heating [15]. Furthermore, the viscosity of triglyceride melts increases with longer side chains [18]. Consequently, the viscosities of differently formulated lipid melts at 75 and 90 °C were tested. D114 and D116-based formulations with addition of 5-10 % GMS and Span 40 showed viscosities between 10 and 25 mPas measured at 90 °C. With temperature reduction to 75 °C the viscosity slightly increased to 35 mPas (*Figure 4-2*), according to literature [15] [19]. When spraying was performed under a certain temperature, an increase in particle size upon spray congealing can be observed [15]. Formulation effects on the melt viscosity were not detectable during our experiments. Higher viscosities compared to pure lipid melts, as well as shear thinning are reported in literature for lipid suspensions [20].



Figure 4-2: Viscosity of drug-free lipid melts (D116 and D114) with emulsifiers at 90 °C and 75 °C at a shear rate of 1 s⁻¹ (Mean and SD, n = 2)

4.4.2 Yield distribution and material loss during the spray congealing process

Due to low yields of 20 to 50 % in the previously performed spray congealing experiments, it was necessary to evaluate where the high material loss occurred. Other researchers reported yields of over 90 % with the use of spray congealing [4] [9] [21]. With the standard conditions (inlet temperature 90 °C, spray pressure 6 bar, aspirator 100 %, spray flow 414 l/h) a yield of 26 % was exemplarily achieved during implementation of our spray congealing process (*Figure 4-3 A*). Highest material loss occurred in the spray tower itself (49 %), whereas only 1 % ended directly in the spray tower vessel. The material found in the tower was not completely solidified when getting in contact with the glassware. Another critical part for material loss is the sample container. Having a plane bottom and being construed for volumes of up to 300 ml [22], the container is prone for residual melt remaining inside when working with smaller batches. All other parts, like the connection, the high-performance cyclone, as well as the filter did not have a critical influence on material loss. A low material loss in the cyclone was also reported by Scott et al. [6].

The aspirator, generally responsible for drying of particles, is known to influence total batch yield in spray drying [23] [24]. The speed with which the droplets follow the airstream determines their contact time with the tempered air and thus the drying or solidification [25]. Reduction of aspirator power from 100 to 70 % led to a marginally reduced yield as observed with spray-dried insulin [26]. The material found in the spray tower was reduced to 36 %, whereas the particle amount found in the spray tower vessel increased. Obviously, the reduced aspirator force enabled a more effective solidification in the tower. Thus, the amount usually found in the spray tower was transferred to the spray tower vessel, also confirming the theory of a more effective solidification. Other parts of the spray dryer were only of minor importance (*Figure 4-3 B*).



Figure 4-3: Material distribution after spray congealing experiments using standard conditions (spray flow 401 l/h, aspirator 100 %, 90 °C) (A), reduced aspirator power of 70 % (B), reduced spray flow of 301 l/h (C), increased spray flow of 601 l/h (D), melt temperature 75 °C (E) and installation of an additional stirrer (F). "Others" includes filter, high-performance cyclone and connection tube between tower and high-performance cyclone

An important factor in both spray drying and congealing is the spray flow, responsible for the effectiveness in atomization [27]. Reduction of spray flow to 301 l/h led to a remarkable reduction in product yield (*Figure 4-3*). Material loss in the spray tower was increased to about 60 %, whereas more than 8 % of the powder were found in the spray tower vessel, indicating the formation of larger droplets and incomplete solidification. Also, a high amount of residual lipid material was found in the sample container. Application of higher spray flow of 601 l/h resulted in an increased product yield of 45 %. Compared to standard conditions, the material lost in the spray tower was reduced to approximately 6 % and about 16 % of the material could be collected from the spray tower vessel. Surprisingly, again a high residual melt could be found in the sample container. A higher spray flow causes the formation of smaller droplets, which solidify more rapidly [9] [28]. Reduction of spray flow led to larger spheres, which could be responsible for lower yields, due to insufficient hardening and collision with the tower wall. Furthermore, larger spheres generated at lower spray flow values may not be able to follow the air stream [29].

Another approach to influence spray properties is the modification of the melt viscosity by changing the temperature (*Figure 4-3 E*) [4] [15]. A higher product yield of 41 % was reached with a reduction of the melt temperature from 90 to 75 °C. The amount collected in the spray tower could be reduced to approx. 6 %, due to a more effective solidification, as the temperature difference between liquid and solid state of the lipid was reduced. This also caused a higher amount of solidified product in the spray tower vessel. As a downside, a higher amount of residual melt of about 27.35 % resulted.

For an effective and homogenous spray congealing process an additional stirring device was installed. The stirrer led to an increase in product yield to about 51 %. Moreover, the material lost in the spray tower could be reduced to approx. 9 %, whereas the material collected in the spray tower vessel was increased to about 31 %. The residual melt could be minimized by the stirrer to approx. 6 %, as less material gathered at the sample container.

The tested parameters showed that material loss in spray congealing was sensitive to production parameters. Substantial loss of material could be noticed in the spray tower, as well as in the sample container. The most positive effect resulted from an increase of spray flow to 601 l/h and the installation of an additional stirrer.

4.4.3 Investigations on particle size of drug-loaded particles

Particle size is another critical parameter in the development of a spray congealing process. The size of the generated spheres affects drug release and encapsulation capacity [30] [31]. For this reason, particle fractions from the spray tower, spray tower vessel and product container were collected and analyzed for mean particle size (*Figure 4-4*).



Figure 4-4: Mean particle size of samples collected in the product container (A), spray tower vessel (B) and spray tower (C). Standard settings were defined as spray flow 414 l/h, spray pressure 6 bar, aspirator 100 %, melt temperature 90 °C (black arrows)

Most important was the particle size of the product. The largest particles (> 65 μ m) were formed at a reduced spray flow of 301 l/h. When applying standard spray flow of 414 l/h the achieved particles were about 45 μ m in diameter. Increase of spray flow to 601 l/h led to reduction of the particle size to approx. 23 μ m. This effect was reported for both spray drying and spray congealing processes [9] [27] [32] [33]. Besides the product, also the particles found in the spray tower and the spray tower vessel were affected by the spray flow, with larger particles found at a reduced spray flow. The reduction of aspirator power affected the size of the particles collected in the product and spray tower vessel only marginally, in contrast to Esposito et al. during spray drying. A remarkable increase in size of the particles from the tower was observed. These observations might be explained by a longer residence time in the airstream, which allow the solidification of larger droplets [25]. Thus, a possible reduced deposition of droplets on the chamber wall can be achieved. Nevertheless, product size was barely affected. Generally, the product showed a smaller diameter compared to fractions collected in the spray tower and spray tower vessel. Due to their bigger size, particles were not able to follow the airstream and were deposited in the spray tower vessel.

A reduction of the melt temperature did not lead to an increase in particle size of the product, but the particle size of material collected from both the spray tower and the spray tower vessel was substantially reduced to under $100 \,\mu$ m.

The installation of an additional stirrer slightly reduced the product particle size to approx. $30 \,\mu\text{m}$. Particles collected in the spray tower showed a reduced particle size, whereas the particles in the spray tower vessel were larger. These findings indicate that the additional vibration of the stirrer caused a slightly inhomogeneous atomization.

4.4.4 Evaluation of drug distribution and encapsulation efficiency

Ideally, the final product should show an encapsulation efficiency of 100 %. Commonly, high encapsulation efficiencies are reported for spray congealed delivery systems, especially in comparison to emulsion-based methods, where problems in drug encapsulation occur due to migration into the aqueous phase [9] [11] [34] [35].



Figure 4-5: Drug loads of collected samples from the product container (A), spray tower (B) and spray tower vessel (C). Standard settings were defined as spray flow 414 l/h, spray pressure 6 bar, aspirator 100 %, melt temperature 90 °C (black arrows)

The application of standard conditions (*Figure 4-5 A*) led to an encapsulation efficiency of 38 % in the product and higher values of 45-60 % were obtained in the spray tower and spray tower vessel. Reducing the spray flow to 301 l/h resulted in a considerable increase in drug load to 80 %, which might go in accordance with the simultaneous increase in particle size, possibly enabling a better encapsulation [31] [36]. Spheres collected from the spray tower and vessel were not affected. Applying a spray flow of 601 l/h reduced the drug load in all sampled fractions. The reduction of aspirator capacity to 70 or 80 % respectively, affected drug encapsulation only marginally.

Melt temperature reduction to 75 instead of 90 °C increased the loading efficiency compared to standard conditions in the product as well as in the material collected in the spray tower vessel and the spray tower. The expectation of achieving better results with the installation of an additional stirrer were not fulfilled, all three fractions showed drug loads below 40 %.

In brief, with variation of apparatus settings it was possible to influence particle size and encapsulation efficiency to a certain extent. Generally, the spray congealing process can be regarded as complex process, with a variety of parameters having an influence. The reduction of spray flow was a suitable means to increase particle size, which also had positive effects on the encapsulation efficiency. On the other hand, an increase of spray flow affected the drug content negatively, due to the decreased particle size. Acceptable results concerning particle size and encapsulation efficiencies could be reached with the reduction of the melt temperature. Particle size remained constant with slightly enhanced encapsulation efficiencies in all sampled fractions. The installation of a vibrating stirrer led to opposite effects as expected, as an inhomogeneous atomization resulted and the product showed less encapsulated drug. The aspirator capacity did not affect the particle size and encapsulation efficiency substantially.

4.5 Conclusion

The spray congealing process was investigated concerning yield and material distribution, particle size and encapsulation efficiency. The parameters spray flow, aspirator power, melt temperature, as well as the influence of an additional stirring device were evaluated. Our investigations showed, that the highest material loss can be found in the spray tower, spray tower vessel as well as in the sample container. Filter, high performance cyclone and the connection between tower and cyclone were not evaluated as critical parts for material loss. The most pronounced effect on yield of the useable product could be determined by the increase of the spray flow to 601 l/h and by the melt temperature reduction to 75 $^{\circ}$ C. All other variations led to lower yields and a higher material loss. Evaluating particle size of the product, the variation of spray flow was determined as the most influencing factor. Drug distribution, which was an issue during previously performed experiments, could be improved by the reduction of melt temperature to 75 °C, as well as the reduction of spray flow. Viscosity measurements revealed, that the melt temperature was the mainly influencing factor, indicating that the decrease in melt temperature to 75 °C resulted in increased viscosity. Overall, the spray congealing process is influenced by a variety of parameters. Our results showed, that an increase in particle size is going hand in hand with a better encapsulation efficiency, which can be achieved by the reduction of spray flow. The increase in spray flow implemented the formation of smaller particles with considerably lower drug loadings. Increasing viscosity by the means of temperature reduction increased encapsulation efficiency due to faster solidification. Thus, a compromise needs to be found to assure an acceptable drug loading and process yields.

4.6 References

- F. E. Matos-Jr, M. Di Sabatino, N. Passerini, C. S. Favaro-Trindade, and B. Albertini, "Development and characterization of solid lipid microparticles loaded with ascorbic acid and produced by spray congealing," *Food Res. Int.*, vol. 67, pp. 52–59, 2015.
- [2] R. Al-Kassas, R. F. Donnelly, and P. A. McCarron, "Aminolevulinic acid-loaded Witepsol microparticles manufactured using a spray congealing procedure: implications for topical photodynamic therapy," J. *Pharm. Pharmacol.*, vol. 61, no. 9, pp. 1125–1135, 2009.
- [3] M. Emas and H. Nyqvist, "Methods of studying aging and stabilization of spray-congealed solid dispersions with carnauba wax. 1. Microcalorimetric investigation," *Int. J. Pharm.*, vol. 197, no. 1–2, pp. 117–127, 2000.
- [4] B. Albertini, N. Passerini, F. Pattarino, and L. Rodriguez, "New spray congealing atomizer for the microencapsulation of highly concentrated solid and liquid substances," *Eur. J. Pharm. Biopharm.*, vol. 69, no. 1, pp. 348–357, 2008.
- [5] L. Rodriguez, N. Passerini, C. Cavallari, M. Cini, P. Sancin, and A. Fini, "Description and preliminary evaluation of a new ultrasonic atomizer for spray-congealing processes," *Int. J. Pharm.*, vol. 183, no. 2, pp. 133–143, 1999.
- [6] M. W. Scott, M. J. Robinson, J. F. Pauls, and R. J. Lantz, "Spray congealing: Particle size relationships using a centrifugal wheel atomizer," *J. Pharm. Sci.*, vol. 53, no. 6, pp. 670–675, 1964.
- [7] T. Yajima, N. Umeki, and S. Itai, "Optimum spray congealing conditions for masking the bitter taste of clarithromycin in wax matrix," *Chem. Pharm. Bull.*, vol. 47, no. 2, pp. 220–225, 1999.
- [8] I. Ilic, R. Dreu, M. Burjak, M. Homar, J. Kerc, and S. Srcic, "Microparticle size control and glimepiride microencapsulation using spray congealing technology," *Int. J. Pharm.*, vol. 381, no. 2, pp. 176–183, 2009.
- [9] A. Maschke, C. Becker, D. Eyrich, J. Kiermaier, T. Blunk, and A. Gopferich, "Development of a spray congealing process for the preparation of insulin-loaded lipid microparticles and characterization thereof," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 2, pp. 175–187, 2007.
- [10] N. Passerini, B. Perissutti, B. Albertini, E. Franceschinis, D. Lenaz, D. Hasa, I. Locatelli, and D. Voinovich, "A new approach to enhance oral bioavailability of Silybum Marianum dry extract: association of mechanochemical activation and spray congealing," *Phytomedicine*, vol. 19, no. 2, pp. 160–168, 2012.
- [11] S. Scalia, D. Traini, P. M. Young, M. Di Sabatino, N. Passerini, and B. Albertini, "Comparison of spray congealing and melt emulsification methods for the incorporation of the water-soluble salbutamol sulphate in lipid microparticles," *Pharm. Dev. Technol.*, vol. 18, no. 1, pp. 266–273, 2013.
- [12] N. V. N. Jyothi, P. M. Prasanna, S. N. Sakarkar, K. S. Prabha, P. S. Ramaiah, and G. Y. Srawan, "Microencapsulation techniques, factors influencing encapsulation efficiency.," *J. Microencapsul.*, vol. 27, no. 3, pp. 187–197, 2010.
- [13] Y. Yeo and K. Park, "Control of encapsulation efficiency and initial burst in polymeric microparticle systems.," *Arch. Pharm. Res.*, vol. 27, no. 1, pp. 1–12, 2004.
- [14] B. S. Dattatreya, D. V Usha, N. S. Susheelamma, and K. K. Bhat, "Aspartame: studies on UV spectral characteristics," *Int. J. Food Sci. Technol.*, vol. 38, no. 7, pp. 767–775, 2003.
- [15] P. C. H. Wong, P. W. S. Heng, and L. W. Chan, "Viscosity-temperature relationship of lipid-based excipients amenable for spray congealing: Derivation of a rheological parameter with good correlation to particle size," *Eur. J. Lipid Sci. Technol.*, vol. 118, pp. 1062–1073, 2016.

- [16] J. B. Lo, L. E. Appel, S. M. Herbig, S. B. McCray, and A. G. Thombre, "Formulation design and pharmaceutical development of a novel controlled release form of azithromycin for single-dose therapy," *Drug Dev. Ind. Pharm.*, vol. 35, no. 12, pp. 1522–1529, 2009.
- [17] M. Savolainen, J. Herder, C. Khoo, K. Lovqvist, C. Dahlqvist, H. Glad, and A. M. Juppo, "Evaluation of polar lipid-hydrophilic polymer microparticles," *Int. J. Pharm.*, vol. 262, no. 1–2, pp. 47–62, 2003.
- [18] D. P. Geller and J. W. Goodrum, "Rheology of vegetable oil analogs and triglycerides," J. Am. Oil Chem. Soc., vol. 77, no. 2, pp. 111–114, 2000.
- [19] D. Valeri and A. J. A. Meirelles, "Viscosities of fatty acids, triglycerides, and their binary mixtures," J. Am. Oil Chem. Soc., vol. 74, no. 10, pp. 1221–1226, 1997.
- [20] R. M. Martins, S. Siqueira, M. O. Machado, and L. A. Freitas, "The effect of homogenization method on the properties of carbamazepine microparticles prepared by spray congealing," *J. Microencapsul.*, vol. 30, no. 7, pp. 692–700, 2013.
- [21] B. Albertini, N. Passerini, M. Di Sabatino, B. Vitali, P. Brigidi, and L. Rodriguez, "Polymer-lipid based mucoadhesive microspheres prepared by spray-congealing for the vaginal delivery of econazole nitrate," *Eur. J. Pharm. Sci.*, vol. 36, no. 4–5, pp. 591–601, 2009.
- [22] Büchi Labortechnik AG, "Spray Chilling," 2016. [Online]. Available: http://www.buchi.com/en/content/spray-chilling. [Accessed: 13-Nov-2016].
- [23] B. B. Patel, J. K. Patel, S. Chakraborty, and D. Shukla, "Revealing facts behind spray dried solid dispersion technology used for solubility enhancement," *Saudi Pharm. J.*, vol. 23, no. 4, pp. 352–365, 2015.
- [24] A. Chawla, K. M. G. Taylor, J. M. Newton, and M. C. R. Johnson, "Production of spray dried salbutamol sulphate for use in dry powder aerosol formulation," *Int. J. Pharm.*, vol. 108, no. 3, pp. 233–240, 1994.
- [25] E. Esposito, R. Roncarati, R. Cortesi, F. Cervellati, and C. Nastruzzi, "Production of Eudragit microparticles by spray-drying technique: influence of experimental parameters on morphological and dimensional characteristics.," *Pharm. Dev. Technol.*, vol. 5, no. 2, pp. 267–78, 2000.
- [26] K. Ståhl, M. Claesson, P. Lilliehorn, H. Lindén, and K. Bäckström, "The effect of process variables on the degradation and physical properties of spray dried insulin intended for inhalation," *Int. J. Pharm.*, vol. 233, no. 1–2, pp. 227–237, 2002.
- [27] M. J. Maltesen, S. Bjerregaard, L. Hovgaard, S. Havelund, and M. van de Weert, "Quality by design -Spray drying of insulin intended for inhalation," *Eur. J. Pharm. Biopharm.*, vol. 70, no. 3, pp. 828–838, 2008.
- [28] P. He, S. S. Davis, and L. Illum, "Chitosan microspheres prepared by spray drying.," *Int. J. Pharm.*, vol. 187, no. 1, pp. 53–65, 1999.
- [29] K. Cal and K. Sollohub, "Spray drying technique. I: Hardware and process parameters," J. Pharm. Sci., vol. 99, no. 2, pp. 575–586, 2009.
- [30] C. Cavallari, B. Luppi, A. M. Di Pietra, L. Rodriguez, and A. Fini, "Enhanced release of indomethacin from Pvp/stearic acid microcapsules prepared coupling Co-freeze-drying and ultrasound assisted spraycongealing process," *Pharm. Res.*, vol. 24, no. 3, pp. 521–529, 2007.
- [31] S. Jaspart, G. Piel, L. Delattre, and B. Evrard, "Solid lipid microparticles: formulation, preparation, characterisation, drug release and applications," *Expert Opin. Drug Deliv.*, vol. 2, no. 1, pp. 75–87, 2005.
- [32] N. Passerini, S. Qi, B. Albertini, M. Grassi, L. Rodriguez, and D. Q. Craig, "Solid lipid microparticles produced by spray congealing: influence of the atomizer on microparticle characteristics and mathematical

modeling of the drug release," J. Pharm. Sci., vol. 99, no. 2, pp. 916-931, 2010.

- [33] K. B. Prinn, H. R. Costantino, and M. Tracy, "Statistical modeling of protein spray drying at the lab scale.," AAPS PharmSciTech, vol. 3, no. 1, p. E4, 2002.
- [34] B. Albertini, M. Mezzena, N. Passerini, L. Rodriguez, and S. Scalia, "Evaluation of spray congealing as technique for the preparation of highly loaded solid lipid microparticles containing the sunscreen agent, avobenzone," J. Pharm. Sci., vol. 98, no. 8, pp. 2759–2769, 2009.
- [35] M. Mezzena, S. Scalia, P. M. Young, and D. Traini, "Solid lipid budesonide microparticles for controlled release inhalation therapy," AAPS J., vol. 11, no. 4, pp. 771–778, 2009.
- [36] N. Passerini, B. Perissutti, M. Moneghini, D. Voinovich, B. Albertini, C. Cavallari, and L. Rodriguez, "Characterization of carbamazepine-Gelucire 50/13 microparticles prepared by a spray-congealing process using ultrasounds," *J. Pharm. Sci.*, vol. 91, no. 3, pp. 699–707, 2002.

5 INCORPORATION OF GONADORELIN [6-D-PHE] INTO SPRAY CONGEALED SOLID LIPID MICROPARTICLES AND EVALUATION OF RELEASE MECHANISMS

This chapter is intended for publication

5.1 Abstract

The decapeptide gonadorelin [6-D-Phe] was incorporated into spray congealed microparticles consisting of triglycerides of different chain lengths. As a continuous release in the desired time frame of 15 d could not be achieved with the exclusive use of triglycerides, different additives like GMS, CSA and Spans were applied to modify the release behavior. To investigate the underlying mechanisms influencing the release profile, polymorphic behavior, surface morphology and wettability, as well as water uptake were investigated. Crystallization of the α - or β -form and entrance of water were determined to be influenced by the emulsifier type and its concentration. Wettability of lipid surfaces was low regarding pure triglycerides and was increased when surfactants, especially monoglycerides, were added to the formulation. SEM images showed a particle breakdown upon incubation when liquid emulsifiers were used. It was possible to achieve a sustained release for 14 and 22 d with the addition of 10 % GMS to D116 and D118. Release duration of 20 d was obtained with the addition of 5 % Span 40 to D114. Thus, two formulations for subsequent in vivo studies in swine could be identified. The challenge of a high burst release could be minimized by suspending the particles in a thermosensitive poloxamer-based gel.

5.2 Introduction

The treatment with peptides or proteins often comes with frequent injection [1]–[4]. This can be a problem and limitation for application in human and as well veterinary medicine, due to compliance, efforts and cost issues compared to orally applied small molecule drugs. To circumvent these problems, sustained release systems for peptides and proteins are sought after [5]–[12].

One example for an immediate releasing biopharmaceutical formulation in veterinary use is Virbagen[®] Omega (Virbac), delivering interferon omega for the feline immunodeficiency virus (FIV) and feline leukemia virus (FeLV) in cats, as well as parvovirus in dogs [13]–[16]. Furthermore, GnRH-analogues are currently used for induction of ovulation in swine, cattle and mares [17]–[20]. Gonadorelin [6-D-Phe] (G [6-D-Phe]), active compound in Gonavet[®]Veyx (Veyx Pharma GmbH) with a molecular weight of 1000 Da, is a GnRH (LHRH)-I agonist with a higher affinity to GnRH receptors [21]–[24]. Other commonly used GnRH analogues for estrus control are buserelin (Receptal[®], MSD Animal Health) [25] [26], lecirelin (Dalmarelin[®], Fatro) [27] and peforelin (Maprelin[®], Veyx Pharma GmbH) [28]. Besides the veterinary importance, GnRH analogues are used in several indications in humans, for example breast and prostate cancer, endometriosis and fertility disorders [29] [30].

In 1985, one approach for a sustained delivery was the application of nafarelin, a substance related to GnRH-I for estrus suppression in female dogs, via an osmotic pump system [12]. Current, sustained release delivery systems used in veterinary medicine are injectable suspensions, solid implants, in-situ forming implants and microparticles mostly based on lipid compounds [12] [31] [32] [33]. Lipid excipients inherit several advantages compared to other polymers, as biodegradability, high biocompatibility and low costs [34] [35].

The lipid-based implant Suprelorin[®] (Virbac), containing deslorelin acetate, is approved for the non-invasive castration in male dogs [36] [37]. The implant consists mainly of hydrogenated palm oil and is biodegradable, which omits surgical removal [37]. Besides the use in male dogs, a deslorelin-containing implant was also tested in female dogs [36] and was able to achieve a suppression of estrus for 27 months [38].

Estrogen and progesterone-loaded, polymer-based microparticles have been investigated in terms of estrus synchronization in mares [39]. Additionally, ProHeart[®] 6 (Zoetis) is a microparticular delivery system based on tristearin for the antibiotic substance moxidectin,

which has the FDA approval in the USA and provides a 6-month protection against the canine heartworm disease [32] [40] [41].

Consequently, we developed an injectable sustained release system based on triglyceride microparticles for G [6-D-Phe] via spray congealing. The incorporation of the drug G [6-D-Phe] and the adaption of the process was the focus of this work. Additionally, the in vitro release of G [6-D-Phe] was to be investigated to achieve a sustained release over two weeks suitable for estrus synchronization in swine [42]. In order to understand the underlying release mechanism, lipid polymorphism, surface wettability and water uptake were evaluated.

5.3 Materials and Methods

5.3.1 Materials

Lyophilized G [6-D-Phe] was donated by Veyx Pharma GmbH (Schwarzenborn, Germany). Triglycerides Dynasan 114 (trimyristin, D114), Dynasan 116 (tripalmitin, D116) and Dynasan 118 (tristearin, D118) were kindly provided from Cremer Oleo (Witten, Germany). Glycerol monostearate (GMS) with a monoester content of 40-55 % was purchased from Caelo (Caesar & Loretz, Hilden, Germany). Cetylstearyl alcohol (CSA, Lanette[®] O) was purchased from BASF (Ludwigshafen, Germany). Span 40, 60, 80, 85 were kindly provided from Croda (Nettetal, Germany). Poloxamer 407 and Hydranal[®] for Karl-Fischer titration were purchased from Sigma-Aldrich (Taufkirchen, Germany). Ultrapure deionized water with a conductivity of less than 0.055 µS/cm (Milli-Q Water systems, Millipore, MA, USA) was used throughout all experiments. Centrifuge tubes were purchased from VWR International GmbH (Darmstadt, Germany). Visking dialysis tubings with a diameter of 1.6 mm and a cut-off of 12-14 kDa for performance of release studies were of analytical grade.

5.3.2 Cryogenic milling of G [6-D-Phe]

G [6-D-Phe] was milled using a Retsch[®] Cryomill (Retsch Technology, Haan, Germany) equipped with two stainless steel milling balls (diameter 10 mm) with a precooling time of 10 min at 5 Hz followed by the actual milling step of 4 min at 20 Hz. The obtained powder was aliquoted under nitrogen atmosphere.

5.3.3 Preparation of G [6-D-Phe]-loaded microparticles by spray congealing

Lipid compounds were melted on a hot plate at 90 °C, the peptide was added and homogenized using a T-10 basic Ultraturrax (IKA Laboratory Technology, Staufen, Germany) for 2 min. A Büchi B-290 Mini spray dryer (Büchi, Flawil, Switzerland) with additional spray chilling setup, was pre-conditioned for 1.5 hours prior to the production step (sample container 90 °C/110 °C, spray tower 13-15 °C). Additional equipment consisted of a sample container surrounded by a polyethylene glycol (PEG) 400 bath. The connection between sample container and nozzle was heated externally by a resistance wire-based heating to approximately 70 °C. The spray tower was fed with pre-cooled air produced by an additional dehumidifier Deltatherm[®] LT (Deltatherm[®] Hirmer GmbH, Much, Germany). The lipid melt was filled into the sample container and sprayed using nitrogen and a modified nozzle with a channel diameter of 2.5 mm using the following apparatus settings:

Spray parameter	Value
Spray pressure [bar]	6
Aspirator power [%]	100
Valve opening [turns]	1.25-1.5
Temperature oil bath [°C]	90 (D114, D116), 110 (D118)
Inlet temperature [°C]	13-17
Pressure filter [mbar]	92
Spray flow [l/h]	414

All produced microparticle batches had a theoretical drug loading of 1.8 %.

5.3.4 **Determination of particle size**

20 mg of microparticles were suspended in 10 ml of an aqueous solution of 0.02 % PS 20 and measured with the laser diffraction system Horiba Partica LA-950 (Horiba, Kyoto, Japan).

5.3.5 Investigations on polymorphic behavior using differential scanning calorimetry (DSC)

Microparticles, triglycerides and lipid mixtures were analyzed using a Mettler DSC 821e (Mettler Toledo, Columbus, OH, USA). 5-15 mg sample were weighed into aluminum crucibles and analyzed at a heating and cooling rate of 10 K/min between 0 °C to 110 °C in two cycles. Incubated particles were dried on a lint-free cloth for 1 h at room temperature.

5.3.6 X-Ray powder diffraction (XRPD) of lipid microparticles

XRPD measurements were conducted with 2 different systems:

- Small sample quantities were analyzed with a PANalytical Empyrean (PANalytical, Almelo, the Netherlands) equipped with a copper anode (45 kV, 40 mA, K α 1 emission $\lambda = 0.154$ nm) and a PIXcel3D detector between 5 to 50 ° 2-Theta with a step of 0.039 °.
- To analyze multiple samples and to benefit from an autosampler, samples were scanned between 10 and 40 ° 2-Theta at a step size of 0.05 ° with a XRD 3000 TT (Seifert, Ahrensburg, Germany) equipped with a copper anode (λ= 0.15418 nm, 40 kV, 30 mA).

5.3.7 Karl-Fischer titration

10 mg microparticles were weighed into 2 R glass vials, closed with rubber stoppers and analyzed with an Aqua 40.00 titrator (Analytik Jena, Jena, Germany) with a headspace module at 100 °C. The analysis of incubated particles in PBS was performed after drying on a lint-free cloth for 1 h at room temperature.

5.3.8 Characterization of particle morphology using scanning electron microscopy (SEM)

A Jeol JSM 6500-F SEM (Jeol, Tokyo, Japan) was used for SEM-images at a voltage of 2 kV. Microparticles were fixed on aluminum sample holders with black, double adhesive tape (Plano, Wetzlar, Germany) and analyzed without an additional graphite coating. After release studies, particles were washed five times and dried for 48 h in a vacuum dryer (Memmert GmbH, Schwabach, Germany) at 25 mbar and 25 °C.

5.3.9 Contact-angle measurements for prediction of lipid surface wettability

Contact angle measurements were performed using the Krüss Drop Shape Analyzer DSA 25 (Krüss, Hamburg, Germany). 0.5 g of lipid components were melted at 80 °C, poured onto glass microscopy slides and cooled for solidification. A drop volume of 2 μ l water was placed on the lipid surface at a speed of 1 ml/s. Measurements were performed in triplicate after a wait time of 20 s (mean values of left and right contact angle). The ellipse-tangent-1 fit was used for calculation of contact angles.

5.3.10 Determination of drug-content using RP-HPLC after extraction

20 mg microparticles were weighed into 30 ml-centrifuge tubes, dissolved in 2.0 ml of methylene chloride and the same amount of water was added. The tubes were placed on a horizontal shaking incubator GFL 3031 (Gesellschaft für Labortechnik, Burgwedel, Germany) at 39 °C for 12 h. Samples of 1 ml were taken from the water phase and analyzed using RP-HPLC. Extractions were performed in triplicate. RP-HPLC analysis was performed using an Agilent RP-HPLC system (Agilent, Santa Clara, CA, USA), supplied with a 250 x 4.6 mm Luna (5 μ m) C-8 column (Phenomenex, Aschaffenburg, Germany) and a SecurityGuard KJO-4282 C-8 pre-column (5 μ m) (Phenomenex, Aschaffenburg, Germany). Columns were maintained at 40 °C using the following gradient:

Time [min]	A [%]	B [%]	
0	65	35	
1	65	35	
21	65	35	
22	10	90	
26	10	90	
35	65	35	

Mobile Phase A: 1000 ml highly purified water + 1 ml trifluoracetic acid

Mobile Phase B: 800 g Acetonitrile + 200 g water + 1.2 ml trifluoracetic acid

5.3.11 In vitro release behavior of G [6-D-Phe] from lipid microparticles

Approximately 50 mg of microparticles were transferred into dialysis tubings, 2.0 ml of phosphate-buffered saline (PBS) pH 7.4 preserved with 0.05 % NaN₃ were added, followed by the addition of glass balls which served as additional weight. The closed tubings were placed in 50 ml centrifuge tubes, 15 ml PBS were added and incubated on a horizontal shaking incubator at 39 °C and 60 rpm. When poloxamer gels were used as suspension medium, the centrifuge tubes were pre-filled with PBS and incubated overnight at 39 °C. Dialysis tubings with an aliquot of microparticles and 2 ml of poloxamer gels were closed and put into pre-heated surrounding buffer. At predetermined time points, samples of 1 ml volume were taken and removed solution was replaced by fresh PBS. Samples were analyzed with RP-HPLC as described in *5.3.10*.

5.3.12 Preparation of poloxamer gels

Poloxamer 407 (Sigma-Aldrich, Taufkirchen, Germany) was dissolved in water in a refrigerator at 2-8 °C overnight.

5.3.13 Temperature-dependent viscosity measurements using a rotation viscometer

For analysis of temperature-dependent gelling behavior, an MCR 100 rheometer (Physica Anton Paar, Ostfildern, Germany) with a plate/plate system PP-25 and a temperature ramp from 4 to 34 °C was used at a shear stress of 1 Pa. Each measurement was performed in triplicate.

5.4 Results and Discussion

5.4.1 Preparation of drug-loaded lipid microparticles by spray congealing

Based on previous spray congealing tests with placebo and model substance-loaded microparticles, a first set of spray congealing conditions and formulations for G [6-D-Phe] was selected (*Table 5-1*). Initial criteria for the performed pre-tests were yield, particle size and encapsulation efficiency (EE).

TG	Additive	Concentration [%]	Yield [%]	EE [%]
D 118	-	-	13.8	61.3
D 116	-	-	20.6	56.7
D 118	GMS	10	17.8	61.4
D 118	GMS	20	16.6	40.4
D 118	GMS	40	3.1	15.9
D 118	GMS	60	4.9	23.5
D116	GMS	10	26.1	38.9
D116	GMS	20	30.6	37.0
D116	GMS	40	21.0	41.0
GMS	-	-	≤ 1	23.5
D116	CSA	10	32.3	47.1
D116	CSA	20	19.1	56.8
D116	CSA	60	26.6	54.8

Table 5-1: First generation G [6-D-Phe]-loaded formulations produced by spray congealing

G [6-D-Phe]-loaded microparticles could be sprayed in analogy to placebo and model substance-loaded particles. Compared to the incorporation of model substances, the peptide led to a more persistent and continuous spray cone and less valve blockages. The use of pure GMS resulted in needle-like, inadequate particles, which showed pronounced electrical charging. All other batches showed higher yields. Pure triglyceride formulations showed yields below 21 % and GMS addition led to slightly increased product yields. More than 30 % product could be obtained with 10-20 % GMS. Generally, obtained yields were lower as reported in literature [43] [44] [45], since we only considered the powder in the product vessel.

EE was about 60 % for pure triglycerides. The addition of 10 % GMS to D118 led to comparable results whereas EE was slightly decreased when added to D116. With higher amounts of monoglycerides added to D118 the drug load decreased. The addition of higher amounts of GMS and CSA to D116-based microparticles did not affect the drug load, which

went in accordance with data reported by Eustáquio Matos-Jr. et al. comparing hydrogenated palm oil and glycerol monostearate particles [45].

TG	Additive	Concentration [%]	Yield [%]	EE [%]
D114	GMS	10 %	33.9	78.4
D114	GMS	5 %	22.7	78.7
D114	Span40	5 %	28.8	51.2
D114	Span 40	10 %	12.1	21.1
D114	Span 40	30 %	1.6	n.d.
D114	Span 85	10 %	17.8	60.9
D114	Span 85	20 %	16.2	46.5
D114	Span 80	10 %	2.1	57.4
D114	Span 80	20 %	16.4	21.9
D114	Span 85	30 %	3.4	21.1
D116	Span 85	10 %	15.9	71.9
D116	Span 85	20 %	17.8	72.4
D116	Span 85	30 %	7.6	71.4
D116	D112	10 %	7.9	81.8
D116	Span 40	5 %	15.6	29.9
D116	Span 40/D112	10 %	4.1	14.4
D116	GMS/ D112	5 %/10 %	22.88	64.34
D116	GMS/ D112	10 %/10 %	18.22	65.30
D116	Span 80	10 %	7.27	40.15
D116	Span 80	30 %	10.59	43.56

Table 5-2: Second generation G [6-D-Phe]-loaded formulations produced by spray congealing

Second generation of lipid microparticles was produced using short-chain triglycerides and a broader spectrum of surfactants to achieve a faster release [46]. D114 rendered adequate yields between 23 and 34 % upon spray congealing in combination with 5 and 10 % GMS, respectively. The encapsulation efficiency was raised to 79 %. The addition of Span 40 with a HLB value of 6.7 [47] [48] caused an irregular spray cone due to the formation of foam, also reasoning lower yields and decreased EE. The use of surfactants with lower HLB-values, Span 80 and 85 [47], also reduced process yield. EE was acceptable (71 %) when Span 85 was used in combination with D116 and reduced when combined with D114 (21-60 %). Span 80 led to lower EE. D112 was previously used as low melting component in the manufacturing of solid lipid implants, where a faster degradation rate was obtained [49] [50]. A lower yield but

acceptable EE of 81.75 % was achieved. Triple combination of D112 with GMS and Span 40 resulted in appropriate results regarding GMS, but both reduced yield and drug content when Span 40 was added.

Sprayability, yield and EE are critical parameters in the production of spray congealed lipid microparticles. The parameters yield and EE seemed to be influenced by the concentration and state of emulsifier, also affecting solubility of the drug in the lipid [51]. In general, for spray congealed lipid microparticles high yields and encapsulation efficiencies above 90 % are reported in literature, which could not be achieved during our experiments [44] [52]. No pronounced effects regarding EE were found by Mccarron et al., comparing different emulsifier-triglyceride compositions [53]. Possible approaches to enhance EE were achieved by addition of Aerosil[®] and a followed increased viscosity [54], bigger particle sizes [55] and higher drug-lipid ratio [56].

5.4.2 Characterization of polymorphic behavior using DSC and XPRD

The polymorphic behavior of the microparticles was analyzed by DSC and XRPD. The heating curve of D116 showed a melting endotherm with an onset of 42 °C corresponding to the α -polymorph [57] followed by an exothermic recrystallization and subsequent melting of the thermodynamically stable β -polymorph (*Figure 5-1*) [58]. Particles with 5 % Span 40 showed similar trace. A single sharp peak corresponding to the β -form could be observed when Span 85, GMS or the lower melting D112 were added. The small endotherm at 45 °C for microparticles of 80 % D116, 10 % D112 and 10 % GMS was either due to melting of D112 or the α -polymorph of D116 [57] [59]. For pure GMS, polymorphic transformation is also known with the α -polymorph melting at 67.9 °C and β -polymorph at 71.9 °C [60]. An event in this temperature region could not be detected in the GMS-containing microparticles.



Figure 5-1: DSC thermograms of G [6-D-Phe]-loaded microparticles (n = 2). Thermograms represent the first heating scan of one sample exemplarily and were shifted on the ordinate for better visualization

The addition of 5 and 10 % Span 40, 80 and 85 to D114 resulted in crystallization of the thermodynamically stable β -polymorph with a melting onset of 53 °C and a melting peak of 55-58 °C [61] [62]. The instable α -form with a melting point of 32 °C [62] was not detected in any sample. Whittam and Rosano observed that the transformation to the β -modification was already completed after 31 min at room temperature [62].

To further investigate the crystallization behavior of lipids, DSC measurements were conducted after incubation in PBS. D114-based samples showed the β -polymorph with a melting temperature of 57 °C [58] [63] after incubation, irrespective of the additive. D116-based microspheres with addition of GMS/D112 showed a small endotherm at 46 °C [64] prior to incubation. Upon incubation, re-crystallization to the β -polymorph melting at 60 °C occurred, which is a slightly lower melting point compared to literature [62]. Furthermore, a wider peak was observed. When Span 85 was added, no changes were observed.



Figure 5-2: Representative DSC thermograms of G [6-D-Phe]-loaded triglyceride microparticles before and after incubation (4 h, 24 h, n = 3) in PBS buffer pH 7.4 at 39 °C. Scans were shifted on the ordinate for better visualization

XRPD measurements were conducted to further investigate the polymorphs (*Figure 5-3*). D114 raw material showed the characteristic triglyceride pattern of the β -polymorph with three peaks at 18, 22 and 23 ° 2-Theta [65]. This pattern was also visible upon analysis of spray congealed particles of D114 with 10 % Span 40 and D114 + 5 % GMS. GMS raw material revealed two peaks at 18 and 24 ° 2-Theta also indicating the β -polymorph [66]. D116 raw material showed the peaks of the stable β -modification [65]. Microparticles with 5 % Span 40 addition as well as pure Span 40 showed one single peak at 21 ° 2-Theta, confirming a different crystal structure [57]. In contrast, D116 microparticles with 10 % Span 85 clearly exhibited the β -polymorph. As already indicated in DSC measurements, crystallization of the stable modification was not completed in the microparticle formulation D116 + 10 % GMS/10 % D112.



Figure 5-3. Representative XRPD measurements of lipid raw materials and spray congealed lipid microparticles (n = 3). Diffraction patterns were shifted on the ordinate for better visualization

Crystallization behavior of lipid-based delivery systems is influenced by a variety of exogenic factors, especially when manufactured at higher temperatures. Underlying principles are not yet fully understood. In our study, the crystallizing polymorph depended on the used triglyceride itself, as well as the emulsifier. The delay or promotion of crystallization in the thermodynamically most stable β -form depends on the ability of the emulsifier to interact with the crystal lattice of the lipid [67]. D114 with its shorter fatty acid side chains seemed to have an enhanced property to crystallize in the β -polymorph upon spray congealing, irrespective of the emulsifier. D116 showed a partial complete formation of the α -polymorph in combination with 5 % Span 40, as well as in combination with GMS and D112 which indicates a stabilization of α -crystal lattice with these emulsifiers.

5.4.3 Dependency of water uptake of G [6-D-Phe]-loaded microparticles from the lipid composition

To evaluate moisture of freshly prepared particles and their water uptake, the produced formulations were investigated by the means of Karl-Fischer titration. Water uptake between 3 and 30 % was reported for triglyceride cylinders depending on cholesterol content over six months [68]. For triglyceride implants, a water uptake of maximum 10 % depending on the chain length could be demonstrated [69]. The higher water uptake of microspheres compared to cylinders can be explained by a higher surface to volume ratio [68]. Investigations on Gelucire[®] 50/13 tablets revealed that a continuous water uptake over 8 h to more than 80 % was possible [70]. In our case, particles showed an initial water content between 0.05 % (D114 + 10 % GMS) and 0.19 % (D114 + 20 % Span 80, *Figure* 5-4) after production.



Figure 5-4: Water content of spray congealed microparticles consisting of D114 with different emulsifying additives (mean and SD, n = 3)

After 24 h of incubation, water content of D116-based microparticles with Span 40 and 85 stayed below 3 % (*Figure 5-5*). A higher water uptake between 10 and 30 % was found when D114 was combined with Span 80 and 85. A water content below 10 % was obtained with 5 % Span 40, whereas the combination of 10 % Span 40 with D112 increased the incorporated water to more than 40 %. Vogelhuber et al. identified the chain length of triglycerides as possible tool for swelling and release modification [68]. Addition of GMS to D114 led to an increased water uptake, whereas 5 % GMS addition did not enhance water penetration. In general, D114-based microparticles showed higher water uptake especially in combination with liquid and low melting emulsifiers compared to D116 microparticles.



Figure 5-5: Water uptake of spray congealed lipid microparticles after incubation in PBS at 39 °C determined by Karl-Fischer titration (mean and SD, n = 3)

5.4.4 Impact of lipid-composition on contact angle and surface wettability

To understand release behavior from lipid matrices, surface wettability can be a helpful tool [71] [72]. In our case, the preparation of flat lipid surfaces was possible with all lipid mixtures. *Figure 5-6* shows the contact angle of highly purified water on formulations based only on mono- or triglycerides. Resulting contact angles are complex results of different influencing factors, like surface roughness and surface tension [72]–[74]. The highest contact angle of 114 $^{\circ}$ was observed for the triglyceride with the shortest fatty acid chain, D114. Tripalmitin (D116), tristearin (D118) and triarachidin (D120) showed lower contact angles and following better wettability. Glycerol monostearate showed the best wettability with a contact angle below 90 $^{\circ}$. Koennings et al. found out, that wettability of lipid matrices decreases with increasing chain lengths [72]. This trend could not be observed in our experiments. One reason might be, that the lipid was not casted out of an organic solution on the microscopy slide to achieve similar conditions and contraction behavior compared to the spray congealing process. This procedure seemed suitable for our purpose, as the surface structure and the crystallization behavior affect wettability [75]. This was also suggested by Del Curto et al., who reported improved wettability of co-melted particles compared to those prepared by solvent-stripping [4].



Figure 5-6: Contact angle of highly purified water on triglycerides (D114, D116, D118, D120) and GMS (mean and SD, n = 3)

Figure 5-7 shows the contact angles of D114-emulsifier mixtures. GMS addition led to improved wettability at 10%, whereas the contact angle was increased at only 5%. Concentrations above 20% Span 40 and 10-20% Span 80 reduced the contact angle dramatically. No pronounced effect was visible for a Span 85 addition.



Figure 5-7: Contact angle of highly purified water on D114-emulsifier mixtures (mean and SD, n = 3) Replacing D114 by D116, no pronounced effect on contact angles was observed in combination with 5 % Span 40 and 10 % D112 (*Figure 5-8*). A stepwise reduction was found when using Span 85. Formulation with the low melting triglyceride D112, the contact angle was raised to 123.61 °. A remarkable reduction in contact angle was observed using GMS and Span 60 as well as Span 85 in higher concentrations.


Figure 5-8: Contact angle of highly purified water on D116-emulsifier mixtures (mean and SD, n = 3) Admixing emulsifiers to triglycerides led to enhanced wettability depending on type of triglyceride and emulsifier type and its concentration. Surface roughness and the so called "blooming" of triglycerides needs to be taken into account, when investigating the wettability of lipid surfaces. The smooth surfaces reported for the α -modification [76] are known to obtain increased wettability compared to those of the platelet- or flake-like surface of the β -polymorph [77] [78]. Consequently, polymorphic transformations need to be taken into consideration, as we showed that crystallization behavior also depended on emulsifier-triglyceride composition.

5.4.5 In vitro release behavior of G [6-D-Phe] from lipid microparticles

Microparticles based on pure D116 and D118 were investigated concerning their G [6-D-Phe] release behavior. In both cases a low burst between 10 and 25 % resulted, followed by only minimal release within the subsequent two weeks (*Figure 5-9*), according to literature [2]. This goes in line with the fact that the pure triglyceride microparticles do not take up substantial amounts of water and G [6-D-Phe] only dissolves at the surface. Comparable release profiles from lipid microparticles were observed for desmopressin [79] [80]. Additionally, the used theoretical drug load of 1.8 % was insufficient to form a connected pore network. In literature, drug loads above 10 % are reported for implants to be sufficient for a formation of connected pores and to achieve sustained protein release [81].



Figure 5-9: G [6-D-Phe] release from triglyceride microparticles (D116 solid line, D118 dashed line; mean and SD, n = 3)

To enhance peptide release, combinations of triglycerides with non-ionic emulsifiers were tested. Glycerol mono- and distearate have been reported to enhance peptide release from lipid microspheres [46] [79]. *Figure 5-10* gives an overview on the influence of a CSA- and GMS-addition to triglycerides on G [6-D-Phe] release. The addition of CSA to D116 led to an increased burst compared to pure D116 already at 10 %. Increasing the CSA content above 20 %, about 85 % of the drug were already released as burst. Microparticles prepared from 10 % GMS and 90 % triglyceride exhibited a high burst release of 55 %, followed by a continuing release until day 14. In total, 70 % of the incorporated drug was released. With 20 % GMS, the burst release was comparable and the sustained release could be monitored until day 12. An immediate release was obtained when 40 % GMS were added.

SEM micrographs of freshly prepared D118 microparticles with 10 % GMS showed particular shape before and after incubation, with increased roughness after incubation, indicating possible matrix changes like "blooming" (*Figure 5-11*) [78].



Figure 5-10: G [6-D-Phe] release from triglyceride microparticles with varied amounts of CSA (10, 20, 40 %) (A). G [6-D-Phe] release from D116 (B) and D118 (C) microparticles with varied concentrations of GMS (10 %, 20 %, 40 %, 60 %), mean and SD, n = 3

The use of D118 with 10 and 20 % GMS was most promising, resulting in a burst release between 45 and 60 % and a continuous release over 14 to 22 d releasing more than 80 % of the incorporated drug. Further increase of GMS did not lead to a sustained release profile, possibly due to a monoglyceride-induced matrix instability (*Figure 5-10 C*) [79]. Incomplete drug release (only 80-90 %) is frequently observed for lipid-based drug delivery systems [82], possibly due to an incomplete water accessibility of the core [83] [84] or an adsorption to matrix material [80] [85].

D118 + 10 % GMS

Figure 5-11: SEM images of freshly prepared microparticles of D118 + 10 % GMS (left) and after a 4week incubation in PBS at 39 °C (right)

To generate particles with a shorter release than 22 d additional mixtures of triglycerides with emulsifiers were tested. Therefore, a variety of emulsifying substances with different HLB-values were introduced. Furthermore, the low melting lipid D112 was added to enhance release rates as reported for solid lipid implants [49]. *Figure 5-12* gives an overview on release behavior of screened batches.

Having achieved adequate results with the combination of D116 with GMS, this additive was tested in combination with D114. Release from D114 microparticles with as well 5 and 10 % GMS was completed after 3 d releasing 70 % to 90 %. SEM images revealed that the GMS-containing microparticles still remained their shape (*Figure 5-13* exemplarily shows data of D114 + 10 % GMS). An increased roughness upon incubation could be observed, indicating swelling of the matrix followed by an enhanced porosity and water uptake [68] [86].

The addition of the liquid emulsifiers Span 80 and 85 with low HLB values did not show concentration dependent effects. The faster and more complete release could be explained by the liquid state of Span 85 leading to a faster drug liberation and complete degradation of the microparticles, independent of the added concentration as visualized by SEM images (*Figure 5-13* exemplarily shows D114 + 20 % Span 85). Overall, an almost complete drug release could be observed in all three tested formulations.



Figure 5-12: G [6-D-Phe] release from D114 microparticles with addition of Span 40 and GMS (A) and Span 80 and 85 (B), D116 microparticles with addition of Span 85 (C) and Span 80 (D) and with addition of D112/GMS and D112 (E), mean and SD, n = 3

The addition of Span 85 to D116 instead of D114 resulted in almost complete drug release within one day, whereas the particles were stable over two weeks exemplarily shown for D116 + 30 % Span 85 (*Figure 5-13*). Addition of Span 60 and 80 (*Figure 5-12 D*), as well as the combination of GMS and D112 (E) also resulted in a high burst release of 80 to 100 % from D116 microparticles. Only in case of the Span 60 containing microparticles the high burst release could be explained by complete disintegration of the matrix. The other combinations resulted in microparticles which kept their shape.

Adequate release until day 20 could be observed for D114 microparticles with 5 % Span 40. Using D116 as base lipid, release was limited to 9 days. (*Figure 5-12 C*). Both particle qualities remained intact (*Figure 5-13*). A G [6-D-Phe] fraction remained inside the particles due to insufficient interconnectivity of pores. The development of a rougher surface upon incubation reflects the enhanced water uptake into the core of the microparticles [83].

CHAPTER FIVE



CHAPTER FIVE





Figure 5-13: SEM images of freshly prepared (left) and incubated D114 and D116-based particles with addition of Span 40, 60, 80, 85, GMS and D112 for 2 weeks in PBS buffer at 39 °C (right)

Consequently, pure triglyceride matrices did not lead to appropriate release profiles as they do not allow substantial water uptake. Furthermore, the drug load is low and thus an interconnected pore network cannot form upon drug dissolution. Thus, the addition of surfactants plays a major role [84] [87]. Mixing emulsifying agents with triglyceride base materials led in general to an enhanced drug release. Some emulsifiers like Span 40, GMS and CSA showed concentration dependent effects. GMS is reported to either retard [83] [84] or enhance drug release [82] [88] from lipid-based microparticles. We noticed the latter in our study.

Furthermore, the parameters wettability, polymorphic state as well as surface structure seem to influence the release behavior. In some test formulations, we noticed particle breakdown as a consequence of good wettability (D114 + 10 % Span 60). In other formulations, the particle collapsed despite insufficient wettability (D114 + 10 % Span 85), indicating that incubation time and temperature caused the mechanical breakdown.

Crystallization processes occurring in lipid matrices upon accelerated temperatures are reported to influence release behavior [67] [75] [78]. It is not fully understood, whether the polymorph itself or the rearrangements of the crystal lattice are responsible for changes in drug liberation [44] [78]. The influence of the polymorphic state on release could be explained by different mechanisms. On the one hand, the less densely packed hexagonal α -modification may lead to an enhanced drug release compared to the more dense, triclinic β -form [44] [89]. On the other

hand, structural changes upon re-crystallization like blooming, are supposed to influence the drug release [78]. An appropriate release profile could be found throughout the study in microparticles crystallizing in either α - or β -modification. As shown for the D116, GMS and D112-composed particles upon incubation at 39 °C, a transition to the stable polymorph is enabled. Appropriate release profiles over two weeks could be achieved in vitro with microparticles containing either 10 % GMS or 5 % Span 40. These particles kept their shape and had the ability to take up adequate amounts of water.

The ideal candidate for a potential sustained release delivery system for G [6-D-Phe] should guarantee a sustained peptide release of 15 days to ensure cycle blockage in gilts. Based on this requirement, in the first pre-clinical study we focused on the D116 and D118 formulations with 10 % GMS which provided an in vitro release over 22 days with a high burst (*Figure 5-14 A*). Taking the presence of lipases in the mammalian body [90] [91] and a possibly higher degradation rate of the lipid microspheres, body movement and temperature into account, a longer in vitro release than 15 d may result in a promising in vivo effect [50] [92].



Figure 5-14: Identified candidates for clinical studies. (A) Suitable microparticle batches based on D116 and D118 with 10 % GMS applied in pre-clinical study I (September 2014). (B) Formulations prepared of D114 and D116 with different amounts of Span 40 and GMS used in pre-clinical study II (August 2015)

In the second pre-clinical study, microparticles with a faster release based on the combination of D114 with 5 and 10 % Span 40, 10 % GMS as well as D116 + 5 % Span 40 were evaluated (*Figure 5-14 B*).

5.4.6 G [6-D-Phe] release from lipid-based microparticles suspended in thermosensitive poloxamer gels

In order to cope with the problem of a high initial burst, suspensions of microparticles in a thermosensitive poloxamer-based gel were evaluated. Wenzel et al. reported the delivery of deslorelin and GnRH in a thermosensitive poloxamer gel assuring delayed plasma peak levels for the gel compared to the solution after injection in cattle [93].



Figure 5-15: Viscosity measurements of PG in water (A) and PBS pH 7.4 (B) using a temperature ramp from 4 to 34 $^{\circ}$ C (mean and SD, n = 3)

Viscosity measurements revealed a temperature-dependent gelling behavior controlled by poloxamer concentration, whereas gelling showed an earlier onset when PBS was used (15 % poloxamer 407, *Figure 5-15*) confirming findings from literature [94]. The formulation should show a low viscosity at ambient temperature for application and a rapid gelling when brought into the tissue. Thus, formulations with 18 and 20 % poloxamer 407 were thought not feasible for our purpose, as they already showed an increased viscosity at 24 °C and 20 °C. Kempwade et al. found gelling temperatures of 32-34 °C for 18 % and 26 °C for 20 % poloxamer gels [95]. Compared to the aqueous solution of G [6-D-Phe], the release through the gel was delayed. According to other studies, the in vitro release from poloxamer 407 gels is rather short in a range of a few days [96]. 80 % of rIL-2 was released within 8 h from a poloxamer 407 gel in vitro [97]. The prolonged release of G [6-D-Phe] from the gel observed in our study may be caused by protection of the gel from dissolution by the dialysis tubing and the incubation temperature of 39 °C [98] [99]. As free fatty acids are reported to enhance mechanical strength of poloxamer gels, possible interactions with lipid microparticles and surfactants cannot be excluded [94].

The initial burst of microparticles could be reduced compared to suspension in PBS. D114 + 5 % Span 40 did not reveal a reduced burst, but an overall decelerated release (*Figure 5-16*).



Figure 5-16: Release profiles of G [6-D-Phe] dissolved in water and PG 16 % (A) and drug-loaded microparticles after suspension in thermosensitive poloxamer gels (PG) with concentrations of 14 and 16 % (B-E) mean and SD, n = 3

5.5 Conclusion

The aim of this chapter was to investigate whether it is possible to incorporate G [6-D-Phe] into lipid-based microparticles to achieve a continuous release over 15 days, as required for estrus synchronization. Based on previous results with hydrophilic model substances, different triglycerides and surfactants were tested concerning in vitro release behavior, wettability of the lipid matrix as well as particle morphology before and after incubation. The use of pure triglycerides did not result in appropriate release behavior. The addition of release modifying substances was mandatory. With addition of emulsifiers it was possible to tailor the release kinetics concerning burst and completeness of drug release. With the use of GMS and Span 40 it was possible to achieve a release duration of 22 (D118) and 20 d (D114). The burst release could be reduced by the use of suspensions of triglyceride microparticles in thermosensitive poloxamer gels.

To understand the underlying release mechanisms from lipid-based microspheres, DSC and XRPD measurements gave an insight into the crystallized polymorph. In addition, Karl-Fischer titration showed that water uptake and swelling behavior of triglyceride matrices could not be generalized for surfactant type or HLB-value and was more pronounced for short-chain triglycerides in combination with liquid emulsifiers. Surface wettability of triglyceride matrices was insufficient and could mostly be improved by addition of surfactants. The combination of different triglycerides resulted in higher contact angles compared to the individual triglyceride formulation.

5.6 References

- S. D. Putney and P. A. Burke, "Improving protein therapeutics with sustained-release formulations," *Nat. Biotechnol.*, vol. 16, no. 2, pp. 153–157, 1998.
- [2] W. Vogelhuber, E. Magni, M. Mouro, T. Spruss, C. Guse, A. Gazzaniga, and A. Gopferich, "Monolithic triglyceride matrices: a controlled-release system for proteins," *Pharm. Dev. Technol.*, vol. 8, no. 1, pp. 71–79, 2003.
- [3] U. Bilati, E. Allémann, and E. Doelker, "Strategic approaches for overcoming peptide and protein instability within biodegradable nano- and microparticles," *Eur. J. Pharm. Biopharm.*, vol. 59, no. 3, pp. 375–388, 2005.
- M. D. Del Curto, D. Chicco, M. D'Antonio, V. Ciolli, H. Dannan, S. D'Urso, B. Neuteboom, S. Pompili,
 S. Schiesaro, and P. Esposito, "Lipid microparticles as sustained release system for a GnRH antagonist (Antide)," *J. Control. Release*, vol. 89, no. 2, pp. 297–310, 2003.
- [5] S. Mohl and G. Winter, "Continuous release of rh-interferon alpha-2a from triglyceride matrices," J. Control. Release, vol. 97, no. 1, pp. 67–78, 2004.
- [6] H. Grohganz, I. Tho, and M. Brandl, "Development and in vitro evaluation of a liposome based implant formulation for the decapeptide cetrorelix," *Eur. J. Pharm. Biopharm.*, vol. 59, no. 3, pp. 439–448, 2005.
- [7] S. P. Schwendeman, M. Tobio, M. Joworowicz, M. J. Alonso, and R. Langer, "New strategies for the microencapsulation of tetanus vaccine," *J. Microencapsul.*, vol. 15, no. 3, pp. 299–318, 1998.
- [8] D. Yin, Y. Lu, H. Zhang, G. Zhang, H. Zou, D. Sun, and Y. Zhong, "Preparation of glucagon-like peptide-1 loaded PLGA microspheres: characterizations, release studies and bioactivities in vitro/in vivo," *Chem. Pharm. Bull.*, vol. 56, no. 2, pp. 156–161, 2008.
- [9] T. L. Blanchard, D. D. Varner, P. J. Burns, K. A. Everett, S. P. Brinsko, and L. Boehnke, "Regulation of estrus and ovulation in mares with progesterone or progesterone and estradiol biodegradable microspheres with or without PGF2alpha," *Theriogenology*, vol. 38, no. 6, pp. 1091–1106, 1992.
- [10] B. H. Vickery, G. I. McRae, W. V Briones, B. B. Roberts, A. C. Worden, B. D. Schanbacher, and R. E. Falvo, "Dose-response studies on male reproductive parameters in dogs with nafarelin acetate, a potent LHRH agonist," *J. Androl.*, vol. 6, no. 1, pp. 53–60, 1985.
- [11] B. W. Knol, S. J. Dieleman, M. M. Bevers, and W. E. van den Brom, "GnRH in the male dog: dose-response relationships with LH and testosterone," *J. Reprod. Fertil.*, vol. 98, no. 1, pp. 159–161, 1993.
- [12] G. I. McRae, B. B. Roberts, A. C. Worden, A. Bajka, and B. H. Vickery, "Long-term reversible suppression of oestrus in bitches with nafarelin acetate, a potent LHRH agonist," *J. Reprod. Fertil.*, vol. 74, no. 2, pp. 389–397, 1985.
- [13] K. de Mari, L. Maynard, A. Sanquer, B. Lebreux, and H.-M. Eun, "Therapeutic effects of recombinant feline interferon-omega on feline leukemia virus (FeLV)-infected and FeLV/feline immunodeficiency virus (FIV)-coinfected symptomatic cats.," J. Vet. Intern. Med., vol. 18, no. 4, pp. 477–482, 2004.
- [14] A. Doménech, G. Miró, V. M. Collado, N. Ballesteros, L. Sanjosé, E. Escolar, S. Martin, and E. Gomez-Lucia, "Use of recombinant interferon omega in feline retrovirosis: From theory to practice," *Vet. Immunol. Immunopathol.*, vol. 143, no. 3–4, pp. 301–306, 2011.
- [15] T. Minagawa, K. Ishiwata, and T. Kajimoto, "Feline interferon-omega treatment on canine parvovirus infection," *Vet. Microbiol.*, vol. 69, no. 1–2, pp. 51–53, 1999.
- [16] V. Martin, W. Najbar, S. Gueguen, D. Grousson, H. M. Eun, B. Lebreux, and A. Aubert, "Treatment of

canine parvoviral enteritis with interferon-omega in a placebo-controlled challenge trial," *Vet. Microbiol.*, vol. 89, no. 2–3, pp. 115–127, 2002.

- [17] K. P. Brussow, W. Jochle, and U. Huhn, "Control of ovulation with a GnRH analog in gilts and sows," *Theriogenology*, vol. 46, no. 6, pp. 925–934, 1996.
- [18] F. López-Gatius and M. López-Béjar, "Reproductive performance of dairy cows with ovarian cysts after different GnRH and cloprostenol treatments.," *Theriogenology*, vol. 58, no. 7, pp. 1337–48, 2002.
- [19] W. Kanitz, F. Schneider, H. O. Hoppen, C. Unger, G. Nurnberg, and F. Becker, "Pregnancy rates, LH and progesterone concentrations in mares treated with a GnRH agonist," *Anim. Reprod. Sci.*, vol. 97, no. 1–2, pp. 55–62, 2007.
- [20] F. N. Kojima, B. E. Salfen, J. F. Bader, W. A. Ricke, M. C. Lucy, M. F. Smith, and D. J. Patterson, "Development of an estrus synchronization protocol for beef cattle with short-term feeding of melengestrol acetate: 7-11 synch," J. Anim. Sci., vol. 78, no. 8, pp. 2186–2191, 2000.
- [21] H. M. Fraser and D. T. Baird, "Clinical applications of LHRH analogues," *Baillières Clin. Endocrinol. Metab.*, vol. 1, no. 1, pp. 43–70, 1987.
- [22] H. Berger, N. Heinrich, E. Albrecht, U. Kertscher, J. Oehlke, M. Bienert, H. Schafer, I. Baeger, and B. Mehlis, "Gonadotropin-releasing hormone (GnRH) analogs: relationship between their structure, proteolytic inactivation and pharmacokinetics in rats," *Regul. Pept.*, vol. 33, no. 3, pp. 299–311, 1991.
- [23] F. Schneider, W. Tomek, and C. Grundker, "Gonadotropin-releasing hormone (GnRH) and its natural analogues: a review," *Theriogenology*, vol. 66, no. 4, pp. 691–709, 2006.
- [24] H. M. Fraser, J. Sandow, H. Seidel, and W. von Rechenberg, "An implant of a gonadotropin releasing hormone agonist (buserelin) which suppresses ovarian function in the macaque for 3-5 months," *Acta Endocrinol.*, vol. 115, no. 4, pp. 521–527, 1987.
- [25] F. Martinat-Botte, E. Venturi, P. Guillouet, M. A. Driancourt, and M. Terqui, "Induction and synchronization of ovulations of nulliparous and multiparous sows with an injection of gonadotropinreleasing hormone agonist (Receptal)," *Theriogenology*, vol. 73, no. 3, pp. 332–342, 2010.
- [26] J. S. Stevenson, A. P. Phatak, I. Rettmer, and R. E. Stewart, "Postinsemination Administration of Receptal: Follicular Dynamics, Duration of Cycle, Hormonal Responses, and Pregnancy Rates," *J. Dairy Sci.*, vol. 76, pp. 2536–2547, 1993.
- [27] A. Dal Bosco, R. Cardinali, G. Brecchia, P. G. Rebollar, M. Fatnassi, P. Millán, S. Mattioli, and C. Castellini, "Induction of ovulation in rabbits by adding Lecirelin to the seminal dose: In vitro and in vivo effects of different excipients," *Anim. Reprod. Sci.*, vol. 150, no. 1–2, pp. 44–49, 2014.
- [28] E. De Jong, J. Kauffold, S. Engl, J. Jourquin, and D. Maes, "Effect of a GnRH analogue (Maprelin) on the reproductive performance of gilts and sows," *Theriogenology*, vol. 80, no. 8, pp. 870–877, 2013.
- [29] H. C. F. Moore, J. M. Unger, K.-A. Phillips, F. Boyle, E. Hitre, D. Porter, P. A. Francis, L. J. Goldstein, H. L. Gomez, C. S. Vallejos, A. H. Partridge, S. R. Dakhil, A. A. Garcia, J. Gralow, J. M. Lombard, J. F. Forbes, S. Martino, W. E. Barlow, C. J. Fabian, L. Minasian, F. L. Meyskens, R. D. Gelber, G. N. Hortobagyi, and K. S. Albain, "Goserelin for ovarian protection during breast-cancer adjuvant chemotherapy.," *N. Engl. J. Med.*, vol. 372, no. 10, pp. 923–32, 2015.
- [30] G. Bartfai, "Clinical applications of gonadotrophin-releasing hormone and its analogues," *Hum. Reprod.*, vol. 3, no. 1, pp. 51–57, 1988.
- [31] C. Matschke, U. Isele, P. Van Hoogevest, and A. Fahr, "Sustained-release injectables formed in situ and

their potential use for veterinary products," J. Control. Release, vol. 85, no. 1-3, pp. 1-15, 2002.

- [32] L. T. Glickman, N. W. Glickman, G. E. Moore, J. B. Lok, J. W. McCall, and H. B. Lewis, "Comparative effectiveness of sustained-release moxidectin (ProHeart 6) and ivermectin (Heartgard Plus) for the prevention of heartworm infection in dogs in the United States," *Int. J. Appl. Res. Vet. Med.*, vol. 4, no. 4, pp. 339–354, vi–vii, 2006.
- [33] N. J. Medlicott, N. A. Waldron, and T. P. Foster, "Sustained release veterinary parenteral products," *Adv. Drug Deliv. Rev.*, vol. 56, no. 10, pp. 1345–1365, 2004.
- [34] R. Cavalli, O. Caputo, and M. R. Gasco, "Preparation and characterization of solid lipid nanospheres containing paclitaxel," *Eur. J. Pharm. Sci.*, vol. 10, no. 4, pp. 305–309, 2000.
- [35] M. Rawat, D. Singh, S. Saraf, and S. Saraf, "Lipid carriers: a versatile delivery vehicle for proteins and peptides," *J. Pharm. Soc. Japan*, vol. 128, no. 2, pp. 269–280, 2008.
- [36] C. Maenhoudt, N. Santos, and A. Fontbonne, "Suppression of fertility in adult dogs," *Reprod. Domest. Anim.*, vol. 49, no. SUPPL.2, pp. 58–63, 2014.
- [37] C. A. Herbert and T. E. Trigg, "Applications of GnRH in the control and management of fertility in female animals," *Anim. Reprod. Sci.*, vol. 88, no. 1–2, pp. 141–153, 2005.
- [38] T. E. Trigg, P. J. Wright, A. F. Armour, P. E. Williamson, A. Junaidi, G. B. Martin, A. G. Doyle, and J. Walsh, "Use of a GnRH Analogue Implant to Produce Reversible Long-term Suppression of Reproductive Function in Male and Female Domestic Dogs.," *J. Reprod. Fertil. Suppl.*, vol. 57, pp. 255–261, 2001.
- [39] D. J. Jasko, M. E. Farlin, H. Hutchinson, D. M. Moran, E. L. Squires, and P. J. Burns, "Progesterone and estradiol in biodegradable microspheres for control of estrus and ovulation in mares," *Theriogenology*, vol. 40, no. 3, pp. 465–478, 1993.
- [40] L. T. Glickman, N. W. Glickman, G. E. Moore, R. Cobb, S. A. Connell, M. Morrison, and H. B. Lewis,
 "Safety profile of moxidectin (ProHeart 6) and two oral heartworm preventives in dogs," *Int. J. Appl. Res. Vet. Med.*, vol. 3, no. 2, pp. 49–61, 2005.
- [41] Food and Drug Administration, "ProHeart® 6 (Moxidectin)," 2005. [Online]. Available: http://www.fda.gov/downloads/AdvisoryCommittees/CommitteesMeetingMaterials/VeterinaryMedicine AdvisoryCommittee/UCM205054.pdf. [Accessed: 17-Dec-2016].
- [42] W. Zaremba, J. Kauffold, W. Friess, K. Hoffmann, and Y. Yordanova, "Veterinary pharmaceutical composition and use thereof, WO2015EP58265 20150416," 2015.
- [43] B. Albertini, N. Passerini, M. Di Sabatino, B. Vitali, P. Brigidi, and L. Rodriguez, "Polymer-lipid based mucoadhesive microspheres prepared by spray-congealing for the vaginal delivery of econazole nitrate," *Eur. J. Pharm. Sci.*, vol. 36, no. 4–5, pp. 591–601, 2009.
- [44] A. Maschke, C. Becker, D. Eyrich, J. Kiermaier, T. Blunk, and A. Gopferich, "Development of a spray congealing process for the preparation of insulin-loaded lipid microparticles and characterization thereof," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 2, pp. 175–187, 2007.
- [45] F. E. Matos-Jr, M. Di Sabatino, N. Passerini, C. S. Favaro-Trindade, and B. Albertini, "Development and characterization of solid lipid microparticles loaded with ascorbic acid and produced by spray congealing," *Food Res. Int.*, vol. 67, pp. 52–59, 2015.
- [46] P. C. Christophersen, L. Zhang, A. Müllertz, H. M. Nielsen, M. Yang, and H. Mu, "Solid Lipid Particles for Oral Delivery of Peptide and Protein Drugs II - The Digestion of Trilaurin Protects Desmopressin from Proteolytic Degradation," *Pharm. Res.*, pp. 1–9, 2014.

- [47] T. Yoshioka, B. Sternberg, and A. T. Florence, "Preparation and properties of vesicles (niosomes) of sorbitan monoesters (Span 20, 40, 60 and 80) and a sorbitan triester (Span 85)," *Int. J. Pharm.*, vol. 105, no. 1, pp. 1–6, 1994.
- [48] W. C. Griffin, "Classification of surface-active agents by 'HLB," *Chicago Chapter Meeting*. Journal of the society of cosmetic chemists, Chicago, pp. 311–326, 1949.
- [49] M. Schwab, G. Sax, S. Schulze, and G. Winter, "Studies on the lipase induced degradation of lipid based drug delivery systems," J. Control. Release, vol. 140, no. 1, pp. 27–33, 2009.
- [50] M. Schwab, C. M. McGoverin, K. C. Gordon, G. Winter, T. Rades, J. Myschik, and C. J. Strachan, "Studies on the lipase-induced degradation of lipid-based drug delivery systems. Part II - Investigations on the mechanisms leading to collapse of the lipid structure," *Eur. J. Pharm. Biopharm.*, vol. 84, no. 3, pp. 456– 463, 2013.
- [51] S. Jaspart, G. Piel, L. Delattre, and B. Evrard, "Solid lipid microparticles: formulation, preparation, characterisation, drug release and applications," *Expert Opin. Drug Deliv.*, vol. 2, no. 1, pp. 75–87, 2005.
- [52] N. Passerini, B. Perissutti, M. Moneghini, D. Voinovich, B. Albertini, C. Cavallari, and L. Rodriguez, "Characterization of carbamazepine-Gelucire 50/13 microparticles prepared by a spray-congealing process using ultrasounds," *J. Pharm. Sci.*, vol. 91, no. 3, pp. 699–707, 2002.
- [53] P. A. McCarron, R. F. Donnelly, and R. Al-Kassas, "Comparison of a novel spray congealing procedure with emulsion-based methods for the micro-encapsulation of water-soluble drugs in low melting point triglycerides," *J. Microencapsul.*, vol. 25, no. 6, pp. 365–378, 2008.
- [54] B. Albertini, N. Passerini, F. Pattarino, and L. Rodriguez, "New spray congealing atomizer for the microencapsulation of highly concentrated solid and liquid substances," *Eur. J. Pharm. Biopharm.*, vol. 69, no. 1, pp. 348–357, 2008.
- [55] N. Passerini, S. Qi, B. Albertini, M. Grassi, L. Rodriguez, and D. Q. Craig, "Solid lipid microparticles produced by spray congealing: influence of the atomizer on microparticle characteristics and mathematical modeling of the drug release," *J. Pharm. Sci.*, vol. 99, no. 2, pp. 916–931, 2010.
- [56] R. Al-Kassas, R. F. Donnelly, and P. A. McCarron, "Aminolevulinic acid-loaded Witepsol microparticles manufactured using a spray congealing procedure: implications for topical photodynamic therapy," J. *Pharm. Pharmacol.*, vol. 61, no. 9, pp. 1125–1135, 2009.
- [57] M. Windbergs, C. J. Strachan, and P. Kleinebudde, "Investigating the Principles of Recrystallization from Glyceride Melts," AAPS PharmSciTech, vol. 10, no. 4, pp. 1224–1233, 2009.
- [58] M. Takeuchi, S. Ueno, and K. Sato, "Synchrotron Radiation SAXS / WAXS Study of Polymorph-Dependent Phase Behavior of Binary Mixtures of Saturated Monoacid Triacylglycerols," *Cryst. Growth Des.*, vol. 3, no. 3, pp. 369–374, 2003.
- [59] Cremer Oleo GmbH & Co KG, "Excipients for Pharmaceuticals," 2015. [Online]. Available: http://www.cremeroleo.de/de-wAssets/docs/download/01-borschuere-excipient.pdf. [Accessed: 02-Nov-2016].
- [60] T. Yajima, S. Itai, H. Takeuchi, and Y. Kawashima, "Determination of optimum processing temperature for transformation of glyceryl monostearate.," *Chem. Pharm. Bull. (Tokyo).*, vol. 50, no. 11, pp. 1430– 1433, 2002.
- [61] C. Reitz and P. Kleinebudde, "Solid lipid extrusion of sustained release dosage forms," *Eur. J. Pharm. Biopharm.*, vol. 67, no. 2, pp. 440–448, 2007.

- [62] J. H. Whittam and H. L. Rosano, "Physical aging of even saturated monoacid triglycerides," J. Am. Oil Chem. Soc., vol. 52, no. 4, pp. 128–133, 1975.
- [63] K. M. Rosenblatt and H. Bunjes, "Poly(vinyl alcohol) as emulsifier stabilizes solid triglyceride drug carrier nanoparticles in the alpha-modification," *Mol. Pharm.*, vol. 6, no. 1, pp. 105–120, 2009.
- [64] M. Kellens, W. Meeussen, C. Riekel, and H. Reynaers, "Time resolved x-ray diffraction studies of the polymorphic behaviour of tripalmitin using synchrotron radiation," *Chem. Phys. Lipids*, vol. 52, no. 2, pp. 79–98, 1990.
- [65] E. Laine, P. Auramo, and P. Kahela, "On the structural behaviour of triglycerides with time," Int. J. Pharm., vol. 43, no. 3, pp. 241–247, 1988.
- [66] D. Xia, F. Cui, Y. Gan, H. Mu, and M. Yang, "Design of lipid matrix particles for fenofibrate: Effect of polymorphism of glycerol monostearate on drug incorporation and release," *J. Pharm. Sci.*, vol. 103, no. 2, pp. 697–705, 2014.
- [67] J. Schlichter Aronhime, S. Sarig, and N. Garti, "Mechanistic Considerations of Polymorphic Transformations of Tristearin in the Presence of Emulsifiers," J. Am. Oil Chem. Soc., vol. 64, no. 4, pp. 529–533, 1987.
- [68] W. Vogelhuber, E. Magni, A. Gazzaniga, and A. Gopferich, "Monolithic glyceryl trimyristate matrices for parenteral drug release applications," *Eur. J. Pharm. Biopharm.*, vol. 55, no. 1, pp. 133–138, 2003.
- [69] F. Kreye, F. Siepmann, J. F. Willart, M. Descamps, and J. Siepmann, "Drug release mechanisms of cast lipid implants," *Eur. J. Pharm. Biopharm.*, vol. 78, no. 3, pp. 394–400, 2011.
- [70] N. Khan and D. Q. Craig, "The influence of drug incorporation on the structure and release properties of solid dispersions in lipid matrices," J. Control. Release, vol. 93, no. 3, pp. 355–368, 2003.
- [71] V. Rosilio, M. de L. Costa, and A. Baszkin, "Wettability of drug loaded polymer matrices," J. Dispers. Sci. Technol., vol. 19, no. 6–7, pp. 821–841, 1998.
- [72] S. Koennings, A. Berie, J. Tessmar, T. Blunk, and A. Goepferich, "Influence of wettability and surface activity on release behavior of hydrophilic substances from lipid matrices," *J. Control. Release*, vol. 119, no. 2, pp. 173–181, 2007.
- [73] D. Y. Kwok and A. W. Neumann, "Contact angle measurement and contact angle interpretation," Adv. Colloid Interface Sci., vol. 81, no. 3, pp. 167–249, 1999.
- [74] T. Yasuda, T. Okuno, and H. Yasuda, "Contact Angle of Water on Polymer Surfaces," *Langmuir*, vol. 10, no. 7, pp. 2435–2439, 1994.
- [75] F. Kreye, F. Siepmann, A. Zimmer, J. F. Willart, M. Descamps, and J. Siepmann, "Cast lipid implants for controlled drug delivery: importance of the tempering conditions," *J. Pharm. Sci.*, vol. 100, no. 8, pp. 3471–3481, 2011.
- [76] H. Mayama, "Blooming theory of tristearin," Soft Matter, vol. 5, no. 4, pp. 856–859, 2009.
- [77] W. Fang, H. Mayama, and K. Tsujii, "Spontaneous formation of fractal structures on triglyceride surfaces with reference to their super water-repellent properties," J. Phys. Chem. B, vol. 111, no. 3, pp. 564–571, 2007.
- [78] N. Khan and D. Q. M. Craig, "Role of blooming in determining the storage stability of lipid-based dosage forms," J. Pharm. Sci., vol. 93, no. 12, pp. 2962–2971, 2004.
- [79] P. C. Christophersen, L. Zhang, M. Yang, H. M. Nielsen, A. Mullertz, and H. Mu, "Solid lipid particles for oral delivery of peptide and protein drugs I--elucidating the release mechanism of lysozyme during

lipolysis," Eur. J. Pharm. Biopharm., vol. 85, no. 3 Pt A, pp. 473-480, 2013.

- [80] H. Reithmeier, J. Herrmann, and A. Gopferich, "Lipid microparticles as a parenteral controlled release device for peptides," J. Control. Release, vol. 73, no. 2–3, pp. 339–350, 2001.
- [81] S. Herrmann, S. Mohl, F. Siepmann, J. Siepmann, and G. Winter, "New insight into the role of polyethylene glycol acting as protein release modifier in lipidic implants," *Pharm. Res.*, vol. 24, no. 8, pp. 1527–1537, 2007.
- [82] C. M. Adeyeye and J. C. Price, "Development and Evaluation of Sustained-Release Ibuprofen–Wax Microspheres. II. In Vitro Dissolution Studies," *Pharm. Res.*, vol. 11, no. 4, pp. 575–579, 1994.
- [83] L. Lewis, R. Boni, and C. M. Adeyeye, "Effect of emulsifier blend on the characteristics of sustained release diclofenac microspheres," *J. Microencapsul.*, vol. 15, no. 3, pp. 283–298, 1998.
- [84] V. D. Vilivalam and C. M. Adeyeye, "Development and evaluation of controlled-release diclofenac microspheres and tabletted microspheres.," J. Microencapsul., vol. 11, no. 4, pp. 455–470, 1994.
- [85] G. Crotts, H. Sah, and T. G. Park, "Adsorption determines in-vitro protein release rate from biodegradable microspheres: Quantitative analysis of surface area during degradation," *J. Control. Release*, vol. 47, no. 1, pp. 101–111, 1997.
- [86] H. Reithmeier, J. Herrmann, and A. Gopferich, "Development and characterization of lipid microparticles as a drug carrier for somatostatin," *Int. J. Pharm.*, vol. 218, no. 1–2, pp. 133–143, 2001.
- [87] S. Jaspart, P. Bertholet, G. Piel, J. M. Dogne, L. Delattre, and B. Evrard, "Solid lipid microparticles as a sustained release system for pulmonary drug delivery," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 1, pp. 47– 56, 2007.
- [88] C. M. Adeyeye and J. C. Price, "Development and evaluation of sustained-release ibuprofen-wax microspheres. I. Effect of formulation variables on physical characteristics," *Pharm. Res.*, vol. 8, no. 11, pp. 1377–1383, 1991.
- [89] Y. W. Choy, N. Khan, and K. H. Yuen, "Significance of lipid matrix aging on in vitro release and in vivo bioavailability," *Int. J. Pharm.*, vol. 299, no. 1–2, pp. 55–64, 2005.
- [90] B. Richelsen, S. B. Pedersen, K. Kristensen, J. D. Børglum, H. Nørrelund, J. S. Christiansen, and J. O. Jørgensen, "Regulation of lipoprotein lipase and hormone-sensitive lipase activity and gene expression in adipose and muscle tissue by growth hormone treatment during weight loss in obese patients.," *Metabolism.*, vol. 49, no. 7, pp. 906–11, 2000.
- [91] R. Zechner, P. C. Kienesberger, G. Haemmerle, R. Zimmermann, and A. Lass, "Adipose triglyceride lipase and the lipolytic catabolism of cellular fat stores," *J. Lipid Res.*, vol. 50, no. 1, pp. 3–21, 2008.
- [92] C. Olbrich, O. Kayser, and R. H. Muller, "Lipase degradation of Dynasan 114 and 116 solid lipid nanoparticles (SLN)--effect of surfactants, storage time and crystallinity," *Int. J. Pharm.*, vol. 237, no. 1– 2, pp. 119–128, 2002.
- [93] J. G. W. Wenzel, K. S. S. Balaji, K. Koushik, C. Navarre, S. H. Duran, C. H. Rahe, and U. B. Kompella, "Pluronic F127 gel formulations of deslorelin and GnRH reduce drug degradation and sustain drug release and effect in cattle," *J. Control. Release*, vol. 85, no. 1–3, pp. 51–59, 2002.
- [94] G. Dumortier, J. L. Grossiord, F. Agnely, and J. C. Chaumeil, "A review of poloxamer 407 pharmaceutical and pharmacological characteristics," *Pharm. Res.*, vol. 23, no. 12, pp. 2709–2728, 2006.
- [95] A. Kempwade and A. Taranalli, "Formulation and evaluation of thermoreversible, mucoadhesive in situ intranasal gel of rizatriptan benzoate," *J. Sol-Gel Sci. Technol.*, vol. 72, no. 1, pp. 43–48, 2014.

- [96] E. Ruel-Gariépy and J.-C. Leroux, "In situ-forming hydrogels--review of temperature-sensitive systems.," *Eur. J. Pharm. Biopharm.*, vol. 58, no. 2, pp. 409–26, 2004.
- [97] T. P. Johnston, M. A. Punjabi, and C. J. Froelich, "Sustained Delivery of Interleukin-2 from a Poloxamer 407 Gel Matrix Following Intraperitoneal Injection in Mice," *Pharm. Res.*, vol. 9, no. 3, pp. 425–434, 1992.
- [98] P. D. A. Matthias, "In situ-hardening hydrogels for the sustained release of protein pharmaceutics [Dissertation]," LMU Munich, 2015.
- [99] T. Moore, S. Croy, S. Mallapragada, and N. Pandit, "Experimental investigation and mathematical modeling of Pluronic® F127 gel dissolution: Drug release in stirred systems," J. Control. Release, vol. 67, no. 2–3, pp. 191–202, 2000.

6 EX VIVO PRE-EVALUATION OF PARTICLE- AND MODEL-DRUG DISTRIBUTION IN CARCASSES

Parts of this chapter are intended for publication

Investigations on swine carcasses were performed with the help of *Dr. Haukur Lindberg Sigmarsson* at the University of Leipzig (December 2013)

6.1 Abstract

In order to understand the distribution behavior of spray congealed lipid microparticles, dyeloaded microparticles were injected after suspension in a hydrophilic solvent into the neck or caudal thigh muscles of swine carcasses. The injection was performed using 2-ml syringes with 16 or 18 G needles and with an injection revolver for the application of multiple units. The aqueous resuspension medium spread rapidly, confirmed by ultrasound as well as opening the injection site by incision. The microparticles could be found at the injection site upon incision. With the help of ultrasound, it was not possible to detect the microparticles in the muscle.

6.2 Introduction

Microparticulate drug delivery systems are typically applied subcutaneously or intramuscularly [1] [2]. Aiming mostly for a systemic effect with a controlled or delayed release [3]–[5], it is important to investigate the distribution of microparticles after injection. This includes whether the microparticles are enclosed by a fibrous capsule, and if a local inflammatory reaction or transport takes place in vivo [6] [7]. A general problem arising from a depot formation may be a high local concentration of the residual drug [8]–[11]. The injection of depot formulations may also impair meat quality at the injection site on food producing animals [12]. The use of biodegradable microspheres or liposomes may minimize tissue lesions in food producing animals [11] [10].

Our purpose was the development of a delivery system, which is able to control the release of the peptide Gonadorelin [6-D-Phe] in the porcine muscle tissue and which is easy to inject. The reconstituted microparticles are intended for application via a syringe with a 16 or 18 G needle, but for a more effective injection for multiple animals, the use of an injection-revolver should be preferred. This device is commonly used for veterinary products with a higher importance of multiple-dose delivery systems, also mirrored in a variety of patents [13]–[15]. Consequently, the distribution of microparticles was to be studied in carcasses in order to understand the spreading behavior and to gain a preliminary insight into suspension performance in the injection device for the upcoming pre-clinical studies. Additionally, using the dye-loaded microparticles, the dye distribution in the surrounding tissue was to be assessed.

6.3 Materials and Methods

6.3.1 Materials

Methylene blue and crystal violet, as well as polysorbate 20 (PS 20) were purchased from Merck KGaA (Darmstadt, Germany). Dynasan 116 (tripalmitin, D116) was kindly provided from Cremer Oleo (Witten, Germany). The gelling agent Tylopur[®] C 300 (sodium carboxymethylcellulose, Na CMC) was kindly donated from Clariant (Muttenz, Switzerland). The suspended microparticles were applied via a 2-ml syringe equipped with a 16 or 18 G-needle (B. Braun Melsungen AG, Melsungen, Germany) or an injection revolver HSW ECO-MATIC[®] (Henke Sass Wolf GmbH, Tuttlingen, Germany) with an injection volume of 2.0 ml. The needle used in this study had a length of 40 mm and a diameter of 2 mm.

Two swine carcasses used in this study were fully mature German Landrace x Piétrain and were kindly donated by the Faculty of veterinary medicine of the University of Leipzig.

6.3.2 **Preparation of dye loaded lipid microparticles by spray congealing**

For preparation of 1 % dye-loaded microparticles, the lipid dispersion was filled in the sample container of the B-290 Mini spray dryer with additional spray chilling setup (Büchi, Flawil, Switzerland), which was pre-conditioned for 1.5 h prior to the production step. The additional equipment consisted of a sample container, surrounded by a bath filled with polyethylene glycol (PEG) 400, also heating the nozzle. The connection between nozzle and sample container was heated externally by a resistance wire-based heating, generating a temperature of approximately 70 °C. The spray tower was fed with pre-cooled air produced by an additional dehumidifier Deltatherm[®] LT (Deltatherm[®] Hirmer GmbH, Much, Germany). Lipid melt was sprayed using nitrogen as spray gas and the following apparatus settings:

Spray-parameter	Value
Spray pressure [bar]	5
Spray flow [l/h]	357
Aspirator power [%]	100
Valve opening	1.25 turns
Temperature oil bath [°C]	90
Inlet temperature [°C]	15
Pressure filter [mbar]	96

The produced lipid microparticles were separated using the high-performance cyclone from Büchi and stored at 2-8 °C.

6.3.3 Scanning Electron Microscopy

A Jeol JSM 6500-F SEM (Jeol, Tokyo, Japan) was used at 2 kV. Particles were fixed on aluminum sample holders with double adhesive tape (Plano, Wetzlar, Germany) and were analyzed without an additional graphite coating.

6.3.4 **Preparation of the reconstitution medium**

Na CMC was added to an aqueous 0.02 % PS 20 solution under stirring in concentrations of 0.5, 1.0 and 2.0 % (m/V) until a clear solution was obtained.

6.3.5 Application of microparticles

Approximately 50 mg of microparticles were mixed with 7 ml of reconstitution medium and shaken until a homogenous suspension was obtained. For ultrasound investigations, it was tried to visualize the microparticles in a higher concentration by suspension of 50 mg in 4 ml of the reconstitution medium. 2 ml of the suspension were injected into the lateral neck (*m. rhomboideus cervices, m. rhomboideus capitis, m. semispinalis capitis, m. complexus, m. serratus ventralis cervices*) or caudal thigh muscles (*m. biceps femoris, m. semitendinosus, m. semimembranosus*) via a 2-ml syringe using a 16/18 G needle or the injection revolver. The muscle tissues were examined concerning visibility of a bolus, distribution of hydrophilic color and localization of the microparticles. Ultrasonographic examination was performed using an esaote myLab ultrasonic device (Esaote Biomedica Deutschland GmbH, Köln, Germany).

Additionally, the injection sites were examined surgically. Therefore, the skin was cut crosswise with a scalpel. The cross-section was carefully held open to examine and to document the color of the spreading dye and the localization of the microparticles.

6.4 Results and Discussion

None of the tested suspensions with 0.5, 1 or 2 % Na CMC showed sedimentation effects during the handling procedure within the study and could be loaded into 2-ml syringes and into the injection revolver. Suspensions of methylene blue-loaded microparticles showed a light blue color approximately 10 min after preparation (*Figure 6-1 B*), indicating a surface burst release of the dye into the surrounding medium. Suspensions of crystal violet showed a minor coloration and were therefore used for the determination of the particle behavior within the tissue (C).



Figure 6-1: SEM image of methylene blue-loaded microparticles (A). Suspension of methylene blueloaded lipid microparticles prior to application via a syringe (B); crystal violet-loaded lipid particles prior to application (C) distribution of methylene blue along the muscle fibers in the lateral neck muscles after injection and longitudinal and horizontal incision (D); crystal violet-loaded-lipid microparticles after injection in the lateral neck muscles (E) and after injection in the caudal thigh muscles (F)

Both suspensions of methylene blue- and crystal violet-loaded microparticles were easily syringeable via 16 and 18 G needles, as well as via the injection revolver into the lateral neck muscle. *Figure 6-1 D* shows the distribution of the dye along the muscle fibers, indicating a rapid diffusion of hydrophilic dye into the surrounding tissues although using a carcass. The diffusion might even be enhanced when additional movement and body temperature are given. Particle residues were not carried away from the injection site and remained in place (*Figure 6-1 E, F*). Generally, lesions at injection sites are reported in literature for different kinds of food producing animals, which can affect the meat quality concerning tenderness and texture [16]. Moreover, it is a fact that a more severe damage in food producing animals is caused by the injection of suspensions compared to the injection of solutions [10] [12].



Figure 6-2: Ultrasonographic examination of the lateral neck muscles (A), the caudal thigh muscles directly after injection (B) and 1 h after injection (C)

Additionally, both injection sites, lateral neck and caudal thigh muscles, were characterized via ultrasonography (*Figure 6-2 A-C*). The pig skin and underlying fat layers demonstrate several peculiarities in contrast to other mammalian skin. The fat layer consists of five sublayers with different thicknesses and fatty acid compositions [17] [18]. Since it was not possible to visualize the placement of microparticles after injection into the lateral neck muscles, as they are covered by a thicker fat layer, the caudal thigh muscles seemed to be a more suitable site for injection and monitoring by ultrasonography. Ultrasound images showed faint shadows along the muscle fibers indicating the reconstitution medium, underlining the results obtained by incision [19]. Immediately after injection, the liquid reconstitution medium was clearly noticeable (*Figure 6-2 B*), but almost faded away after one hour (*Figure 6-2 C*). Nevertheless, the tested microparticles remained in the muscle tissue. The formation of a temporary depot in the tissue

due to enhanced viscosity of the reconstitution medium, as visible after injection of oily suspensions or in-situ forming implants [20], was not observable in both injection sites. For oily suspensions, the depot-effect is dependent on the solubility of the drug in the medium and the surrounding tissue [21].

6.5 Conclusion

The ex vivo pre-testing studies gave us first insights into the behavior of lipid microparticles after injection into the two major injection sites in swine. Furthermore, it was possible to evaluate the injection via a single-dose injection with a syringe, or a multiple-dose device, the commonly used injection revolver. In both cases, the injection could be easily performed.

The tested hydrophilic dye was partly dissolved out of the lipid particles upon reconstitution and could be found along the muscle fibers, whereas the lipid particles remained on the injection site. These findings could be confirmed by investigations using ultrasounds, whereby a visualization of the microparticles was not possible. The reconstitution medium consisting of Na CMC in concentrations of 0.5, 1.0, 2.0 % (m/V) and PS 20 0.02 % (m/V) seemed to be a good candidate for further development and optimization, with respect to pH and isotonicity for a following use in the in vivo study. In addition, it would be beneficious to keep the amount of injected microparticles in a low range to avoid tissue lesions leading to minor meat quality.

6.6 References

- [1] H. Reithmeier, J. Herrmann, and A. Gopferich, "Development and characterization of lipid microparticles as a drug carrier for somatostatin," *Int. J. Pharm.*, vol. 218, no. 1–2, pp. 133–143, 2001.
- M. D. Del Curto, D. Chicco, M. D'Antonio, V. Ciolli, H. Dannan, S. D'Urso, B. Neuteboom, S. Pompili,
 S. Schiesaro, and P. Esposito, "Lipid microparticles as sustained release system for a GnRH antagonist (Antide)," *J. Control. Release*, vol. 89, no. 2, pp. 297–310, 2003.
- [3] N. Passerini, B. Perissutti, B. Albertini, D. Voinovich, M. Moneghini, and L. Rodriguez, "Controlled release of verapamil hydrochloride from waxy microparticles prepared by spray congealing," *J. Control. Release*, vol. 88, no. 2, pp. 263–275, 2003.
- [4] S. Jaspart, P. Bertholet, G. Piel, J.-M. Dogné, L. Delattre, and B. Evrard, "Solid lipid microparticles as a sustained release system for pulmonary drug delivery," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 1, pp. 47– 56, 2007.
- [5] C. M. Adeyeye and J. C. Price, "Development and evaluation of sustained-release ibuprofen-wax microspheres. I. Effect of formulation variables on physical characteristics," *Pharm. Res.*, vol. 8, no. 11, pp. 1377–1383, 1991.
- [6] E. I. Shishatskaya, O. N. Voinova, A. V. Goreva, O. A. Mogilnaya, and T. G. Volova, "Tissue reaction to intramuscular injection of resorbable polymer microparticles," *Bull. Exp. Biol. Med.*, vol. 144, no. 6, pp. 786–790, 2007.
- [7] A. Cambronero-Rojas, P. Torres-Vergara, R. Godoy, C. Von Plessing, J. Sepúlveda, and C. Gómez-Gaete, "Capreomycin oleate microparticles for intramuscular administration: Preparation, in vitro release and preliminary in vivo evaluation," *J. Control. Release*, vol. 209, pp. 229–237, 2015.
- [8] A. Sanquer, G. Wackowiez, and B. Havrileck, "Qualitative assessment of human exposure to consumption of injection site residues," *J. Vet. Pharmacol. Ther.*, vol. 29, no. 5, pp. 345–353, 2006.
- [9] A. Sanquer, G. Wackowiez, and B. Havrileck, "Critical review on the withdrawal period calculation for injection site residues," J. Vet. Pharmacol. Ther., vol. 29, no. 5, pp. 355–364, 2006.
- [10] P. T. Reeves, "Residues of veterinary drugs at injection sites," J. Vet. Pharmacol. Ther., vol. 30, no. 1, pp. 1–17, 2007.
- [11] G. Winzenburg, C. Schmidt, S. Fuchs, and T. Kissel, "Biodegradable polymers and their potential use in parenteral veterinary drug delivery systems," *Adv. Drug Deliv. Rev.*, vol. 56, no. 10, pp. 1453–1466, 2004.
- [12] J. Van Donkersgoed, P. L. Dubeski, M. VanderKop, J. L. Aalhus, S. Bygrove, and W. N. Starr, "The effect of animal health products on the formation of injection site lesions in subprimals of experimentally injected beef calves," *Can. Vet. J.*, vol. 41, no. 8, pp. 617–622, 2000.
- [13] W. R. Tarello, C. Lopez, L. A. Gordon, and W. B. Whalen, Thoams D. Harvey, "Multiple medicament cartridge clip and medicament discharging device therefor," US4518384 A, 1985.
- [14] R. Denance, "Apparatus for performing intradermal, subcutaneous of intramuscular injections," US4512767 A, 1985.
- [15] K. Schreijag, "Veterinary syringe," US7341574 B2, 2008.
- [16] M. H. George, P. E. Heinrich, D. R. Dexter, J. B. Morgan, K. G. Odde, R. D. Glock, J. D. Tatum, G. L. Cowman, and G. C. Smith, "Injection-site lesions in carcasses of cattle receiving injections at branding

and at weaning.," J. Anim. Sci., vol. 73, no. 11, pp. 3235-3240, 1995.

- [17] H. K. Dean and T. P. Hilditch, "The body fats of the pig," *Biochem. J.*, vol. 27, no. 6, pp. 1950–1956, 1933.
- [18] J. D. Wood, M. Enser, F. M. Whittington, C. B. Moncrieff, and A. J. Kempster, "Backfat composition in pigs: Differences between fat thickness groups and sexes," *Livest. Prod. Sci.*, vol. 22, no. 3–4, pp. 351– 362, 1989.
- K. Hynynen, "The threshold for thermally significant cavitation in dog's thigh muscle in vivo," Ultrasound Med. Biol., vol. 17, no. 2, pp. 157–169, 1991.
- [20] N. J. Medlicott, N. A. Waldron, and T. P. Foster, "Sustained release veterinary parenteral products," *Adv. Drug Deliv. Rev.*, vol. 56, no. 10, pp. 1345–1365, 2004.
- [21] J. Zuidema, F. A. J. M. Pieters, and G. S. M. J. E. Duchateau, "Release and absorption rate aspects of intramuscularly injected pharmaceuticals," *Int. J. Pharm.*, vol. 47, no. 1–3, pp. 1–12, 1988.

7 IN VIVO EVALUATION OF G [6-D-PHE]-LOADED SOLID LIPID MICROPARTICLES WITH SPECIAL REGARD ON ESTRUS CONTROL

Parts of this chapter are intended for publication

Data and results in this chapter were gained in cooperation with *Prof. Dr. Johannes Kauffold* and the **Faculty of Veterinary Medicine of the University of Leipzig**. Investigations on swine were performed by *Catherine Poser* and *Rosa Stark*, PhD students of *Prof. Dr. J. Kauffold* with the help of *Dr. Haukur Lindberg Sigmarsson* and *Dr. Matthias Hoops*. Analysis of the obtained data with corresponding statistics was performed by me at the LMU in Munich.

7.1 Abstract

For evaluation of an ecologically compatible alternative intended for cycle blockage and estrus control, G [6-D-Phe]-loaded lipid microparticles were tested in vivo. Two different reconstitution media were applied to investigate applicability and syringeablilty of the suspension, revealing suspension properties for a PVP containing medium. In general, lipidbased microparticles were well tolerated throughout the studies with only mild reactions on injection sites. To evaluate the appropriate dose, 750 µg per animal and 3750 µg per animal were administered via different formulations. The low peptide concentration was sufficient for a reliable cycle blockage in swine. It was possible to achieve a cycle blockage between 6.4 ± 0.2 d with D114 + 5 % Span 40 and 14.2 \pm 4.9 d with D118 + 10 % GMS. Onset of follicular growth could be controlled between 7 and 17 d depending on the formulation, with significantly delayed onset of follicular growth in the group treated with D114 + 5 % Span 40 microparticles. Ovulation was significantly reduced using the high dose formulations compared to placebo. The incidence of temporary and permanent ovarian cysts was higher in the treatment groups compared to the controls, without statistical evidence. Thus, it could be demonstrated that G [6-D-Phe] loaded lipid microparticles are a promising environmentally sustainable alternative for cycle blockage and estrus control in swine.

7.2 Introduction

For economic reasons in livestock breeding, animals are estrus synchronized to allow a fixedtime insemination at a predictable time point [1]–[4]. A variety of gestagens is currently used to achieve a cycle blockage [5]–[7]. Allyl trenbolone, also referred to as altrenogest, was one of the first synthetic progestogens daily administered to influence the estrus in gilts. A highly reliable estrus control can be achieved when fed in doses between 10 and 20 mg/d per animal over 14 to 19 days [8]–[11]. Further investigations indicate, that a treatment with altrenogest positively influences fertility and litter size of treated animals [12].

Besides the use in livestock breeding, gestagens like levonorgestrel and drospirenone are also active ingredients of hormonal contraceptives for women [13] [14]. Concerns about the intensive use of gestagens arise from their entry into the environment harming fish reproduction which results in infertility and masculinization [15]–[17]. Carlsson and co-workers tested a variety of drugs towards their effect on the water environment and identified the gestagen norethisterone as a substance with high toxicity for water organisms with possible long-term effects [18]. The problem is more pronounced due to the absent or if only minimal metabolization [16] [19] [20]. Therefore, there is a need for alternatives in the food-producing animal industry, where large amounts of gestagens are fed daily and which can enter the surrounding runoff, ground and water [19] [21]–[23]

To achieve estrus control with minimal effects on the environment, the application of GnRH analogues, like G [6-D-Phe] is of high interest. The original GnRH decapeptide pGlu¹-His²-Trp³-Ser⁴-Tyr⁵-Gly⁶-Leu⁷-Arg⁸-Pro⁹-Gly¹⁰-NH₂ and G [6-D-Phe] are completely metabolized to non-toxic components [24]–[26]. The introduction of bulky and hydrophobic amino acids in position 6 of the peptide backbone is an important structural change to reduce degradation and increase bioactivity [25] [27] [28]. G [6-D-Phe] with D-Phe in position 6 instead of Gly is a so called superagonist for GnRH I, which has a 2.5-fold longer half-life compared to GnRH and shows higher resistance to degradation, due to modification in the cleavage point for endogenous peptidases [26] [27] [29].

A drawback of G [6-D-Phe] against gestagens in estrus synchronization is the need for parenteral application. Since daily injection is not desirable in food animal health, a sustained release system appears mandatory. Different delivery systems for GnRH analogues have been reported in literature for a diversity of mammalian species [30] [31]. Wenzel et al. incorporated GnRH in a sustained release poloxamer gel to achieve a sustained release after depot injection

in cattle and to protect the drug against degradation [32]. Further approaches with a successful achievement of synchronized estrus was reported in beagle bitches with the incorporation of leuprolide acetate into PLGA microparticles followed by treatment with fertirelin [33]. Furthermore, a long-term suppression of sexual functions in male and female dogs can be achieved with Suprelorin[®] (Virbac), a deslorelin containing implant based on hydrogenated palm oil and Phosphatidylcholine. Similarly, the sexual functions in the macaque could be controlled for 3 to 5 months with the delivery of buserelin via an implant [30] [31] [34]–[36]. These approaches and promising results from previously performed in vitro studies on G [6-D-Phe] loaded microparticles pathed the way for in vivo testing of these systems in swine. Effects on cycle blockage, ovulatory behavior, side effects and biocompatibility were to be evaluated [37]. Based on a previous study on continuous application of G [6-D-Phe] via pumps, consequently 750 μ g or 3750 μ g per animal were tested [38].

CHAPTER SEVEN

7.3 Materials and Methods

7.3.1 Materials

Gonadorelin [6-D-Phe] (G [6-D-Phe]) was donated by Veyx Pharma GmbH (Schwarzenborn, Germany). Triglycerides Dynasan 114 (trimyristin, D114), Dynasan 116 (tripalmitin, D116) and Dynasan 118 (tristearin, D118) were kindly provided from Cremer Oleo (Witten, Germany). Glycerol monostearate (GMS) with a monoester content of 40-55 % was purchased from Caelo (Caesar & Loretz, Hilden, Germany). Sorbitan monopalmitate (Span 40) was purchased from Croda (Nettetal, Germany). Mannitol and centrifuge tubes were purchased from VWR International GmbH (Darmstadt, Germany). Distilled water produced with a distillation apparatus (Wagner und Munz, München, Germany) was used for preparation of reconstitution media. Highly purified water was produced using a Milli-Q Water system, (Millipore, MA, USA). The gelling agent Tylopur[®] C 300 (sodium carboxymethylcellulose, Na CMC) was kindly donated from Clariant (Muttenz, Switzerland) and Kollidon[®] 90F (Polyvinyl pyrrolidone, PVP) was acquired from BASF (Ludwigshafen, Germany). Polysorbate 20 (PS 20) and polysorbate 80 (PS 80) were purchased from Merck KGaA (Darmstadt, Germany).

7.3.2 Animals

German Landrace x Piétrain gilts with a mean weight of 143.8 kg and an average age of 211 d at the beginning of the study were used. The animals were weaned from the mother after 28 d and kept at a flat deck for further 49 d. Afterwards, the animals were housed in a separate stable in pens à 2 animals and fed dependent on the age according to the guidelines of the Deutsche Landwirtschaftliche Gesellschaft (DLG) [39].

7.3.3 Preparation of G [6-D-Phe] loaded microparticles by spray congealing

Lyophilized G [6-D-Phe] was filled in high grade stainless steel beakers and milled using a Retsch[®] Cryomill (Retsch Technology, Haan, Germany) with two stainless steel balls (10 mm diameter) applying a precooling time of 10 minutes at 5 Hz followed by the actual milling step of 4 minutes at 20 Hz. The obtained powder was aliquoted in 6 R vials, closed with a rubber stopper and crimped. Lipid compounds were melted on a hot plate at approximately 90 °C (D114, D116) and 110 °C (D118), the aliquot of peptide was added and homogenized using a T-10 basic Ultraturrax (IKA Laboratory Technology, Staufen, Germany) for 2 min. The lipid dispersion was poured into the sample container of the B-290 Mini spray dryer (Büchi, Flawil, Switzerland) with additionally installed spray congealing equipment, which was preconditioned for 1.5 h prior to the production step. The lipid melt was atomized using nitrogen at a pressure of 6 bar. A spray flow of 414 l/h was used resulting in a filter pressure of 92 mbar.

The aspirator power was set to 100 % throughout all experiments. A dehumidifier Deltatherm[®] LT (Deltatherm[®] Hirmer GmbH, Much, Germany) was installed keeping the inlet temperature at 13-15 °C.

7.3.4 **Determination of drug content**

20 mg microparticles were weighed into centrifuge tubes, 2.0 ml of methylene chloride and 2.0 ml of highly purified water were added and the tubes were placed on a horizontal shaking incubator GFL 3031 (Gesellschaft für Labortechnik, Burgwedel, Germany) at 39 °C and 60 rpm for 12 h. Samples of 1 ml were taken from the water phase and analyzed using RP-HPLC. All extractions were performed in triplicate. RP-HPLC analysis was performed using an Agilent RP-HPLC system (Agilent, Santa Clara, CA, USA), supplied with a 250 x 4.6 mm Luna C-8 column(5µm) (Phenomenex, Aschaffenburg, Germany) and a Security Guard[™] KJO-482 precolumn (5µm) (Phenomenex, Aschaffenburg, Germany). Columns were maintained at 40 °C using the following gradient:

Time [min]	A [%]	B [%]
0	65	35
1	65	35
21	65	35
22	10	90
26	10	90
35	65	35

Mobile Phase A: 1000 ml highly purified water + 1 ml trifluoroacetic acid

Mobile Phase B: 800 g Acetonitrile + 200 g water + 1.2 ml trifluoracetic acid

7.3.5 Aliquoting of the particles under laminar air flow

Based on the drug content (*see* 7.3.4) microparticles delivering 750 μ g or 3750 μ g per animal for the treatment of six animals were weighed under laminar air flow (LAF) into sterile 10 R vials, closed with rubber stoppers and crimped.

7.3.6 **Preparation of the reconstitution medium**

For the pre-clinical studies, two different reconstitution media were used. Firstly, the reconstitution medium consisted of 1 % Na CMC and 0.02 % PS 20. The solution was filtrated through a 5 μ m-syringe filter into sterilized 10 R glass vials under LAF closed with sterile rubber stoppers and sterilized in a table autoclave GTA 50 (Medizin und Labortechnik Fritz Gössner, Hamburg, Germany). Optimized reconstitution medium was prepared under LAF

conditions using the above-mentioned apparatus by adding 0.5 % PS 80, mannitol for isotonization and 3 % Kollidon[®] 90 F to distilled water while stirring followed by a sterile filtration through a 0.22 μ m syringe filter. Each vial contained 20 ml of reconstitution medium and was closed with sterilized rubber stoppers and crimped. The vials were stored in the refrigerator until application.

7.3.7 Viscosity measurements

Both reconstitution media were analyzed on a rotation viscometer MCR 100 rheometer (Physica, Anton Paar, Ostfildern, Germany) applying the cone-plate-method at 25 °C. Shear rates between 1 s⁻¹ and 100 s⁻¹ were applied. Both pure reconstitution medium as well as particle suspension was tested. 100 mg of particles were suspended in 4.0 ml of reconstitution medium. Each measurement was performed in triplicate.

7.3.8 General study set-up and pre-treatment of the animals

Both in vivo studies were conducted at the Versuchsgut Oberholz of the University of Leipzig between September 2014 and August 2015. Gilts specifically bred for this purpose at age of 180 days were brought in contact with a boar to detect possible estruses. The study started with the ultrasound examination to assure onset of puberty in all participating animals. If this was not the case, puberty was induced using Intergonan[®] 240 IE/ml (equine chorionic gonadotropin, eCG, Intervet, Unterschleißheim, Germany). This treatment was only necessary for six animals out of 25 in the first study. These animals were again examined for detection of puberty status 8 d later [39].

Prior to application, all gilts were estrus-synchronized to assure that every animal reached day 12 of the sexual cycle with the 46 day of the study. For cycle synchronization 20 mg/animal Regumate[®] (altrenogest, Intervet, Unterschleißheim, Germany) were applied starting on day 11 for 28 d. 40 h after the last altrenogest application, 1.000 IE Intergonan[®] were injected i.m. to induce follicular growth. Ovulation was induced by intramuscular injection of 50 μ g/ml Gonavet[®] Veyx (G [6-D-Phe], Veyx Pharma GmbH, Schwarzenborn, Germany). With day 46 of the study, gilts were randomized into five groups, consisting of four treatment groups and one control group. The injection of the different formulations started on the same day. *Table 7-1* gives an overview on formulations tested in the first and second pre-clinical study.
Pre-clinical study 1	Triglyceride	Additive/concentration	G [6-D-Phe]
			concentration
			[µg/animal]
1	D118	-	0 (Placebo)
2	D116	GMS/10 %	750
3	D118	GMS/10 %	750
4	D116	GMS/10 %	3750
5	D118	GMS/10 %	3750
Pre-clinical study 2	Triglyceride	Additive/concentration	G [6-D-Phe]
			concentration
			[µg/animal]
6	D114	Span 40/5 %	0 (Placebo)
7	D116	Span 40/5 %	750
8	D114	GMS/5 %	750
9	D114	Span 40/5 %	750
10	D114	Span 40/10 %	750

Table 7-1: Overview of applied treatments and G [6-D-Phe] concentrations per group. Groups 1-5 were investigated during the first pre-clinical study in September 2014, groups 6-10 were investigated in August 2015 in the second study

From day 34 on, the gilts were examined clinically and sonographically with focus on cycle state. Moreover, an additional detection of possible estruses was performed with the presence of a boar. Injection of microparticle formulations was performed on day 46 followed by daily visible inspection of injection sites. All animals were investigated at least 25 days and maximum 31 days after injection of the formulations to detect the end of cycle blockage and following follicular growth and ovulation [39].

7.3.9 **Reconstitution and application of the suspension**

Microparticles and reconstitution medium were delivered in two separate vials and mixed prior to application. In general, 50 to 100 mg microparticles (depending on the actual drug load) were mixed with 2.0 ml of corresponding reconstitution medium. Each vial contained single doses for potentially six animals. The vials were gently shaken for homogenization avoiding foam formation. Previous experiments suggested a manual shaking for 2 min, whereas the practical application needed reconstitution times of 15 min until homogeneity was reached with the high dose formulations. The multi-dose containers were assembled with the injection revolver and the respective amount of suspension was injected into the lateral neck muscles. High dose

formulations (3750 μ g/animal) were suspended in 4 ml of reconstitution medium and 2 x 2 ml were injected [39].

7.3.10 General examination and evaluation of injection sites

Gilts were investigated concerning interior body temperature with the means of rectal temperature measurement, food intake, general condition and clinical signs of disease between days 34 to 72/78. Injection sites were evaluated with special focus on redness, pain, swelling and temperature. Results were documented photographically [39].

7.3.11 Determination of estrus control

Besides contact with a boar, estrus was detected via redness and swelling of the genitals, as well as beginning and end of "standing" according to a pre-defined schedule. "Standing" was detected via established tests, e.g. back-pressure test by instructed personnel. This data was gained as additional information to assure a gilt's estrus [39].

7.3.12 Ultrasound examination of ovaries

Ultrasound examinations were performed using a Fazone CB device (Fujifilm, Tokyo, Japan) equipped with a convex transducer type C 9-3, achieving a frequency of 3-9 MHz. For our investigations, a frequency of 6 MHz was used, which assured a penetration depth of 10 cm. Overall gain was set at 84 dB. For examination, the animals were brought into a custom-made box and immobilized by feeding. Ultrasound was performed transcutaneously according to Kauffold et al., by positioning the transducer from the right ventro-lateral abdominal wall in dorsal direction to the last pair of teats and cranial to the hind leg [40]. Structures like follicles, corpora haemorrhagica and corpora lutea were monitored and documented. The term ovarial cysts was used for structures consisting of growing follicles with a diameter larger than 10 mm showing no ovulation. These structures were persistent besides ovulating follicles (solitary cysts) or exclusively, then called polycystic ovary degeneration (POD). If they were persistent solely over a defined period without influencing the cycle blockage, they were called temporary cysts. If monitored over the whole-time period, they were called permanent cysts. Ovulation was set as complete, if pre-ovulatory follicles vanished and just corpora haemorrhagica were present. Each measurement consisted of at least three representative follicles and if present, three *corpora lutea* to depict the onset of new follicle growth [41]. This value consisted of the mean value out of three single measurements for each day, respectively [39].

7.3.13 Statistical Analysis

Statistical Analysis was carried out using SigmaPlot (Systat Software Inc., San Jose, CA, USA).

7.4 Results and Discussion

7.4.1 Characterization of reconstitution behavior

Viscosity measurements were conducted to evaluate applicability of the particle suspensions (*Table 7-2*).

Formulation	Viscosity [Pas]	Shear rate [s ⁻¹]
Na CMC 1 % medium	0.041 (± 0.002)	1
Na CMC 1 % suspension	20.9 (± 13.1)	1
Na CMC 1 % medium	0.036 (± 0.002)	100
Na CMC 1 % suspension	0.050 (± 0.006)	100
PVP 3 % medium	0.014 (± 0.007)	1
PVP 3 % suspension	8.0 (± 4.2)	1
PVP 3 % medium	$0.007~(\pm 0.004)$	100
PVP 3 % suspension	$0.012 (\pm 0.002)$	100

Table 7-2: Viscosities of reconstitution media and final suspensions

In literature, shear thinning and dilatancy are reported for suspensions, depending on medium and concentration of the dispersed phase [42] [43]. Both reconstitution media and suspensions showed shear thinning behavior. Shear thinning was more pronounced when Na CMC was used as thickening agent compared to PVP, possibly due to linear orientation of the colloids like cellulose upon mechanical stress [44]. Na CMC based reconstitution medium showed a higher viscosity without particles at low shear rates (0.041 ± 0.002 Pas), as well as after suspension (20.9 ± 13.1 Pas) compared to the PVP based medium (0.014 ± 0.005 Pas without particles, 8.0 ± 4.2 Pas with particles). With suspension of particles, both media showed a remarkably reduced viscosity after exposure to shear stress in comparison to the unsheared sample indicating an orientation of the particles upon shearing. Foam formation was only noticeable upon shaking the Na CMC based formulation with the potential of particle loss at the surface and inhomogeneity (*Figure 7-1*). Thus, the reconstitution medium based on PVP appeared to be more suitable for reconstitution of the particles, especially with a suspension start viscosity of 8 Pas, and a marked shear thinning to 0.012 Pas, which should enable easy syringeability and injectability.

7.4.2 Applicability of the final suspensions in vivo

For both studies two different compositions of reconstitution media were evaluated.



Figure 7-1: Particles suspended in reconstitution medium (Na CMC 1 %, PS 20 0.02 %) with formation of foam due to manual shaking and absorption of particles to the surface (A) kindly provided from the Faculty of Veterinary Medicine, University of Leipzig. Placebo formulation (D114 + 5 % Span 40) during homogenization in reconstitution medium (PVP 3 %, PS 80 0.5 %, mannitol 5 %) (B). Homogenous suspension of particles in PVP-based reconstitution medium (C)

Na CMC-based medium did not show satisfying reconstitution properties specifically for the high dose formulations (3750 μ g/animal). Particles showed a marked tendency towards agglomeration (*Figure 7-1 A*). To achieve a homogenous suspension, the vials needed to be shaken manually for 15 to 30 min, which is not practicable for a market product. Furthermore, the formation of foam made it difficult to withdraw the desired volume, which necessitated improvement for the second pre-clinical study. Formation of foam could be reduced by exchanging Na CMC with PVP, as well as PS 20 by PS 80 and despite increasing the concentration of emulsifier from 0.02 to 0.5 %. All verum formulations of the second pre-clinical study could be suspended and injected. The placebo formulation still required longer reconstitution times (*Figure 7-2*).



Figure 7-2: Particle suspension in optimized reconstitution medium (PVP 3 %, PS 80 0.5 %, mannitol 5 %) prior to the application in the injection revolver (A). Injection of the suspension into the pig's lateral neck muscle (B). Marking of the injection site for further examination and documentation of adverse reactions (C)



7.4.3 **Biocompatibility and documentation of adverse drug reactions**

Figure 7-3: Adverse reactions documentated in first (A) and second pre-clinical study (B). All values are calculated as percentage of the treated animals (control: 5, black bars; treatment 20 animals, grey bars). Statistics were performed using a Mann-Whitney-Rank-Sum-test

The lipid microparticles were well tolerated, which went in accordance with observations from other researchers [45]–[47]. None of the gilts showed pain at the injection site in both control and treated groups. Throughout the first study, both control and treatment groups did not show signs of illness or fever. In the second study one gilt showed fever symptoms directly after injection and another after 24 d. Redness of the injection site could be detected in both studies, without significant difference between control and treatment groups, possibly reasoned by rapid injection without fixation of the animals. In both pre-clinical trials, more animals of the control groups showed redness at the injection sites in the second study, where a higher percentage of the control group showed increased temperature, potentially due to reconstitution problems.

Mild swelling was observed in all animals in both control and treatment groups during the first study, which could be minimized to 60 % due to better reconstitution properties. Two animals showed an intermediate swelling persisting for 4 d (treatment) and for 13 d (control). Isotonization with 5 % mannitol and replacement of Na CMC with PVP might be reasons for the better compatibility. In general, the treatment with G [6-D-Phe] did not show a significant increase in occurrence of adverse reactions compared to control.

7.4.4 Effects on cycle blockage in vivo using different formulations and G [6-D-Phe] concentrations

The sustained release G [6-D-Phe]-microparticles were evaluated with respect to minimal effective concentration as well as effect on cycle blockage and estrus control. Both placebo formulations of the first (D118) and second pre-clinical study (D114 + 5 % Span 40) did not influence the cycle (*Figure 7-4*).



Figure 7-4: Duration of cycle blockage after injection of lipid based microparticle formulations. Results of the first pre-clinical study using G [6-D-Phe] 750 μ g/animal and 3750 μ g/animal in D116/D118 formulations with 10 % GMS (A) as well as D114/D116 and G [6-D-Phe] 750 μ g/animal with addition of GMS and Span 40 tested in the second study (B). Data shown are individual results with mean and SEM. Statistical analysis was performed by one-way ANOVA followed by pairwise multiple comparisons (Holm-Sidak-method)

Using the formulation D116 + 10 % GMS 750 µg/animal, a cycle blockage of 10.0 ± 2.6 d could be achieved with one animal not responding to the therapy. A longer cycle blockage of 14.2 ± 4.9 d was observed in animals receiving the longer chain triglyceride D118 + 10 % GMS 750 µg/animal, with one animal not showing a cycle blockage in the observed time. Increasing the therapeutic dose to 3750μ g/animal resulted in a cycle blockage of 9.8 ± 4.3 d with two gilts not responding to the therapy when treated with D116 + 10 % GMS. The delivery of 3750μ g/animal G [6-D-Phe] formulated in D118 + 10 % GMS led to a cycle blockage of 15.8 ± 4.2 d in all animals. Due to high variation within the groups receiving the drug containing microparticles, no significant difference between placebo and treatment was observed. The

treatment with the higher dose of G [6-D-Phe] did not lead to a longer cycle blockage and did only in case of D118 + 10 % GMS result in a higher response within the group (5/5). In this group, one gilt showed a cycle blockage until the end of the observed time (31 d). In the treatment group with $3750 \mu g/animal$ (D116 + 10 % GMS) two animals did not show a response to drug application. Formulations tested in the first study evidenced, that the achievement of cycle blockage with the means of lipid based microparticles is generally possible using the low dose concentration, confirming the study with osmotic pumps [38]. Increasing the G [6-D-Phe] content did not lead to a significant improvement in therapy success or higher synchronicity. The highest synchronicity could be observed in the group of D116 + 10 % GMS 750 μ g/animal. Furthermore, the delivery of a higher dose seemed to have the consequence of a permanent cycle blockage, as it was only observed in one of the high-dose groups.

For the second pre-clinical study, four different formulations were tested delivering 750 µg/animal of G [6-D-Phe]. The use of D116 + 5 % Span 40 resulted in a cycle blockage of 4.6 ± 1.5 d, with all animals responding to the therapy but a high variability within responses. The effect of cycle blockage was significantly different compared to placebo (P-value: 0.017). The group treated with D114 + 5 % GMS showed a cycle blockage of 2.8 ± 1.4 d with one animal not showing a cycle blockage. Formulation consisting of D114 + 10 % Span 40 led to a cycle blockage of 1 day \pm 0.5 days with two animals not responding to the therapy. Best results throughout all studies could be achieved with the formulation D114 + 5 % Span 40, where all treated gilts showed a cycle blockage and the highest synchronicity could be observed (5/5). Hence, a cycle blockage of 6.4 ± 0.2 d was achieved with a significant difference to the control group (P-value < 0.001) and to the group treated with the formulation D114 + 10 % Span 40 (P-value: 0.005). Consequently, duration of cycle blockage was closest to the desired cycle blockage of 15 days using D116 + 10 % GMS 750 µg/animal, whereas highest synchronicity was observed when D114 + 5 % Span 40 was used. These findings confirmed that it was possible to achieve a cycle blockage with the use of a sustained release G [6-D-Phe] formulation.

7.4.5 **Influence on follicular growth**

Onset of follicular growth is another parameter indicating success of the therapy. The follicular growth started after 7 to 7.5 d in both control groups, which represented the regular duration time between diminishment of altrenogest and onset of follicular growth. After treatment with D116 + 10 % GMS 750 μ g/animal, follicles started to grow after 14.4 \pm 2.8 d. Exchanging the triglyceride to D118 + 10 % GMS 750 μ g/animal led to an onset of follicular growth after

 14.8 ± 3.1 d, leading to a permanent cycle blockage in one treated animal. An increase in G [6-D-Phe] dose to 3750 µg/animal resulted in onset of follicular growth after 14.2 ± 4.1 d (D116 + 10 % GMS) and 17 ± 2.0 d (D118 + 10 % GMS), where one animal did not show follicles after the monitored time. Due to considerably high variability within the groups, no significant difference between treatment and control groups could be detected, although follicular growth was delayed.



Figure 7-5: Onset of follicular growth after treatment with lipid microparticle formulations. (A) Results of the first pre-clinical study using of G [6-D-Phe] 750 μ g/animal and 3750 μ g/animal) in D116/D118 formulations with 10 % GMS. (B) Results of follow-up study using D114/D116 with addition of GMS and Span 40 and 750 μ g/animal G [6-D-Phe]. Data shown are individual results with mean and SEM. Statistical analysis was performed by Kruskal-Wallis-one-way ANOVA on ranks (A) and by one-way-ANOVA followed by pairwise multiple comparisons (Tukey-test) (B). * one animal with persistent cycle blockage observed

In the second study, treatment with D116 + 5 % Span 40 resulted in an onset of follicular growth after 11.6 \pm 1.5 d. After the treatment with D114 + 5 % GMS follicles occurred after 9 \pm 1.9 d. Most promising results were achieved with the use of D114 + 5 % Span 40 particles, where an onset after 13.4 \pm 0.2 d was monitored. This difference compared to the control group was significant (P-value < 0.05). Increasing the concentration of Span 40 to 10 % shortened the interval between discontinuation and onset to 7.8 \pm 0.6 d. Growing follicles were visible in all treated animals in this study. Thus, the formulation D114 + 5 % Span 40 showed in cycle blockage, synchronicity and onset of follicular growth the most promising results.

7.4.6 Influence of treatment on ovulatory behavior

Another important parameter for a successful reproduction is the occurrence of ovulation of growing follicles. Treatment with altrenogest is known to improve ovulation rates and following litter sizes compared to controls [10]. GnRH and its analogues are indicated in ovulation induction in human and veterinary medicine [48]–[50]. Therapy-associated ovulation within all groups was compared in order to investigate, if the treatment affected ovulation

behavior in gilts. Both control groups, D118 and D114 + 5 % Span 40 resulted in a 100 %ovulation observation within the ten tested animals. In the group treated with D116 + 10 % GMS 750 μ g/animal only 60 % of treated gilts showed an ovulation. This value was reduced to 40 %, when the formulation D118 + 10 % GMS 750 μ g/animal was used instead. A complete anovulatory behavior was observed in the treatment group consisting of D116 + 10 % GMS 3750 μ g/animal where none of the treated gilts showed ovulatory behavior. Here, a significant difference compared to the placebo group was monitored (P-value < 0.008). Since the other high dose group did also show a decreased ovulation behavior with statistical significance (20 % of the gilts in this group ovulated, P-value 0.043), the application of the high G [6-D-Phe] dose could be regarded as responsible for this effect.



Figure 7-6: Observed ovulation percentage after treatment with lipid-based microparticles in the first study. Each bar represents the percentage out of 5 animals per group. Statistical analysis was performed using the one-way ANOVA test followed by a Holm-Sidak pairwise comparison.

In the second study, treatment with D116 + 5 % Span 40 and D114 + 5 % GMS resulted in reduced ovulation percentage to 20 %. The formulation with best performance concerning cycle blockage and follicular growth, D114 + 5 % Span 40, led to an ovulation in 40 % of treated animals. The highest percentage regarding ovulation was observed with the use of D114 + 10 % Span 40 (80 %). In the second study, no statistical evidence of reduced ovulation compared to control was given. Nevertheless, a higher incidence of anovulatory behavior was observed upon treatment with sustained release G [6-D-Phe], whereas only the high dose treatment led to significantly reduced occurrence of ovulation. The possible connection between sustained release treatment with a GnRH-analogue and absence of ovulation and development of ovarian cysts needs to be clarified due to a possible disruption in down-regulation mechanism [51] [52].



Figure 7-7: Observed ovulation percentage after treatment with lipid-based microparticles in the second pre-clinical study. Each bar represents the percentage out of 5 animals per group. Statistical analysis was performed using the Kruskal-Wallis one-way ANOVA on ranks followed by a pairwise multiple comparison (Tukey-test)

7.4.7 Evaluation of the relation between treatment and the occurrence of temporary and permanent cysts

The occurrence of ovarian cysts is a critical point in estrus synchronization, as it has negative influence on performance of ovaries and success of insemination. Although GnRH analogues are used for the treatment of ovarian cysts [53] [54], therapy-induced formation of abnormal follicular structures is reported in literature in both human and veterinary medicine [51] [55] [56]. Polycystic ovarian degeneration (POD) is characterized by the occurrence of numerous liquid-filled cysts accompanied by an absence of *c. lutea* [40]. A higher incidence of POD was observed in estrus synchronization when lower doses (16 mg) of altrenogest were compared to higher doses (20 mg) [4]. Consequently, the parameter cyst formation was included in evaluation of treatment with a sustained release G [6-D-Phe] delivery system. In both control groups, none of the gilts showed permanent or temporary cysts (*Figure 7-8*). In contrast, treatment with D116/D118 + 10 % GMS 750 μ g/animal led to the development of permanent cysts in 40 % of treated gilts (*Figure 7-8 A black bars*). Increasing the dose to 3750 μ g/animal resulted in the development of permanent and temporary cysts in 20 % of treated gilts when treated with the formulation D118 + 10 % GMS 3750 μ g/animal.

The formulation D118 + 10 % GMS 3750 μ g/animal did not lead to the occurrence of cysts at all. During the second pre-clinical study, three treatment groups (D116 + 5 % Span 40, D114 + 5 % GMS, D114 + 5 % Span 40) showed an occurrence of permanent cysts in 40 % of the animals per group. A lower value of 20 % resulted with the formulation D114 + 10 % Span

40. Only the treatment with D116 + 5 % Span 40 microparticles resulted in the development of temporary cysts in 20 % of the animals.



Figure 7-8: Percentage of observed cysts (permanent: black bars, temporary: grey bars) during the treatment with G [6-D-Phe]-loaded lipid based microparticles monitored in the first pre-clinical study (A) and the follow-up study (B). Each bar represents the percentage out of the population of 5 animals per group. Statistical analysis was performed using the Kruskal-Wallis one-way ANOVA test on ranks

Kauffold et al. reported a very low incidence of the severe POD in gilts compared to sows upon estrus synchronization with gestagens and gonadotropins [40]. Besides the occurrence of single temporary or permanent cysts in our experiments, the severe POD did not occur. Although there was no statistical evidence, treatment with sustained release G [6-D-Phe] and the occurrence of cysts should be evaluated in a larger group of treated animals. The occurrence of follicular cysts is reported to be a severe disorder followed by infertility [40]. In literature, the treatment with LHRH-analogues is indicated in the presence of ovarian cysts in dairy cows also in combination with cloprostenol to treat these abnormal follicles [53] [57]. A possible explanation for this contradictory effect might be, that a treatment with gonadotropins possibly changes LH:FSH ratio, necessary for normal follicles as stated by Breen and Knox [58]. Also in human medicine, an increased development of follicular and luteal cysts was reported as complication during preparation for in vitro fertilization by the application of GnRH-analogues [55] [56] [59].

7.5 Conclusion

Our results showed that it was possible to achieve a cycle blockage with a sustained release formulation of G [6-D-Phe] in swine. Spray congealed solid lipid microparticles are a suitable

delivery system to influence and control the estrus. Particles were prepared without using organic solvents and were evaluated to be well tolerated and highly biocompatible in both in vivo studies.

Reconstitution and application of the microparticular suspension was difficult with the use of a Na CMC-based medium. Especially when high particle doses (3750 μ g/animal) needed to be administered, foam formation and attachment of particles to the liquid/air interface occurred. These drawbacks could be avoided using the optimized, PVP-based medium with higher emulsifier concentration, facilitating reconstitution and application.

The first study confirmed the successful estrus control with the use of 750 µg/animal G [6-D-Phe] resulting in a mean cycle blockage of 10 to 16 d with an onset of follicular growth after 15 to 17 d. These values came very close to the desired cycle blockage of 15 d. Here, a treatment with D116 in combination with 10 % GMS resulted in the highest synchronicity within the group. Compared to the placebo treatment, a significant reduction in ovulation was observed after the application of the higher dose (3750 µg/animal) and without statistical evidence after treatment with 750 µg/animal. Three out of four treated groups showed development of temporary and permanent follicular cysts without statistic relevance compared to controls.

Having proven our concept with the first pre-clinical study, **the second study** focused on the application of 750 μ g/animal via four different formulations to narrow the time window to the desired 15 d with a higher synchronicity. Best results were achieved with the formulation D114 + 5 % Span 40, where a duration of 6 d and the highest synchronicity in all trials could be observed. Follicular growth was setting in between 9 and 13 d after application. No significant differences in ovulation behavior between treatment and control was observed. All treatment groups showed development of ovarian cysts indicating a correlation between treatment with G [6-D-Phe] and the occurrence of these structures, without statistical evidence.

In summary, it is possible to achieve a cycle blockage in variable time frames with the application of G [6-D-Phe]-loaded microparticles by variating the type of emulsifier, as well as its concentration. Furthermore, we confirmed, which is economically favorable, the successful control of estrus cycle with the low dose G [6-D-Phe].

7.6 References

- L. L. Larson and P. J. H. Ball, "Regulation of estrous cycles in dairy cattle: A review," *Theriogenology*, vol. 38, no. 2. pp. 255–267, 1992.
- [2] K. P. Brussow, F. Schneider, W. Kanitz, J. Ratky, J. Kauffold, and M. Wahner, "Studies on fixed-time ovulation induction in the pig," *Soc. Reprod. Fertil. Suppl.*, vol. 66, pp. 187–195, 2009.
- [3] F. Martinat-Botte, F. Bariteau, B. Badouard, and M. Terqui, "Control of pig reproduction in a breeding programme," J. Reprod. Fertil. Suppl., vol. 33, pp. 211–228, 1985.
- [4] J. Kauffold, J. Beckjunker, A. Kanora, and W. Zaremba, "Synchronization of estrus and ovulation in sows not conceiving in a scheduled fixed-time insemination program," *Anim. Reprod. Sci.*, vol. 97, no. 1–2, pp. 84–93, 2007.
- [5] F. N. Kojima, B. E. Salfen, J. F. Bader, W. A. Ricke, M. C. Lucy, M. F. Smith, and D. J. Patterson, "Development of an estrus synchronization protocol for beef cattle with short-term feeding of melengestrol acetate: 7-11 synch," *J. Anim. Sci.*, vol. 78, no. 8, pp. 2186–2191, 2000.
- [6] L. N. Brown, K. G. Odde, M. E. King, D. G. LeFever, and C. J. Neubauer, "Comparison of Melengestrol Acetate-Prostaglandin F2-alpha to Syncro-Mate B for estrus synchronization in beef heifers," *Theriogenology*, vol. 30, no. 1, pp. 1–12, 1988.
- [7] K. N. Bretzlaff, L. C. Nuti, R. G. Elmore, S. A. Meyers, J. N. Rugila, S. P. Brinsko, T. L. Blanchard, and P. G. Weston, "Synchronization of estrus in dairy goats given norgestomet and estradiol valerate at various stages of the estrous cycle," *Am. J. Vet. Res.*, vol. 53, no. 6, pp. 930–934, 1992.
- [8] C. M. Wood, E. T. Kornegay, and C. F. Shipley, "Efficacy of altrenogest in synchronizing estrus in two swine breeding programs and effects on subsequent reproductive performance of sows," *J. Anim. Sci.*, vol. 70, no. 5, pp. 1357–1364, 1992.
- [9] J. S. Stevenson and D. L. Davis, "Estrous synchronization and fertility in gilts after 14- or 18-day feeding of altrenogest beginning at estrus or diestrus.," J. Anim. Sci., vol. 55, no. 1, pp. 119–123, 1982.
- [10] F. Martinat-Botté, F. Bariteau, Y. Forgerit, C. Macar, P. Poirier, and M. Terqui, "Synchronization of oestrus in gilts with altrenogest: effects on ovulation rate and foetal survival," *Anim. Reprod. Sci.*, vol. 39, no. 4, pp. 267–274, 1995.
- [11] D. A. Redmer and B. N. Day, "Estrus and ovulation in gilts fed a synthetic progestogen," *Theriogenology*, vol. 16, no. 2, pp. 195–199, 1981.
- [12] F. Martinat-Botte, F. Bariteau, Y. Forgerit, C. Macar, A. Moreau, M. Terqui, and J. P. Signoret, "Control of oestrus in gilts II. Synchronization of oestrus with a progestagen, altrenogest (Regumate): Effect on fertility and litter size," *Anim. Reprod. Sci.*, vol. 22, no. 3, pp. 227–233, 1990.
- T. Luukkainen, P. Lähteenmäki, and J. Toivonen, "Levonorgestrel-releasing intrauterine device.," *Ann. Med.*, vol. 22, no. 2, pp. 85–90, 1990.
- [14] W. Braendle, "Ein neues orales kontrazeptivum mit einem antimineralokortikoid und antiandrogen wirksamen gestagen (30 µg ethinylestradiol in kombination mit 3 mg drospirenon)," Geburtshilfe Frauenheilkd., vol. 61, no. 3, pp. 101–105, 2001.
- [15] E. F. Orlando and L. E. Ellestad, "Sources, concentrations, and exposure effects of environmental gestagens on fish and other aquatic wildlife, with an emphasis on reproduction," *Gen. Comp. Endocrinol.*, vol. 203, pp. 241–249, 2014.
- [16] J. E. Morthorst, H. Holbech, and P. Bjerregaard, "Trenbolone causes irreversible masculinization of

zebrafish at environmentally relevant concentrations," Aquat. Toxicol., vol. 98, no. 4, pp. 336-343, 2010.

- [17] J. Zeilinger, T. Steger-Hartmann, E. Maser, S. Goller, R. Vonk, and R. Länge, "Effects of synthetic gestagens on fish reproduction.," *Environ. Toxicol. Chem.*, vol. 28, no. 12, pp. 2663–2670, 2009.
- [18] C. Carlsson, A. K. Johansson, G. Alvan, K. Bergman, and T. Kühler, "Are pharmaceuticals potent environmental pollutants?. Part I: Environmental risk assessments of selected active pharmaceutical ingredients," *Sci. Total Environ.*, vol. 364, no. 1–3, pp. 67–87, 2006.
- [19] J. P. Webster, S. C. Kover, R. J. Bryson, T. Harter, D. S. Mansell, D. L. Sedlak, and E. P. Kolodziej, "Occurrence of trenbolone acetate metabolites in simulated confined animal feeding operation (CAFO) runoff," *Environ. Sci. Technol.*, vol. 46, no. 7, pp. 3803–3810, 2012.
- [20] J. P. Besse and J. Garric, "Progestagens for human use, exposure and hazard assessment for the aquatic environment," *Environ. Pollut.*, vol. 157, no. 12, pp. 3485–3494, 2009.
- [21] Z. A. Dequattro, E. J. Peissig, D. S. Antkiewicz, E. J. Lundgren, C. J. Hedman, J. D. C. Hemming, and T. P. Barry, "Effects of progesterone on reproduction and embryonic development in the fathead minnow (Pimephales promelas)," *Environ. Toxicol. Chem.*, vol. 31, no. 4, pp. 851–856, 2012.
- [22] S. M. Havens, C. J. Hedman, J. D. C. Hemming, M. G. Mieritz, M. M. Shafer, and J. J. Schauer, "Stability, preservation, and quantification of hormones and estrogenic and androgenic activities in surface water runoff," *Environ. Toxicol. Chem.*, vol. 29, no. 11, pp. 2481–2490, 2010.
- [23] S. L. Bartelt-Hunt, D. D. Snow, W. L. Kranz, T. L. Mader, C. A. Shapiro, S. J. Van Donk, D. P. Shelton, D. D. Tarkalson, and T. C. Zhang, "Effect of growth promotants on the occurrence of endogenous and synthetic steroid hormones on feedlot soils and in runoff from beef cattle feeding operations," *Environ. Sci. Technol.*, vol. 46, no. 3, pp. 1352–1360, 2012.
- [24] P. M. Conn, W. R. Huckle, W. V. Andrews, and C. A. MacArdle, "The Molecular Mechanism of Action of Gonadotropin Releasing Hormone (GnRH) in the pituitary," *Recent Prog. Horm. Res.*, pp. 29–61, 1987.
- [25] W. Zaremba, S. Engl, and A. Wehrend, "Gonavet®Veyx," 2015. [Online]. Available: https://www.veyx.de/uploads/tx_sitscontentattribute/KB_Gonavet_Veyx_DE_150421.pdf. [Accessed: 18-Oct-2016].
- [26] E. C. Griffiths and J. A. Kelly, "Mechanisms of Inactivation of Hypothalamic Regulatory Hormones," *Mol. Cell. Endocrinol.*, vol. 14, pp. 3–17, 1979.
- [27] E. C. Griffiths and J. R. McDermott, "Enzymic Inactivation of Hypothalamic Regulatory Hormones," *Mol. Cell. Endocrinol.*, vol. 33, no. 1, pp. 1–25, 1983.
- [28] A. M. Padula, "GnRH analogues—agonists and antagonists.," Anim. Reprod. Sci., vol. 88, pp. 115–126, 2005.
- [29] H. Berger, N. Heinrich, E. Albrecht, U. Kertscher, J. Oehlke, M. Bienert, H. Schafer, I. Baeger, and B. Mehlis, "Gonadotropin-releasing hormone (GnRH) analogs: relationship between their structure, proteolytic inactivation and pharmacokinetics in rats," *Regul. Pept.*, vol. 33, no. 3, pp. 299–311, 1991.
- [30] T. E. Trigg, P. J. Wright, A. F. Armour, P. E. Williamson, A. Junaidi, G. B. Martin, A. G. Doyle, and J. Walsh, "Use of a GnRH Analogue Implant to Produce Reversible Long-term Suppression of Reproductive Function in Male and Female Domestic Dogs.," *J. Reprod. Fertil. Suppl.*, vol. 57, pp. 255–261, 2001.
- [31] S. Goericke-Pesch, P. Georgiev, A. Antonov, M. Albouy, and A. Wehrend, "Clinical efficacy of a GnRHagonist implant containing 4.7 mg deslorelin, Suprelorin®, regarding suppression of reproductive function in tomcats," *Theriogenology*, vol. 75, no. 5, pp. 803–810, 2011.

- [32] J. G. W. Wenzel, K. S. S. Balaji, K. Koushik, C. Navarre, S. H. Duran, C. H. Rahe, and U. B. Kompella, "Pluronic F127 gel formulations of deslorelin and GnRH reduce drug degradation and sustain drug release and effect in cattle," *J. Control. Release*, vol. 85, no. 1–3, pp. 51–59, 2002.
- [33] T. Inaba, H. Tani, M. Gonda, A. Nakagawa, M. Ohmura, J. Mori, R. Torii, H. Tamada, and T. Sawada, "Induction of fertile estrus in bitches using a sustained-release formulation of a GnRH agonist (Leuprolide acetate)," *Theriogenology*, vol. 49, no. 5, pp. 975–982, 1998.
- [34] T. E. Trigg, J. D. Walsh, and D. A. Rathjen, "Sustained-release formulations comprising peptides, polypeptides, proteins or nucleic acids as active ingredients," CA2336879 C, 1999.
- [35] G. I. McRae, B. B. Roberts, A. C. Worden, A. Bajka, and B. H. Vickery, "Long-term reversible suppression of oestrus in bitches with nafarelin acetate, a potent LHRH agonist," *J. Reprod. Fertil.*, vol. 74, no. 2, pp. 389–397, 1985.
- [36] H. M. Fraser, J. Sandow, H. Seidel, and W. von Rechenberg, "An implant of a gonadotropin releasing hormone agonist (buserelin) which suppresses ovarian function in the macaque for 3-5 months," *Acta Endocrinol.*, vol. 115, no. 4, pp. 521–527, 1987.
- [37] W. Zaremba, J. Kauffold, W. Friess, K. Hoffmann, and Y. Yordanova, "Veterinary pharmaceutical composition and use thereof, WO2015EP58265 20150416," 2015.
- [38] K.-P. Brüssow, F. Schneider, and W. Kanitz, "Die Langzeitapplikation eines GnRH-Agonisten als eine alternative Methode zur Zyklussteuerung beim Schwein," 2000.
- [39] W. Zaremba, J. Kauffold, and W. Frieß, "Abschlussbericht: Förderinitiative Nachhaltige Pharmazie-Pilot-Entwicklung eines umweltfreundlichen Depotpräparates zur Zyklussteuerung," 2016.
- [40] J. Kauffold, T. Rautenberg, S. Gutjahr, A. Richter, and A. Sobiraj, "Ultrasonographic characterization of the ovaries in non-pregnant first served sows and gilts," *Theriogenology*, vol. 61, no. 7–8, pp. 1407–1417, 2004.
- [41] N. Soede, W. Hazeleger, and B. Kemp, "Follicle size and the process of ovulation in sows as studied with Ultrasound," *Reprod. Domest. Anim.*, vol. 33, no. 3–4, pp. 239–244, 1998.
- [42] S. Comba and R. Sethi, "Stabilization of highly concentrated suspensions of iron nanoparticles using shearthinning gels of xanthan gum," *Water Res.*, vol. 43, no. 15, pp. 3717–3726, 2009.
- [43] H. a. Barnes, "Shear-Thickening ('Dilatancy') in Suspensions of Nonaggregating Solid Particles Dispersed in Newtonian Liquids," *J. Rheol. (N. Y. N. Y).*, vol. 33, no. 2, p. 329, 1999.
- [44] A. Benchabane and K. Bekkour, "Rheological properties of carboxymethyl cellulose (CMC) solutions," *Colloid Polym. Sci.*, vol. 286, no. 10, pp. 1173–1180, 2008.
- [45] C. Guse, S. Koennings, A. Maschke, M. Hacker, C. Becker, S. Schreiner, T. Blunk, T. Spruss, and A. Goepferich, "Biocompatibility and erosion behavior of implants made of triglycerides and blends with cholesterol and phospholipids," *Int. J. Pharm.*, vol. 314, no. 2, pp. 153–160, 2006.
- [46] S. Koennings, A. Sapin, T. Blunk, P. Menei, and A. Goepferich, "Towards controlled release of BDNF -Manufacturing strategies for protein-loaded lipid implants and biocompatibility evaluation in the brain," *J. Control. Release*, vol. 119, no. 2, pp. 163–172, 2007.
- [47] H. Reithmeier, J. Herrmann, and A. Gopferich, "Lipid microparticles as a parenteral controlled release device for peptides," J. Control. Release, vol. 73, no. 2–3, pp. 339–350, 2001.
- [48] K. P. Brussow, W. Jochle, and U. Huhn, "Control of ovulation with a GnRH analog in gilts and sows," *Theriogenology*, vol. 46, no. 6, pp. 925–934, 1996.

- [49] F. von Kaufmann and W. Holtz, "Induction of ovulation in gonadotropin treated gilts with synthetic gonadotropin releasing hormone," *Theriogenology*, vol. 17, no. 2, pp. 141–157, 1982.
- [50] G. Bartfai, "Clinical applications of gonadotrophin-releasing hormone and its analogues," *Hum. Reprod.*, vol. 3, no. 1, pp. 51–57, 1988.
- [51] K. Wongkaweewit, P. Prommachart, R. Raksasub, K. Buranaamnuay, M. Techakumphu, F. De Rensis, and P. Tummaruk, "Effect of the administration of GnRH or hCG on time of ovulation and the onset of estrus-to-ovulation interval in sows in Thailand," *Trop. Anim. Health Prod.*, vol. 44, no. 3, pp. 467–470, 2012.
- [52] M. A. Driancourt, P. Cox, S. Rubion, G. Harnois-Milon, B. Kemp, and N. M. Soede, "Induction of an LH surge and ovulation by buserelin (as Receptal) allows breeding of weaned sows with a single fixed-time insemination," *Theriogenology*, vol. 80, no. 4, pp. 391–399, 2013.
- [53] T. C. Cantley, H. A. Garverick, C. J. Bierschwal, C. E. Martin, and R. S. Youngquist, "Hormonal responses of dairy cows with ovarian cysts to GnRH.," *J. Anim. Sci.*, vol. 41, no. 6, pp. 1666–1673, 1975.
- [54] A. R. Peters, "Veterinary clinical application of GnRH Questions of efficacy," *Anim. Reprod. Sci.*, vol. 88, no. 1–2 SPEC. ISS., pp. 155–167, 2005.
- [55] Z. Ben-Rafael, D. Bider, Y. Menashe, R. Maymon, M. Zolti, and S. Mashiach, "Follicular and luteal cysts after treatment with gonadotropin-releasing hormone analog for in vitro fertilization.," *Fertil. Steril.*, vol. 53, no. 6, pp. 1091–4, 1990.
- [56] H. S. Qublan, Z. Amarin, Y. A. Tahat, A. Z. Smadi, and M. Kilani, "Ovarian cyst formation following GnRH agonist administration in IVF cycles: Incidence and impact," *Hum. Reprod.*, vol. 21, no. 3, pp. 640– 644, 2006.
- [57] F. López-Gatius and M. López-Béjar, "Reproductive performance of dairy cows with ovarian cysts after different GnRH and cloprostenol treatments.," *Theriogenology*, vol. 58, no. 7, pp. 1337–48, 2002.
- [58] S. M. Breen and R. V. Knox, "The impact of dose of FSH (Folltropin) containing LH (Lutropin) on follicular development, estrus and ovulation responses in prepubertal gilts," *Anim. Reprod. Sci.*, vol. 132, no. 3–4, pp. 193–200, 2012.
- [59] D. Feldberg, J. Ashkenazi, D. Dicker, A. Yeshaya, G. A. Goldman, and J. A. Goldman, "Ovarian cyst formation: a complication of gonadotropin-releasing hormone agonist therapy.," *Fertil. Steril.*, vol. 51, no. 1, pp. 42–5, 1989.

8 LONG-TERM STABILITY STUDY OF G [6-D-PHE]-LOADED MICROPARTICLES

Parts of this chapter are intended for publication

HPLC analysis for determination of drug content was performed by *BFC-Feinchemie*, Weinbergen Grabe.

8.1 Abstract

The lead candidate from the first clinical study was investigated concerning long-term stability. Microparticles were stored at 2-8, 25 and 40 °C and samples were taken after 0, 1, 3, 6, 9 and 12 months. Analysis of particle size, reconstitution properties, lipid polymorph, peptide content and release behavior was performed. Formulations showed particle agglomeration upon storage, which was minor at 2-8 °C and most pronounced when stored at 40 °C. Reconstitution to a homogenous suspension was possible with samples stored at 2-8 °C and 25 °C but limited after 40 °C storage. Drug content was not affected in any sample by storage even after 12 months at 40 °C and no degradation products were detected. The initially observed release over two weeks was preserved when microparticles were stored at 2-8 °C. Storage at higher temperatures led to a higher burst and a more complete release. Thermal analysis revealed the transformation of the initially obtained α -polymorph to the thermodynamically more stable β -form after storage at 40 °C. Upon storage at 2-8 °C, the α -modification was maintained over 12 months, whereas XRPD of samples stored at 25 °C already indicated ongoing transitions. Thus, storage at 2-8 °C was highly recommended to assure stability with a reliable release profile accompanied by good reconstitution behavior and peptide integrity.

8.2 Introduction

Lipid microparticles are a promising tool for the sustained release of peptides and proteins [1]– [5]. We developed microparticles loaded with G [6-D-Phe] which showed drug release over two weeks in vitro as well as efficacy in vivo [6]. Consequently, the stability of these microparticles was tested at different temperatures over 12 months. Ideally, the product should be stable at ambient temperature to enable easy handling by veterinarians and farmers. It should further guarantee a continuous release in a two-week time window. Furthermore, particle size and lipid polymorph must be preserved [7] [8]. Another major aspect is the integrity of the incorporated peptide.

CHAPTER EIGHT

8.3 Materials and Methods

8.3.1 Materials

Gonadorelin [6-D-Phe] (G [6-D-Phe]) was donated by Veyx Pharma GmbH (Schwarzenborn, Germany). The triglyceride Dynasan 118 (tristearin, D118) was kindly provided from Cremer Oleo (Witten, Germany). Glycerol monostearate (GMS) with a monoester content of 40-55 % was purchased from Caelo (Caesar & Loretz, Hilden Germany). Ultrapure deionized water with a conductivity of less than 0.055 μ S/cm (Milli-Q Water systems, Millipore, MA, USA) was used for preparation of buffers and mobile phases in this study. Visking dialysis tubings with a diameter of 1.6 mm and a cut-off of 12-14 kDa for performance of release studies were sourced from Serva Electrophoresis GmbH (Heidelberg, Germany). Centrifuge tubes were purchased from VWR International GmbH (Darmstadt, Germany). The reconstitution medium consisted of 1 % Tylopur[®] C 300 (sodium carboxymethylcellulose, Na CMC, Clariant, Muttenz, Switzerland) and 0.02 % polysorbate 20 (PS 20, Merck KGaA, Darmstadt, Germany) in distilled water (Wagner und Munz, München, Germany). All other chemicals were of analytical grade.

8.3.2 Production of Gonadorelin [6-D-Phe]-loaded microparticles

Lyophilized G [6-D-Phe] was milled using a Retsch[®] Cryomill (Retsch Technology, Haan, Germany) using two stainless steel balls (10 mm diameter) with a precooling time of 10 min at 5 Hz followed by the actual milling step of 4 min at 20 Hz. The obtained powder was aliquoted under nitrogen atmosphere. 10.6 g of D118 and 1.2 g of GMS were melted on a hot plate at approximately 90 °C. 200 mg of peptide was added and homogenized using a T-10 basic Ultraturrax (IKA Laboratory Technology, Staufen, Germany) for 2 min. The lipid dispersion was poured into the sample container of a B-290 Mini spray dryer (Büchi, Flawil, Switzerland) with additional spray congealing equipment, which was pre-conditioned for 1.5 h (sample container 110 °C, inlet temperature 13-15 °C) prior to the production step. The additional equipment consisted of a sample container, surrounded by a PEG 400 bath, whereas the sample container was connected to the nozzle via a silicon tubing heated with an external resistance wire-based heating to approximately 70 °C. The melt was sprayed using nitrogen as spray gas at 6 bar spray pressure resulting in a spray flow of 414 l/h and with aspirator power set at 100 %.

8.3.3 Storage conditions and study set-up

For each sampling point, samples of approximately 350 mg were stored at 2-8, 25 and 40 °C in two glass vials. Microparticles were weighed under LAF conditions using a HeraSafe HSP 12 (Heraeus, Hanau, Germany), closed with rubber stoppers and crimped. Analysis was performed after 0, 1, 3, 6, 9 and 12 months.

8.3.4 Macroscopic inspection of the microparticles

Microparticles were visually inspected concerning appearance and aggregation behavior. Documentation was performed using a Nikon D5300 (Nikon, Tokyo, Japan) equipped with a macro-objective.

8.3.5 Observation of surface morphology by scanning electron microscopy (SEM)

A Jeol JSM 6500-F SEM (Jeol, Tokyo, Japan) was used for SEM images at a voltage of 2 kV using a magnification of 150, 900 and 1500 for each sample. Microparticles were fixed on aluminum sample holders with black, double adhesive tape (Plano GmbH, Wetzlar, Germany) and were analyzed without additional graphite coating.

8.3.6 Stability of the suspension

Approximately 200 mg of stored microparticles were shortly shaken in a glass vial and 2 ml of reconstitution medium were added via a syringe through the rubber stopper. The dispersion was vigorously shaken by hand for approximately 20 s to suspend the particles in the surrounding medium. Photographic images were taken immediately after reconstitution and after standing times of 2, 5, and 10 min.

8.3.7 Particle size analysis using laser diffraction

Approximately 20 mg of microparticles were suspended in an aqueous 0.02 % PS 20 solution and analyzed using the laser diffraction system Horiba Partica-LA 950 (Horiba, Kyoto, Japan).

8.3.8 Determination of polymorphic behavior by differential scanning calorimetry (DSC)

Samples of 5-15 mg microparticles in aluminum crucibles were analyzed using a Mettler DSC 821e (Mettler Toledo, Columbus, OH, USA) at a heating and cooling rate of 10 K/min from 0 $^{\circ}$ C to 110 $^{\circ}$ C in two cycles against an empty crucible as reference.

8.3.9 X-Ray powder diffraction (XRPD) measurements

A PANalytical Empyrean (PANalytical, Almelo, the Netherlands) equipped with a copper anode (45 kV, 40 mA, K α 1 emission, $\lambda = 0.154$ nm) and a PIXcel3D detector between 5-50 ° 2-Theta with a step of 0.039 ° was used for XRPD measurements.

8.3.10 Drug content analyzed via RP-HPLC

Approximately 10 mg microparticles were weighed into centrifuge tubes and dissolved in 2.0 ml of methylene chloride. 2.0 ml of HPLC gradient grade water (Fisher Chemical, Loughborough, UK) were added and the tubes were placed in a Stuart tube rotator SB 3 at a

rotation speed of 10 rpm (Bibby Scientific Limited, Stone, UK) at ambient temperature for 45 min. Samples of 1 ml were taken from the water phase and analyzed using RP-HPLC. Each stored microparticle sample was extracted in duplicate. RP-HPLC analysis was performed using a Varian RP-HPLC system (Varian Inc., Palo Alto, CA, USA), supplied with a Luna C-8-column (Phenomenex, Aschaffenburg, Germany) and the guard column SecurityGuard (Phenomenex, Aschaffenburg, Germany). Columns were maintained at 40 °C and using the following gradient:

Time [min]	A [%]	B [%]
0	65	35
1	65	35
21	65	35
22	10	90
26	10	90
35	65	35

Mobile Phase A: 1000 ml highly purified water + 1 ml trifluoracetic acid

Mobile Phase B: 800 g Acetonitrile + 200 g water + 1.2 ml trifluoracetic acid

An injection volume of 20 µl was used for each sample and calibration standards. The actual drug content was normalized to a theoretical initial weight of 10.0 mg. This analysis was performed by Biopept-Feinchemie GmbH, Weinbergen/Grabe.

8.3.11 Determination of release behavior

50 mg of microparticles were transferred into dialysis tubings, which were previously closed at one side with a cotton string. 2.0 ml phosphate-buffered saline (PBS) pH 7.4 preserved with 0.05 % NaN₃ and five glass balls (0.2 mm diameter) for additional weight were added to suspend the microparticles. The closed tubings were placed in 50 ml-centrifuge tubes, 15 ml PBS were added and the samples incubated on a horizontal shaking incubator GFL 3031 (Gesellschaft für Labortechnik, Burgwedel, Germany) at 39 °C and 60 rpm. At predetermined time points, samples of 1 ml volume were taken, replaced by fresh PBS and stored at 2-8 °C until HPLC analysis was performed. An Agilent HPLC system (Agilent, Santa Clara, CA, USA) was used using the setup mentioned in *8.3.10*.

8.4 Results and Discussion

8.4.1 Influence of storage time and -conditions on particle morphology, -size and polymorphic behavior

G [6-D-Phe]-containing, freshly prepared microparticles appeared as fine powder without agglomerates (*Figure 8-1*). Manual shaking of the vial did not lead to adherence of the particles to the chamber wall, which indicated little electrostatic charging, as frequently observed for spray dried products [9] [10]. SEM images revealed a smooth and regular surface [11]–[13]. The smooth surface obtained after production of lipid microparticles by spray congealing frequently comes with the presence of metastable polymorphs [13]–[16]. The α -form with the lowest melting point obtains very small crystals and a hexagonal orientation of the carbon chains, which enable less chain-chain-interaction and a loose packaging [17]–[19]. The α -polymorph was confirmed by DSC and XRPD (*see 8.4.1*).

0 M





After storage time of one month at 2-8 °C, particles still appeared as a fine powder. Reconstitution could be performed easily in the reconstitution medium resulting in a stable suspension over the monitored time of 10 min. Neither irregularities nor an increased roughness were observed on the microsphere's surface. The 25 °C stored samples showed a tendency towards agglomeration (*Figure 8-2*). Upon shaking, the larger agglomerates could be easily separated and reconstitution could be performed without extending the reconstitution time. Samples stored at 40 °C formed a sintered cake already after one month. Complete disintegration could not be achieved by vigorous shaking after adding the suspension medium. The formation of larger agglomerates was also observed in the SEM, where non-spherical, large particles with a rougher and more irregular surface were monitored.

For this reason, analysis after reconstitution was stopped after one month and only polymorph and peptide analysis were performed at the later time points.

The mean particle size of 25 μ m at T₀ was increased to 50 μ m after one month at 2-8 °C, 60 μ m at 25 °C and 58 μ m at 40 °C (*Figure 8-4*). Samples analyzed at 40 °C were not representative 146

since the large sintered clumps were not removed from the sample prior to analysis. In the samples stored at 2-8 °C and 25 °C no increase in particle size can be found in SEM. Thus, the increased particle size obtained in laser diffraction can be referred to larger agglomerates.



Figure 8-2: Appearance of microparticles after 1 M storage at 2-8, 25 and 40 °C

CHAPTER EIGHT



Figure 8-3: Appearance of microparticles after 3 M storage at 2-8, 25 and 40 °C

Further storage for three months at 2-8 °C did not affect the shape and morphology of particles (*Figure 8-3*). Particles still showed a smooth surface without irregularities. Particle size was similar to the one-month time point. Agglomerates could be seen macroscopically after three months at 25 °C which could easily be separated by manual shaking. SEM images revealed the formation of larger particle agglomerates. A closer look on the surface showed, that the particles exhibited a rougher surface indicating lipid crystallization [13] [18]. This effect was even more pronounced in the samples stored at 40 °C with large sintered agglomerates in SEM and flaky crystal structures on the surface. This "blooming" and growth of larger flake- or needle-like structures corresponds to the conversion into the more stable β -polymorph (see *Figure 8-7* and *Figure 8-8*) [18] [20]. Laser diffraction demonstrated further increase in particle size to 90 µm after three months at 25 °C (*Figure 8-4*).



Figure 8-4: Mean particle size measurement using laser diffraction over a 12-month storage time at 2-8, 25 and 40 °C (Mean and SD, n = 3, left); Reconstitution of 2-8 °C-stored samples after 6 months exemplarily for acceptable reconstitution properties (right)

Comparable results were obtained for samples stored for six and nine months. At 2-8 °C particles kept their shape without changes in surface morphology, whereas particles stored at 25 and 40 °C showed a high tendency to agglomeration. *Figure 8-4 (right)* shows exemplarily the suspension of reconstituted microparticles stored at 2-8 °C in the medium, giving evidence to good reconstitution properties. At 40 °C, the microparticles had formed a dense cake which needed to be crushed prior to SEM analysis and could not be analyzed providing meaningful data. Laser diffraction results were comparable to three-month data.

CHAPTER EIGHT



Figure 8-5: Appearance of microparticles after 6 M storage at 2-8, 25 and 40 °C

After twelve months, no further changes were observed for microparticles stored at 2-8 °C (*Figure 8-6*). Compared to previous time points the results after twelve months did not significantly change regarding particle appearance, morphology, size and agglomeration. Overall, storage temperature had a major influence on particle agglomeration and surface roughness. SEM and particle size measurements indicated crystallization phenomena as well as agglomeration in the samples stored at 25 and 40 °C [21]. Samples stored at 2-8 °C only showed a slight tendency to formation of agglomerates.

CHAPTER EIGHT



Figure 8-6: Appearance of microparticles after 12 M storage at 2-8, 25 and 40 °C

Polymorphic behavior of lipids has been discussed as a stability problem [22]–[26]. To investigate polymorphic transitions, DSC-thermograms were recorded, which proved the presence of the lower melting α -polymorph within the particles directly after production (*Figure 8-7*). A melting endotherm of 55 °C indicates the α -form, whereas the β' - and β - modification melt at higher temperatures of approx. 63 and 73 °C [27] [28]. The polymorph did not change over a twelve-month storage at 2-8 °C. This observation was in good agreement with literature that the α -polymorph is stable at low storage temperatures [29]. As polymorphism or recrystallization are known to influence release kinetics, a conservation of the initially observed polymorph is important [8] [30] [31]. At 25 °C, same effects are visible. Upon storage at 40 °C, formation of the higher melting β -polymorph was already completed at least after one month. This observed effect corresponds to the transition obtained in *Chapter two* for microparticles and for tempering of triglyceride-based implants to form the stable β -polymorph as described by Kreye et al. [8]. Whereas addition of emulsifiers like sorbitan esters may stabilize the α -polymorph, others are reported to enhance α - β transition [24] [25] [32]–[34].



Figure 8-7: Representative DSC thermograms of microparticles stored at (A) 2-8 °C; (B) 25 °C; (C) 40 °C (n = 2); sampling was performed after 0, 1, 3, 6, 9 and 12 months

These findings were confirmed by XRPD measurements of the 12 M sample (*Figure 8-8*). After one year at 2-8 °C microparticles were present in the α -polymorph (one single peak at 21 ° characteristic for the α -modification) [19] [35]. Storage at 25 °C resulted in the presence of both α - and β -polymorph with a more α -polymorph pronounced pattern [19] [15]. In contrast, the sample stored at 40 °C showed the characteristic pattern of the β -modification with three peaks at 18, 23 and 25 ° [19]. These observations confirmed the data obtained by DSC measurements and underline that a storage at 2-8 °C is mandatory in terms of crystallization of the lipid matrix. Overall the storage effects are typical for triglycerides, whereas the presence of additives [23] [36] [37] and drug types influence crystallization behavior [18]. Due to rapid cooling during the spray congealing process, the crystallization of the triglyceride D118 in the lower melting α -polymorph is enabled. Several tempering steps, additionally performed directly after production, are reported for implants, to allow an accelerated crystallization to the β -polymorph [8] [38].



Figure 8-8: Representative XRPD measurement of the 12 M-stored samples at 2-8, 25 and 40 °C. Single graphs were displaced along the ordinate for better visualization (n = 2)

8.4.2 **Drug content and release behavior as a function of storage time and temperature** The drug content in the lipid matrix did not change significantly over one year at 2-8, 25 and 40 °C (*Figure 8-9*). Stability was even reported for proteins, like BSA directly after encapsulation into lipid microparticles produced by spray congealing [5]. The minor variability in drug content was due to the used extraction method based on methylene chloride. Furthermore, no degradation peaks were detected in HPLC. Encapsulated G [6-D-Phe] was not affected by higher storage temperature and the process itself, as reported for other peptide drugs [13].



Figure 8-9: G [6-D-Phe] content in D118 + 10 % GMS microparticles during a 12-month storage time at 2-8, 25 and 40 $^{\circ}$ C (mean and SD, n = 3)

The most important characteristic for the developed drug delivery system was to guarantee the required release profile. Particle size and morphology, as well as an integer matrix with the same polymorphic modification have a major influence on release behavior.

Freshly prepared microparticles showed a burst of approximately 45 % followed by a continuous release until day 21 releasing 70 % of incorporated drug. Regarding a storage at 2-8 °C, the release profile did not significantly change over a one year storage.



Figure 8-10: Release profile of G [6-D-Phe] microparticles consisting of D118 + 10 % GMS after a storage at 2-8 °C (mean and SD, n = 3)

Monitoring release behavior after a one year storage at 25 °C, the profile differed considerably from the original formulation. Already after one month, the burst was remarkably increased without a sustained drug release following. Enhanced release profiles could also be observed for triglyceride implants after thermal treatment and applying higher extrusion temperatures [8] [31].

Higher storage temperatures are reported to massively influence release rates from lipid-based microspheres, due to change in microstructure of the lipid matrix [39] as well as the polymorph [18]. After one month at 40 °C, the initial burst increased to 80 % without following sustained release. Overall, an amount of 95 % was released. The effect became more pronounced over one year storage. This change in release profile may be attributed to the polymorph change and the sintering of the microparticles corresponding to change in the surface area and the inner structure [30] [34] [40].



Figure 8-11: Release profile of G [6-D-Phe]-loaded microparticles consisting of D118 + 10 % GMS after a storage at 25 °C (mean and SD, n = 3)



Figure 8-12: Release profile of G [6-D-Phe]-loaded microparticles consisting of D118 + 10 % GMS after a storage at 40 °C (mean and SD, n = 3)

Thus, the G-[6-D-Phe]-loaded D118/GMS microparticles need to be stored at 2-8 °C to assure the preferred release profile with a continuous in vitro release of two weeks. Peptide stability was not impaired by storage at higher temperature. Furthermore, particle agglomeration could be minimized at 2-8 °C and the initial polymorphic structure could be kept upon further storage. In contrast, storage at 25 and 40 °C did lead to transition to the thermodynamically stable β polymorph and particle sintering.

8.5 Conclusion

G [6-D-Phe] in D118/GMS microparticles were stored over one year at 2-8, 25 and 40 °C and analyzed for particle morphology, size, agglomeration, lipid polymorph, drug content and resulting release behavior. Storage at 2-8 °C was mandatory to ensure minimal particle agglomeration and to guarantee the presence of the initially observed α -polymorph. Polymorphic transition, accompanied with blooming on the surface occurred upon storage at 25 and 40 °C: Release behavior could only be maintained at a storage temperature of 2-8 °C. Furthermore, reconstitution behavior was not affected upon one year storage at 2-8 °C. The study confirmed the stability of G [6-D-Phe] after encapsulation into lipid-based microparticles at all storage temperatures without peptide degradation.

8.6 References

- H. Reithmeier, J. Herrmann, and A. Gopferich, "Lipid microparticles as a parenteral controlled release device for peptides," *J. Control. Release*, vol. 73, no. 2–3, pp. 339–350, 2001.
- [2] H. Reithmeier, J. Herrmann, and A. Gopferich, "Development and characterization of lipid microparticles as a drug carrier for somatostatin," *Int. J. Pharm.*, vol. 218, no. 1–2, pp. 133–143, 2001.
- [3] W. Vogelhuber, E. Magni, M. Mouro, T. Spruss, C. Guse, A. Gazzaniga, and A. Gopferich, "Monolithic triglyceride matrices: a controlled-release system for proteins," *Pharm. Dev. Technol.*, vol. 8, no. 1, pp. 71–79, 2003.
- M. D. Del Curto, D. Chicco, M. D'Antonio, V. Ciolli, H. Dannan, S. D'Urso, B. Neuteboom, S. Pompili,
 S. Schiesaro, and P. Esposito, "Lipid microparticles as sustained release system for a GnRH antagonist (Antide)," *J. Control. Release*, vol. 89, no. 2, pp. 297–310, 2003.
- [5] M. Di Sabatino, B. Albertini, V. L. Kett, and N. Passerini, "Spray congealed lipid microparticles with high protein loading: preparation and solid state characterisation," *Eur. J. Pharm. Sci.*, vol. 46, no. 5, pp. 346– 356, 2012.
- [6] W. Zaremba, J. Kauffold, W. Friess, K. Hoffmann, and Y. Yordanova, "Veterinary pharmaceutical composition and use thereof, WO2015EP58265 20150416," 2015.
- [7] S. Jaspart, G. Piel, L. Delattre, and B. Evrard, "Solid lipid microparticles: formulation, preparation, characterisation, drug release and applications," *Expert Opin. Drug Deliv.*, vol. 2, no. 1, pp. 75–87, 2005.
- [8] F. Kreye, F. Siepmann, A. Zimmer, J. F. Willart, M. Descamps, and J. Siepmann, "Cast lipid implants for controlled drug delivery: importance of the tempering conditions," *J. Pharm. Sci.*, vol. 100, no. 8, pp. 3471–3481, 2011.
- P. He, S. S. Davis, and L. Illum, "Chitosan microspheres prepared by spray drying.," *Int. J. Pharm.*, vol. 187, no. 1, pp. 53–65, 1999.
- [10] B. B. Patel, J. K. Patel, S. Chakraborty, and D. Shukla, "Revealing facts behind spray dried solid dispersion technology used for solubility enhancement," *Saudi Pharm. J.*, vol. 23, no. 4, pp. 352–365, 2015.
- [11] S. Scalia, D. Traini, P. M. Young, M. Di Sabatino, N. Passerini, and B. Albertini, "Comparison of spray congealing and melt emulsification methods for the incorporation of the water-soluble salbutamol sulphate in lipid microparticles," *Pharm. Dev. Technol.*, vol. 18, no. 1, pp. 266–273, 2013.
- [12] C. Cavallari, B. Luppi, A. M. Di Pietra, L. Rodriguez, and A. Fini, "Enhanced release of indomethacin from Pvp/stearic acid microcapsules prepared coupling Co-freeze-drying and ultrasound assisted spraycongealing process," *Pharm. Res.*, vol. 24, no. 3, pp. 521–529, 2007.
- [13] A. Maschke, C. Becker, D. Eyrich, J. Kiermaier, T. Blunk, and A. Gopferich, "Development of a spray congealing process for the preparation of insulin-loaded lipid microparticles and characterization thereof," *Eur. J. Pharm. Biopharm.*, vol. 65, no. 2, pp. 175–187, 2007.
- [14] K. M. Rosenblatt and H. Bunjes, "Poly(vinyl alcohol) as emulsifier stabilizes solid triglyceride drug carrier nanoparticles in the alpha-modification," *Mol. Pharm.*, vol. 6, no. 1, pp. 105–120, 2009.
- [15] J. H. Oh, A. R. McCurdy, S. Clark, and B. G. Swanson, "Characterization and thermal stability of polymorphic forms of synthesized tristearin," *J. Food Sci.*, vol. 67, no. 8, pp. 2911–2917, 2002.
- [16] I. Ilic, R. Dreu, M. Burjak, M. Homar, J. Kerc, and S. Srcic, "Microparticle size control and glimepiride microencapsulation using spray congealing technology," *Int. J. Pharm.*, vol. 381, no. 2, pp. 176–183, 2009.
- [17] D. Kodali, D. Atkinson, and T. Rergrave, "Synthesis and Polymorphism of 1,2-Dipalmitoyl-3-acyl-sn-

glycerols," J. Am. Oil Chem. Soc., vol. 61, no. 6, pp. 1078-1084, 1984.

- [18] T. Eldem, P. Speiser, and H. Altorfer, "Polymorphic behavior of sprayed lipid micropellets and its evaluation by differential scanning calorimetry and scanning electron microscopy," *Eur. J. Pharm. Biopharm.*, vol. 8, no. 2, pp. 178–184, 1991.
- [19] M. Windbergs, C. J. Strachan, and P. Kleinebudde, "Investigating the Principles of Recrystallization from Glyceride Melts," AAPS PharmSciTech, vol. 10, no. 4, pp. 1224–1233, 2009.
- [20] H. Mayama, "Blooming theory of tristearin," Soft Matter, vol. 5, no. 4, pp. 856–859, 2009.
- [21] D. G. Lopes, K. Becker, M. Stehr, D. Lochmann, D. Haack, A. Zimmer, and S. Salar-Behzadi, "Role of Lipid Blooming and Crystallite Size in the Performance of Highly Soluble Drug-Loaded Microcapsules," *J. Pharm. Sci.*, vol. 104, no. 12, 2015.
- [22] T. Malkin, "The polymorphism of glycerides," in *Progress in the Chemistry of Fats and other Lipids*, vol. 1, 1954, pp. 1–17.
- [23] K. Sato and T. Kuroda, "Kinetics of melt crystallization and transformation of tripalmitin polymorphs," J. Am. Oil Chem. Soc., vol. 64, no. 1, pp. 124–127, 1987.
- [24] J. Schlichter Aronhime, S. Sarig, and N. Garti, "Dynamic control of polymorphic transformation in triglycerides by surfactants: The button syndrome," J. Am. Oil Chem. Soc., vol. 65, no. 7, pp. 1144–1150, 2014.
- [25] J. Schlichter Aronhime, S. Sarig, and N. Garti, "Mechanistic Considerations of Polymorphic Transformations of Tristearin in the Presence of Emulsifiers," J. Am. Oil Chem. Soc., vol. 64, no. 4, pp. 529–533, 1987.
- [26] M. Kellens, W. Meeussen, C. Riekel, and H. Reynaers, "Time resolved x-ray diffraction studies of the polymorphic behaviour of tripalmitin using synchrotron radiation," *Chem. Phys. Lipids*, vol. 52, no. 2, pp. 79–98, 1990.
- [27] E. Hagemann, J. W., Tallent, W. H., Kolb, "Differential scanning calorimetry of single acid triglycerides:
 Effect of chain length and unsaturation," *J. Am. Oil Chem. Soc.*, vol. 49, no. 2, pp. 118–123, 1972.
- [28] P. Elisabettini, A. Desmedt, and F. Durant, "Polymorphism of stabilized and nonstabilized tristearin, pure and in the presence of food emulsifiers," J. Am. Oil Chem. Soc., vol. 73, no. 2, pp. 187–192, 1996.
- [29] B. Albertini, N. Passerini, M. L. Gonzalez-Rodriguez, B. Perissutti, and L. Rodriguez, "Effect of Aerosil on the properties of lipid controlled release microparticles," *J. Control. Release*, vol. 100, no. 2, pp. 233– 246, 2004.
- [30] N. Khan and D. Q. M. Craig, "Role of blooming in determining the storage stability of lipid-based dosage forms," J. Pharm. Sci., vol. 93, no. 12, pp. 2962–2971, 2004.
- [31] C. Reitz and P. Kleinebudde, "Solid lipid extrusion of sustained release dosage forms," Eur. J. Pharm. Biopharm., vol. 67, no. 2, pp. 440–448, 2007.
- [32] J. Aronhime, S. Sarig, and N. Garti, "Emulsifiers as Additives in Fats Effect on Polymorphic Transformations and Crystal Properties of Fatty Acids and Triglycerides," *Food Structure*, vol. 9. pp. 337– 352, 1990.
- [33] N. Garti, J. Schlichter, and S. Sarig, "Effect of food emulsifiers on polymorphic transitions of cocoa butter," J. Am. Oil Chem. Soc., vol. 63, no. 2, pp. 230–236, 1986.
- [34] O. J. Guth, J. Aronhime, and N. Garti, "Polymorphic transitions of mixed triglycerides, SOS, in the presence of sorbitan monostearate," J. Am. Oil Chem. Soc., vol. 66, no. 11, pp. 1606–1613, 1989.

- [35] E. Laine, P. Auramo, and P. Kahela, "On the structural behaviour of triglycerides with time," *Int. J. Pharm.*, vol. 43, no. 3, pp. 241–247, 1988.
- [36] H. Bunjes, M. H. J. Koch, and K. Westesen, "Influence of emulsifiers on the crystallization of solid lipid nanoparticles," J. Pharm. Sci., vol. 92, no. 7, pp. 1509–1520, 2003.
- [37] H. Bunjes and M. H. Koch, "Saturated phospholipids promote crystallization but slow down polymorphic transitions in triglyceride nanoparticles," J. Control. Release, vol. 107, no. 2, pp. 229–243, 2005.
- [38] M. P. Even, S. Bobbala, K. L. Kooi, S. Hook, G. Winter, and J. Engert, "Impact of implant composition of twin-screw extruded lipid implants on the release behavior," *Int. J. Pharm.*, vol. 493, no. 1–2, pp. 102– 110, 2015.
- [39] Y. Akiyama, M. Yoshioka, H. Horibe, S. Hirai, N. Kitamori, and H. Toguchi, "Novel oral controlledrelease microspheres using polyglycerol esters of fatty acids," *J. Control. Release*, no. 26 (1993), pp. 1 – 10, 1993.
- [40] D. Xia, F. Cui, Y. Gan, H. Mu, and M. Yang, "Design of lipid matrix particles for fenofibrate: Effect of polymorphism of glycerol monostearate on drug incorporation and release," *J. Pharm. Sci.*, vol. 103, no. 2, pp. 697–705, 2014.
9 Summary and Future Work

The main goal of this work was the development of a microparticular, sustained release delivery system for the peptide drug gonadorelin [6-D-Phe], which is currently used in combination with other substances to influence the reproductive cycle in swine [1] [2]. Our final product should provide an environmentally friendly alternative for estrus synchronization in swine. Due to harmful effects on fish with detrimental effects on their reproduction, it is necessary to avoid the pollution of ground water with higher amounts of gestagens and their active metabolites [3]–[6]. Commonly used gestagens are usually fed in amounts of 15-20 mg/d over a time period of 10 to 18 days [7]–[9]. The peptide drug G [6-D-Phe] is a promising, rapidly metabolized alternative.

Chapter One reviews the state of the art of polymer-based delivery systems with special focus on the delivery of peptides and proteins. Manufacturing techniques, especially for lipid microparticles are discussed. Furthermore, different regimen and drugs currently used for estrus synchronization in livestock animals are introduced. Due to good physiological tolerance, lack of side effects and the low costs, solid lipids were of special interest for the development of a sustained release drug delivery system. For the injectable, controlled release system for G [6-D-Phe] for an intended release over two weeks, microparticles represent the ideal system.

The spray congealing process was investigated in **Chapter Two** to establish suitable process settings with a focus on yield and particle size. Microparticles could be produced using the triglycerides D116, D118 and D120 with a round shape and smooth surface. The placebo particles were only marginally influenced by the applied spray flow and product flow, with an overall stable process. The achieved particle size was between 30 and 70 μ m. The triglyceride particles exhibited the low melting α -polymorph, which can be transformed to the β -form upon tempering for 24 h at 45 °C.

Chapter Three dealt with the incorporation of hydrophilic model substances, specifically tryptophan and aspartame. A suitable system for the determination of drug release was developed. Pure triglycerides did not provide a sustained release profile. Addition of surfactants like GMS led to an increased burst release and a more complete drug release of tryptophan. The release could be controlled between 8 (5 %) and 24 d (10 % GMS). The addition of lipase to the release medium enhanced drug release. For aspartame, the addition of 5 and 10 % Span 40 was most promising for the release duration, which reached 12 days.

Since the product yield upon spray congealing was still low, the manufacturing process was to be optimized before production of G [6-D-Phe] microparticles in **Chapter Four.** High material

losses occurred in the spray tower and the sample container. Yield could be improved by increasing the spray flow, reducing the melt temperature and by installation of a vibrating stirrer. The encapsulation efficiency and particle size could be improved by a reduction of the spray flow. Additionally, a reduction of melt temperature followed by an increased viscosity had a positive effect on encapsulation efficiency.

The peptide drug G [6-D-Phe] could successfully be incorporated into spray congealed lipid microparticles at a theoretical concentration of 1.8 %, which was shown in **Chapter Five**. A broad screening revealed formulations with a high potential for a sustained release over two weeks. Most promising release profiles could be obtained by the addition of 5 and 10 % GMS and Span 40, with a continuous release up to 22 d. Burst release could be reduced by the suspension of lipid microparticles in thermosensitive poloxamer gels. Moreover, polymorphic behavior was evaluated using XRPD and DSC measurements showing that the added surfactant influenced crystallization behavior. Additionally, the triglyceride-emulsifier composition also influenced the water uptake upon incubation, which could be confirmed by the performance of contact angle measurements.

In **Chapter Six** the distribution behavior of lipid microparticles in suspension after injection into the neck and caudal thigh muscles ex vivo was evaluated. Therefore, dye-loaded particles were injected and distribution of the reconstitution medium, as well as the particles monitored by the means of surgery and ultrasound. Particles stayed at the injection site and a hydrophilic drug could diffuse out and distribute along the muscle fibers.

The estrus synchronization was evaluated in **Chapter Seven.** The first pre-clinical study investigated the delivery of 750 and 3750 µg/animal in two different formulations containing triglycerides and 10 % GMS. The lower dose was proved to be sufficient to achieve estrus synchronization. Using 3750 µg/animal, neither higher synchronicity nor longer duration of action could be found, but the percentage of gilts that ovulated decreased significantly. The second study focused on the application of 750 µg/animal in formulations containing triglycerides, GMS and Span 40. The drug-containing system did not have statistically more adverse reactions and was generally well tolerated. The longest duration of cycle blockage, 15.8 ± 4.2 days, was observed after the application of microparticles composed of D118 + 10 % GMS 3750 µg/animal G [6-D-Phe]. The highest synchronicity could be found after injection of D114 + 5 % Span 40 microparticles, where 5 out of 5 animals showed a cycle blockage of 6.4 ± 0.2 d. A statistically relevant delay in onset of follicular growth (after

 13.4 ± 0.2 d) could also be observed when using D114 + 5 % Span 40 microparticles, again with the highest synchronicity observed.

The long-term stability of the peptide and drug delivery system was evaluated in **Chapter Eight** over a period of twelve months. Particles were tested in terms of particle size, reconstitution properties, particle morphology, polymorphic forms, drug content and release behavior. Storage at 2-8 °C should be preferred to a storage at 25 °C, although the drug stability was not affected. Regarding particle size, polymorphic transition and release behavior, a storage of 2-8 °C could be guaranteed over the monitored time. Degradation of the peptide did not occur over 12-month storage.

Several milestones in the development of a sustained delivery system for G [6-D-Phe] have been achieved during the work on the current thesis, foremost the incorporation of the peptide into spray congealed microparticles and the variation of release duration in vitro. Additionally, this approach was identified as a possible way for estrus synchronization. The efficacy of the product was confirmed in the pre-clinical studies, revealing a high tolerance and an adequate degree of synchronicity during the follow-up study. Storage stability was confirmed over at least one year.

Future work should focus on the optimization of formulation aspects to keep the synchronous onset of estrus as achieved in the formulation containing G [6-D-Phe] 750 μ g/animal D114 + 5 % Span 40, but to prolong the duration of action to 14 days in vivo as achieved in formulations containing 750 μ g/animal G [6-D-Phe] in D118 + 10 % GMS. It should be evaluated whether a higher drug load could fasten the release, provide an extended duration of the estrus control and facilitate application as less microparticles need to be injected to achieve the same effective dose. The in vivo effects should be confirmed in further clinical studies or larger herds to evaluate the role of non-responders, adverse events and occurrence of ovulation. Moreover, the application of sustained release G [6-D-Phe] may be connected to the occurrence of permanent ovarian cysts, although we did not find statistical evidence. Since the developed drug delivery system was only tested in gilts, it would be of high interest if the effect is also present in sows. Additionally, a sterilization protocol should be developed and the influence of an irradiation sterilization should be evaluated.

9.1 References

- C. A. Herbert and T. E. Trigg, "Applications of GnRH in the control and management of fertility in female animals," Anim. Reprod. Sci., vol. 88, no. 1–2, pp. 141–153, 2005.
- [2] W. Zaremba, S. Engl, and A. Wehrend, "Gonavet®Veyx," 2015. [Online]. Available: https://www.veyx.de/uploads/tx_sitscontentattribute/KB_Gonavet_Veyx_DE_150421.pdf. [Accessed: 18-Oct-2016].
- [3] E. F. Orlando and L. E. Ellestad, "Sources, concentrations, and exposure effects of environmental gestagens on fish and other aquatic wildlife, with an emphasis on reproduction," Gen. Comp. Endocrinol., vol. 203, pp. 241–249, 2014.
- [4] J. Zeilinger, T. Steger-Hartmann, E. Maser, S. Goller, R. Vonk, and R. Länge, "Effects of synthetic gestagens on fish reproduction.," Environ. Toxicol. Chem., vol. 28, no. 12, pp. 2663–2670, 2009.
- [5] J. E. Morthorst, H. Holbech, and P. Bjerregaard, "Trenbolone causes irreversible masculinization of zebrafish at environmentally relevant concentrations," Aquat. Toxicol., vol. 98, no. 4, pp. 336–343, 2010.
- [6] S. L. Bartelt-Hunt, D. D. Snow, W. L. Kranz, T. L. Mader, C. A. Shapiro, S. J. Van Donk, D. P. Shelton, D. D. Tarkalson, and T. C. Zhang, "Effect of growth promotants on the occurrence of endogenous and synthetic steroid hormones on feedlot soils and in runoff from beef cattle feeding operations," Environ. Sci. Technol., vol. 46, no. 3, pp. 1352–1360, 2012.
- [7] F. Martinat-Botte, F. Bariteau, Y. Forgerit, C. Macar, A. Moreau, M. Terqui, and J. P. Signoret, "Control of oestrus in gilts II. Synchronization of oestrus with a progestagen, altrenogest (Regumate): Effect on fertility and litter size," Anim. Reprod. Sci., vol. 22, no. 3, pp. 227–233, 1990.
- [8] F. Martinat-Botté, F. Bariteau, Y. Forgerit, C. Macar, P. Poirier, and M. Terqui, "Synchronization of oestrus in gilts with altrenogest: effects on ovulation rate and foetal survival," Anim. Reprod. Sci., vol. 39, no. 4, pp. 267–274, 1995.
- [9] J. S. Stevenson and D. L. Davis, "Estrous synchronization and fertility in gilts after 14- or 18-day feeding of altrenogest beginning at estrus or diestrus.," J. Anim. Sci., vol. 55, no. 1, pp. 119–123, 1982.